Michael W. Davis

Actions of infinite groups on polyhedra

– Monograph –

April 29, 2020

Springer

Contents

1	Intr	oductio	n	1		
2	Poly	olyhedral preliminaries				
	2.1	Cell co	- omplexes, links	3		
	2.2	The C	AT(0)-inequality	4		
		2.2.1	Piecewise hyperbolic polyhedra	7		
		2.2.2	A few remarks concerning word hyperbolic groups	8		
		2.2.3	Reshetnyak's Gluing Lemma	9		
	2.3	Flag c	omplexes and Gromov's Lemma	9		
	2.4	Some	first examples of NPC cube complexes	10		
		2.4.1	Products	10		
		2.4.2	Alternating link complements	11		
		2.4.3	Square complexes	11		
		2.4.4	Configuration spaces of a graph	11		
		2.4.5	Graph braid groups embed in RAAGs	12		
•	Right-angled complexes and groups					
3	Rigi	ıt-angle	ed complexes and groups	15		
3	Rigi 3.1	it-angle Polyhe	ed complexes and groups	15 15		
3	Rigi 3.1	nt-angle Polyhe 3.1.1	ed complexes and groups edral products The cube complexes K_L , P_L , Z_L , and \mathbb{T}_L	15 15 16		
3	Rigi 3.1	nt-angle Polyhe 3.1.1 3.1.2	ed complexes and groups edral products The cube complexes K_L , P_L , Z_L , and \mathbb{T}_L Polyhedral products of NPC cube complexes	15 15 16 19		
3	Rigi 3.1	nt-angle Polyhe 3.1.1 3.1.2 3.1.3	ed complexes and groups edral products The cube complexes K_L , P_L , Z_L , and \mathbb{T}_L Polyhedral products of NPC cube complexes Graph products of groups	15 15 16 19 19		
3	Rigi 3.1	nt-angle Polyhe 3.1.1 3.1.2 3.1.3 3.1.4	ad complexes and groups $Ext{charge}$ dral products The cube complexes K_L , P_L , Z_L , and \mathbb{T}_L Polyhedral products of NPC cube complexes Graph products of groups Graph wreath products	15 15 16 19 19 21		
3	Rigi 3.1	nt-angle Polyhe 3.1.1 3.1.2 3.1.3 3.1.4 3.1.5	ad complexes and groups Edral products The cube complexes K_L , P_L , Z_L , and \mathbb{T}_L Polyhedral products of NPC cube complexes Graph products of groups Graph wreath products Application: cohomological dimension versus geometric dimension	15 15 16 19 19 21 22		
3	Rigi 3.1 3.2	nt-angle Polyhe 3.1.1 3.1.2 3.1.3 3.1.4 3.1.5 Reflec	ad complexes and groups E dral products The cube complexes K_L , P_L , Z_L , and \mathbb{T}_L Polyhedral products of NPC cube complexes Graph products of groups Graph wreath products Application: cohomological dimension versus geometric dimension tion groups	15 15 16 19 19 21 22 23		
3	Rigi 3.1 3.2	Polyhe 3.1.1 3.1.2 3.1.3 3.1.4 3.1.5 Reflec 3.2.1	ed complexes and groups edral products. The cube complexes K_L , P_L , Z_L , and \mathbb{T}_L . Polyhedral products of NPC cube complexes Graph products of groups Graph wreath products Application: cohomological dimension versus geometric dimension tion groups Davis complex	15 15 16 19 19 21 22 23 24		
3	Rigi 3.1 3.2	nt-angle Polyhe 3.1.1 3.1.2 3.1.3 3.1.4 3.1.5 Reflec 3.2.1 3.2.2	ad complexes and groups adral products. The cube complexes K_L , P_L , Z_L , and \mathbb{T}_L . Polyhedral products of NPC cube complexes Graph products of groups Graph wreath products. Application: cohomological dimension versus geometric dimension tion groups. Davis complex Semidirect products	15 15 16 19 19 21 22 23 24 24		
3	Rigi 3.1 3.2	nt-angle Polyhe 3.1.1 3.1.2 3.1.3 3.1.4 3.1.5 Reflec 3.2.1 3.2.2 3.2.3	ad complexes and groups adral products. The cube complexes K_L , P_L , Z_L , and \mathbb{T}_L . Polyhedral products of NPC cube complexes Graph products of groups Graph wreath products. Application: cohomological dimension versus geometric dimension Davis complex Semidirect products The reflection group trick	15 15 16 19 21 22 23 24 24 24 24		
3	Rigi 3.1 3.2 3.3	nt-angle Polyhe 3.1.1 3.1.2 3.1.3 3.1.4 3.1.5 Reflec 3.2.1 3.2.2 3.2.3 Right-	ad complexes and groups adral products. The cube complexes K_L , P_L , Z_L , and \mathbb{T}_L . Polyhedral products of NPC cube complexes Graph products of groups Graph wreath products. Application: cohomological dimension versus geometric dimension tion groups. Davis complex Semidirect products The reflection group trick angled buildings	15 15 16 19 21 22 23 24 24 24 24 24		
3	Rigi 3.1 3.2 3.3 3.4	nt-angle Polyhe 3.1.1 3.1.2 3.1.3 3.1.4 3.1.5 Reflec 3.2.1 3.2.2 3.2.3 Right- Right-	ad complexes and groups bedral products. The cube complexes K_L , P_L , Z_L , and \mathbb{T}_L . Polyhedral products of NPC cube complexes Graph products of groups Graph wreath products. Application: cohomological dimension versus geometric dimension tion groups. Davis complex Semidirect products The reflection group trick angled buildings angled Artin groups	15 15 16 19 21 22 23 24 24 24 24 24 24		
3	Rigi 3.1 3.2 3.3 3.4	nt-angle Polyhe 3.1.1 3.1.2 3.1.3 3.1.4 3.1.5 Reflec 3.2.1 3.2.2 3.2.3 Right- Right- 3.4.1	ad complexes and groups bdral products. The cube complexes K_L , P_L , Z_L , and \mathbb{T}_L . Polyhedral products of NPC cube complexes Graph products of groups Graph wreath products. Application: cohomological dimension versus geometric dimension tion groups. Davis complex Semidirect products The reflection group trick angled buildings The standard cube complex	15 15 16 19 19 21 22 23 24 24 24 24 24 24 24 25		

Contents

4	Cox	Coxeter groups, Artin groups, buildings			
	4.1	Some simple complexes of groups 2	28		
	4.2	Coxeter groups	29		
		4.2.1 Spherical Coxeter groups	30		
		4.2.2 Coxeter zonotopes	31		
		4.2.3 The Davis-Moussong complex	31		
		4.2.4 Moussong's Lemma	34		
		4.2.5 Piecewise euclidean metrics on <i>K</i>	35		
		4.2.6 The Tits representation	36		
	4.3	Artin groups	37		
		4.3.1 The complement of the complexification of a reflection arrangement	37		
		4.3.2 The Salvetti complex and its universal cover	39		
		4.3.3 The Deligne complex 4	40		
	4.4	Buildings 4	42		
		4.4.1 The combinatorial theory of buildings 4	43		
		4.4.2 Geometric realizations of buildings 4	45		
		4.4.3 Using covering spaces to construct buildings 4	47		
		4.4.4 Constructions, lattices 4	49		
5	Simple complexes of groups				
	5.1	Definitions	52		
	5.2	The basic construction	54		
	5.3	Examples where Ω is the dual face poset of a simplicial complex 5	57		
	5.4	.4 Polygons of groups			
	5.5	.5 Aspherical realizations and the $K(\pi, 1)$ -Question			
	5.6	The semidirect product construction			

vi

Chapter 1 Introduction

The road between topology and group theory goes through the fundamental group. Suppose X is a space that admits a universal covering space (i.e., X is semi-locally 1-connected). Its fundamental group acts as the group of deck transformations on the universal cover \tilde{X} . Given a presentation for a group π , there is a standard construction of a connected, 2-dimensional CW complex X with $\pi_1(X) = \pi$. (Start with a wedge of circles, one for each generator; then attach a 2-cell for each relation.) One can continue to attach cells to X of dimension greater than 2 to kill all the higher homotopy groups $\pi_i(X)$, $i \ge 2$. The result is a CW complex with fundamental group π and all higher homotopy groups = 0. We denote such a space $B\pi$ and call it the classifying space of π . Standard arguments in homotopy theory show that $B\pi$ is unique up to homotopy equivalence, e.g., see [80]. Thus, algebraic topological properties of $B\pi$ such as its cohomology are properties of the group π . The universal cover of $B\pi$, denoted $E\pi$, is contractible. The space $B\pi$ is also called an Eilenberg-MacLane space $K(\pi, 1)$. Such a space with vanishing higher homotopy groups is said to be *aspherical*.

Usually, the CW complex $B\pi$ cannot be taken to have a finite number of cells. For example, if π has nontrivial torsion, then $B\pi$ must be infinite dimensional.

Suppose a group *G* acts on a cell complex *X* by an action that permutes the cells. We write $G \cap X$ to mean that *G* acts on *X*. There are several different situations to consider. First, the complex *X* could be locally infinite. In this case, cell stabilizers might well be infinite. For example this is often the case in Serre's theory of groups acting on trees. If *X* is locally finite, then Aut(*G*) is a locally compact, totally disconnected group. Although the cell stabilizers might still be infinite, we will primarily be interested in the case when *G* is a discrete subgroup of Aut(*X*). The group *G*, is then said to act *properly* on *X*. Finally, the cell stabilizers in *G* might be trivial. In this case, the *G*-action is *free*. The *G*-action is *cocompact* if the orbit space *X*/*G* is compact.

If a group G has nontrivial torsion, then BG cannot be finite dimensional. (The proof is due to P.A. Smith: if H is a finite cyclic subgroup of G, then $BH \rightarrow BG$ is a covering space. If H is nontrivial, BH has cohomology in arbitrarily high degrees. So, BH and a fortiori BG must be an infinite dimensional.)

A group G is virtually torsion-free if it has a torsion-free subgroup Γ of finite index. The virtual cohomological dimension of G, denoted by vcd G, is the cohomolgical dimension of such a Γ . The group G is type VF if there is a finite index subgroup Γ of type F. (These notions are independent of the choice of Γ .)

When a group G has nontrivial torsion, there is a notion of a "universal space for proper actions" which is sometimes more useful than the universal space EG.

Definition 1.1. For a discrete group G, its *universal proper G-space* is a CW complex <u>EG</u> together with a proper cellular action, $G \curvearrowright \underline{E}G$, so that for any finite subgroup H < G, the subcomplex, $(\underline{E}G)^H$, which is

later

fixed by *H*, is contractible. (In particular, if every finite subgroup of *G* is trivial, then $\underline{E}G = EG$.) The quotient complex $\underline{E}G/G$ is denoted by $\underline{B}G$. (The complex $\underline{B}G$ is an orbihedron.)

Definition and next 2 paragraphs might go in intro

For example, if *Y* a locally finite CAT(0) polyhedron and $G \cap Y$ by isometries, then $Y = \underline{E}G$. The universal property of $\underline{E}G$ is that for any CW complex *Y* with a cellular, proper *G*-action (i.e., with finite isotropy groups), there is an equivariant map $Y \to \underline{E}G$, unique up to equivariant homotopy.

Chapter 2 Polyhedral preliminaries

2.1 Cell complexes, links

By a *cell complex* we will mean a space *X* formed by gluing together convex polytopes (in some euclidean space) via isometries of their faces. The decomposition of *X* into polytopes ("cells") is part of the structure. The word "polyhedron" will be used synonymously with "cell complex".

In classical definitions of a convex cell complex, one often adds the conditions that each cell is embedded and that the intersection of two cells is either empty or a common face of each. We call these two requirements the *classical conditions*. The first of these conditions rules out the possibility of a 1-gon while the second rules out a 2-gon. (Even when X does not satisfy the classical conditions, the induced cell structure on its universal cover often will satisfy them.) Two types of cell complexes are of particular interest: first, if each cell is required to be a simplex we have " Δ -complex" in the sense of Hatcher [80], second if each cell is a cube, we have a "cube complex". A Δ -complex or a cube complex can have "multiple faces," e.g., in a Δ -complex two different simplices can have the same vertex set. If a Δ -complex satisfies the classical conditions, then it is sometimes called a *simple cube complex*. (for example this terminology is used in [79]).

Given a face *F* of a convex polytope *P* and a point *x* in its relative interior, the *normal cone* N(F,P) is the set of all inward-pointing tangent vectors at *x* that are orthogonal to *F*. It is a convex polyhedral cone in the the tangent space T_xP . The dimension of N(F,P) is the codimension of *F* in *P*. The *link of F in P*, denoted Lk(F,P), is set of unit tangent vectors in N(F,P). In other words, Lk(F,P) is the space of inward-pointing normal directions to *F* in *P*. It is a geodesically convex polytope in the unit sphere of T_xP . If F < F', where *F'* is another face of *P*, then Lk(F,F') is identified with a face of Lk(F,P). Given a cell τ of a cell complex *X*, we can then glue together the corresponding links to to form a cell complex:

$$Lk(\tau, X) := \bigcup_{\substack{\sigma \\ \tau < \sigma}} Lk(\tau, \sigma), \tag{2.1}$$

called the *link of* τ *in X*. (The definition given in (2.1) needs to be slightly modified when there are self-gluings. For example, *P* is the polytope corresponding to the cell σ and faces F_1 , F_2 of *P* are identified to get a single cell τ of *X*, then Lk(τ, σ) be interpreted to be Lk(F_1, P) \cup Lk(F_2, P)')

For example, if *P* an *n*-simplex and F < P is a *k*-dimensional face, then Lk(*F*,*P*) is a spherical (n-k-1)-simplex. So, if *X* is a Δ -complex and σ is a cell in *X*, then Lk(σ ,*X*) is a Δ -complex. Moreover, if *X* is a simplicial complex, then so is Lk(σ ,*X*). Similarly, if *F* is a *k*-dimensional face of an *n*-cube *P*, then Lk(*F*,*P*)

2 Polyhedral preliminaries

is a (n-k-1)-simplex. Hence, if X is a cube complex, then $Lk(\sigma, X)$ is a Δ -complex and if X is a simple cube complex, then $Lk(\sigma, X)$ is a simplicial complex.

One can define the notion a geodesically convex polytope in any simply connected space of constant curvature κ . In dimension $n \ge 2$ the possibilities are euclidean *n*-space \mathbb{E}^n for $\kappa = 0$, the *n*-sphere of curvature κ , \mathbb{S}_{κ}^n , for $\kappa > 0$, and hyperbolic *n*-space of curvature κ , \mathbb{H}_{κ}^n , , , A cell complex *X* made by gluing together polytopes in a space of constant curvature κ is called a constant curvature κ is called a *constant curvature* κ is called a *constant curvature* μ polyhedron. (See [23] or [48] for more details.) We shall be interested only in the cases $\kappa = +1$, 0, or -1. We shall say that *X* is *piecewise spherical*, *piecewise euclidean* or *piecewise hyperbolic*, respectively. If a constant curvature polyhedron *X* is connected, then it inherits a natural length metric as follows. The length of a geodesic segment in a polytope of constant curvature κ is defined as usual. If $x, y \in X$, then d(x, y) is defined to be the infimum of the lengths of all piecewise linear paths from *x* to *y*. If *X* is locally finite, then this gives *X* the structure of a *geodesic space* meaning that the infimum d(x, y) is always realized by a geodesic path (= isometric embedding), $\gamma : [0,d] \to X$ with $\gamma(0) = x$, $\gamma(d) = y$. (For more details see [23, Ch I.3].)

Is this correct?

Since the link of a face of any polytope in a constant curvature space is always a spherical polytope in \mathbb{S}_1^n , we see that the link of any cell of a constant curvature polyhedron is always a piecewise spherical polyhedron ($\kappa = 1$). Similarly, for any point $x \in X$, let Lk(x,X) be the union of all unit tangent vectors which point into a cell of *X*. (If *x* belongs to the relative interior of a *k*-cell σ , then $Lk(x,\sigma) = S^{k-1}$. When this is the case it follows that Lk(x,X) can be identified with $\mathbb{S}^{k-1} * Lk(\sigma,X)$, the *k*-fold suspension of $Lk(\sigma,X)$.) Let $N_{\varepsilon}(x,X)$ (resp. $\partial N_{\varepsilon}(x,X)$) denote the ball (resp. sphere) of radius ε about *x*, i.e., $N_{\varepsilon}(x,X) = \{y \in X \mid d(x,y) \le \varepsilon\}$ and $\partial N_{\varepsilon}(x,X) = \{y \in X \mid d(x,y) = \varepsilon\}$. One sees $\partial N_{\varepsilon}(x,X)$ can be identified with Lk(x,X) and that

$$N_{\varepsilon}(x,X) \cong \operatorname{Cone}_{\varepsilon}(\operatorname{Lk}(x,X)),$$
 (2.2)

where $\text{Cone}_{\varepsilon}(A)$ stands for the *cone of radius* ε , that is, $A \times [0, \varepsilon]$ with $A \times 0$ collapsed to a point. Formula (2.2) means that Lk(x, X) determines the local topology of X. For example, X is a PL *n*-manifold if and only if Lk(x, X) is piecewise linearly homeomorphic to S^{n-1} for each $x \in X$.

2.2 The CAT(0)-inequality

Using ideas of Alexandrov and Busemann, Gromov [76] defined what it means for a geodesic metric space *X* to have curvature bounded above by a real number κ (where usually, $\kappa \in \{-1, 0, +1\}$). First, a geodesic triangle in *X* satisfies the CAT(κ)-*inequality*, if the distance between any two points of the triangle is \leq the distance between the corresponding points of a comparison triangle in the simply connected 2-manifold of constant curvature κ , i.e., in the hyperbolic plane, the euclidean plane, or the 2-sphere, as $\kappa = -1,0$ or +1. A geodesic space *X* is CAT(κ) if each triangle satisfies the CAT(κ)-inequality. The CAT(κ)-inequality implies the CAT(κ')-inequality for $\kappa' > \kappa$. In particular, if a space is CAT(κ) with $\kappa \leq 0$, then it is CAT(0).

Remark 2.1. he terminology "CAT(κ)-inequality" is due to Gromov [76, p.106]. The acronym CAT stands for "comparison inequality of Alexandov and Toponogov," although sometimes the "C" is said to stand for "Cartan."

The uniqueness of the geodesic connecting two points in a CAT(0) space X follows immediately from the definitions. With a little work one can see that the constant map $X \to X$ which each point to a basepoint is homotopic to the identity via geodesic contraction. This gives the following.

Theorem 2.1. If a complete geodesic space is CAT(0), then it is contractible.

2.2 The CAT(0)-inequality

The space X has curvature $\leq \kappa$ if it satisfies the CAT(κ)-inequality locally. We say that X is nonpositively curved (abbreviated NPC) if this is true for $\kappa = 0$. (See [23].)

Theorem 2.2. (Globalization Theorem of Cartan-Hadamard, cf. [76, p. 119], [23, p. 193], or [4]). Suppose X is a complete geodesic space of curvature $\leq \kappa$

(i) Suppose $\kappa \leq 0$. Then X is CAT(κ) if and only if it is simply connected.

(ii) ([76, p. 122]. Suppose $\kappa > 0$. Then X is $CAT(\kappa)$ if and only if it has no closed geodesic of length $\leq 2\pi/\sqrt{\kappa}$.

An immediate corollary to the first part of this theorem is the following.

Corollary 2.1. Suppose a complete geodesic space X has curvature $\leq \kappa$ with $\kappa \leq 0$. Then its universal cover omit \tilde{X} is CAT(κ).

The relevance of nonpositive curvature to geometric group theory is explained by the next theorem, which combines Theorem 2.1 and Corollary 2.1.

Theorem 2.3. If a complete geodesic space is NPC, then it is aspherical.

Just as in the case where a polyhedron is obtained by gluing together euclidean polytopes, one can consider a polyhedron X where each cell is identified with some convex polytope in a space of constant curvature κ . In general we call such an X a *piecewise constant curvature polyhedron* and we say that it is *piecewise hyperbolic, piecewise euclidean*, or *piecewise spherical* as $\kappa = -1, 0, or + 1$, respectively. As before any such X is a length space and in fact, a geodesic space whenever it is locally finite.

Example 2.1. (*Dimension one*). If X is a connected graph with finitely many edges and we assign a length to each edge, then X is a constant curvature polyhedron for any curvature κ for any κ . It is locally CAT(κ). So, if each edge of X is assigned length 1, then X is an NPC cube complex. If X is a tree and each edge has length 1, then, by Theorem 2.2 (i), X is a CAT(0) cube complex. We can also regard the metric on a graph X as giving it a piecewise spherical structure. In this case, by part (ii) of Theorem 2.2, X is CAT(1) if and only if it has no circuit (= closed geodesic) of length $\leq 2\pi$.

We turn now to the question of when a piecewise constant curvature polyhedron (with the curvature constant $= \kappa$) has curvature $\leq \kappa$. The answer is provided by the "Link Condition," stated in Theorem 2.4 below. The proof of the Link Condition is a combination of three facts: first, the link of any cell in a piecewise constant curvature polyhedron X is naturally a piecewise spherical polyhedron; second, an open ball centered at a point $x \in X$ is isometric to an open neighborhood of the cone point in the cone (or " κ -cone") on a piecewise spherical polyhedron (essentially the link of x), and third, the κ -cone on a piecewise spherical polyhedron L is CAT(κ) if and only if L is CAT(1). Since such graphs occur as links in 2-dimensional constant curvature polyhedra, we get, via the Link Condition in Theorem 2.4, a condition for a 2-dimensional polyhedron to be locally CAT(κ).

As was pointed out in Section 2.1, if *F* is a face of a geodesically convex polytope *P* in constant curvature space, then Lk(F,P) is a polytope in the unit sphere of the tangent space. Hence, if τ is a cell of a constant curvature polyhedron *X*, then $Lk(\tau,X)$ has a natural piecewise spherical structure. If *x* is any point of a face *F* of a polytope *P*, then let Lk(x,P) be the space of inward-pointing tangent vectors at *x*. This space is a join $S_x(F) * Lk(F,P)$ where S_x is the unit sphere in T_xF . This space has a length metric as a subspace of the unit sphere in T_xP . (This is the metric of the "spherical join" defined in [23, p. 63]. The union of these spherical joins at a point *x* of the polyhedron *X* is denoted Lk(x,X) and is called the *space of directions at x*.

Lemma 2.1. Let X be a piecewise constant curvature polyhedron, τ a cell of X and x a point in X. Then $Lk(\tau, X)$ and Lk(x, X) are piecewise spherical polyhedra.

Suppose *L* is a piecewise spherical polyhedron. Define the *euclidean cone* on *L* by

$$\operatorname{Cone} L := (L \times [0, \infty)) / \sim, \tag{2.3}$$

where $L \times 0$ is identified to a point. If $(\theta, r) \in \mathbb{S}^{n-1} \times [0, \infty)$ are polar coordinates in \mathbb{R}^n , then by using the Law of Cosines one can write a formula for the euclidean distance on \mathbb{R}^n . By using $(\theta, r) \in L \times [0, \infty)$ as polar coordinatest he same formula defines a metric on the euclidean cone, Cone(L). is defined using $(\theta, r) \in L \times [0, \infty)$ as polar coordinates and the Law of Cosines in euclidean space. Similarly, for any real number κ and nonnegative real number r, define the κ -cone of radius r on L, by

$$\operatorname{Cone}_{r}^{\kappa}(L) := (L \times [0, r]) / \sim, \tag{2.4}$$

where the metric is defined using polar coordinates and the Law of Cosines in the space of constant curvature κ . (See [23, p. 59] for more details.) If $\kappa > 0$, we require the radius of the cone to be $\leq \pi/\sqrt{\kappa}$.

Lemma 2.2. ([23, pp. 206-207]). $\operatorname{Cone}_{r}^{\kappa}(L)$ is $\operatorname{CAT}(\kappa)$ if and only if L is $\operatorname{CAT}(1)$.

Lemma 2.2 immediately implies the second sentence of the next lemma.

Lemma 2.3. ([23, pp. 206-207]). If x is a point of a piecewise constant curvature polyhedron X, then for small enough ε the ball of radius ε centered at x is isometric to $\text{Cone}_{\varepsilon}^{\kappa}(\text{Lk}(x,X))$. Hence, X has curvature $\leq \kappa$ if and only if for all $x \in X$, Lk(x,X) is CAT(1).

Definition 2.1. A piecewise constant curvature polyhedron X satisfies the *Link Condition* if for each vertex v, each connected component of Lk(v, X) is CAT(1).

Theorem 2.4. (The Link Condition, cf. [23, Theorem 5.4, p. 206]). A piecewise constant curvature polyhedron X has curvature $\leq \kappa$ if and only if it satisfies the Link Condition.

Theorem 2.4 is almost the same as Lemma 2.3. The only difference is that in Lemma 2.3 links are CAT(1) at all points $x \in X$, while in Theorem 2.4 this need only be true at vertices.

Should be Definitionstyle? Example 2.2. (The Link Condition for 2-dimensional piecewise euclidean polyhedra).

a) Let *X* be a piecewise euclidean polyhedron of dimension 2. If *X* is a surface, then the link of any vertex *v* is then a circle. Each edge of the link corresponds to a 2-cell containing *v*. With its natural spherical structure, the length of the edge is equal to the interior angle of the 2-cell at *v*. So, we get the following well-known statement statement from folklore: the surface *X* is NPC if and only if the sum of the angles at each vertex is $\geq 2\pi$. If *X* is a general 2-dimensional piecewise euclidean polyhedron, then the link of each vertex is a graph and the Link Condition says that *X* is NPC if and only if for each vertex *v*, Lk(*v*,*X*) contains no circuit of length $< 2\pi$ (cf. Example 2.1).

b) If X is a square complex, then since each interior angle in a square is $\pi/2$, the Link Condition says each link must be simple graph without 3-circuits. More generally, if each cell of X is a regular euclidean *m*-gon, then each interior angle is $(m-2)\pi/2m$ and hence, this is the length of each edge in Lk(v,X). So, when $m \ge 6$, we see that X is NPC whenever each link is a simple graph (i.e., no circuits of length 1 or 2.

2.2 The CAT(0)-inequality

Example 2.3. (*Regular 2k-gonal complexes*, cf. Examples 4.4 and 4.6 of Section 4.2). Suppose L^1 is a simple graph and that k is an integer ≥ 2 . In Section 4.2 by using the theory of Coxeter groups we shall see that there is a simply connected 2-complex X_k so that each 2-cell is a regular euclidean 2k-gon and the link of each is L^1 . Since the vertex angle of a regular euclidean 2k-gon is $\pi - \pi/k$, L^1 should be given a piecewise spherical structure with each edge having length $\pi - \pi/k$. When k = 2, L^1 is CAT(1) if and only if it has no circuits of length 3, and when k > 2, it is always CAT(1). Thus, for k > 2 the 2-complex X_k is CAT(0) for any L^1 , while for k = 2 it is CAT(0) provided L^1 has no circuits of length 3.

To prove Theorem 2.4 we need Lemmas 2.4 and 2.5 below. Before stating Lemma 2.4 we need the notion of a "spherical join". For i = 1, 2, suppose σ_i is a spherical polytope in \mathbb{S}^{m_i} . The join these two polytopes, denoted $\sigma_1 * \sigma_2$, is naturally a spherical polytope in $\mathbb{S}^{m_1+m_2+1}$. The euclidean cones $\text{Cone}^0(\sigma_i)$ are polyhedral cones in \mathbb{R}^{m_i+1} and the product $\text{Cone}^0(\sigma_1) \times \text{Cone}^0(\sigma_2)$ is the polyhedral cone in $\mathbb{R}^{m_1+m_2+1}$ corresponding to the spherical polytope $\sigma = \sigma_1 * \sigma_2$. In other words, the link of cone point of $\text{Cone}^0(\sigma_1) \times \text{Cone}^0(\sigma_2)$ is the spherical join $\sigma = \sigma_1 * \sigma_2$, where the spherical polytope $\sigma_1 * \sigma_2$ has its natural spherical metric. If, for $i = 1, 2, L_i$ is a piecewise spherical polyhedron, then the *spherical join*, $L_1 * L_2$, is defined by

$$L_1 * L_2 := \bigcup_{(\sigma_1, \sigma_2)} \sigma_1 * \sigma_2, \tag{2.5}$$

(Here $(\sigma_1, \sigma_2) \in \mathcal{P}(L_1) \times \mathcal{P}(L_2)$, where $\mathcal{P}(L_i)$ means the poset of cells in L_i , including the empty cell, and either σ_1 or σ_2 is nonempty.)

The proof of the next lemma is easy.

Lemma 2.4. Suppose L_1 and L_2 are piecewise spherical polyhedra. Then $L_1 * L_2$ is CAT(1) if and only if both L_1 and L_2 are CAT(1).

Lemma 2.5. If X is a constant curvature polyhedron, then either of the following conditions is equivalent to the Link Condition for X.

(1) For each $x \in X$, Lk(x, X) is CAT(1).

(2) For each cell τ of X, Lk (τ, X) is CAT(1).

(3) For each cell τ of X, Lk (τ, X) has no closed geodesic of length $\leq 2\pi$.

Proof. If *x* lies in the relative interior of a *k*-dimensional cell τ of *X*, then Lk(*x*,*X*) is isometric to the spherical join $\mathbb{S}^{k-1} * \text{Lk}(\tau, X)$. By Lemma 2.5, either Lk(*x*,*X*) and Lk(τ, X) are both CAT(1) or neither is CAT(1).

Proof (of Theorem 2.4). Given a vertex $v \in X$, put L = Lk(v, X). If τ is a cell of X containing v, then there is a corresponding cell τ' in L (with dim $\tau' = \dim \tau - 1$) and $Lk(\tau, X)$ can be identified with $Lk(\tau', L)$. If L is locally CAT(1), then $Lk(\tau', L)$ is CAT(1). Hence, if the link of each vertex of X is CAT(1), then $Lk(\tau, X)$ is CAT(1) for each cell τ of X and therefore, by Lemma 2.5, Lk(x, X) is CAT(1) for all points $x \in X$. By Lemma 2.3, this is equivalent to X having curvature $\leq \kappa$.

2.2.1 Piecewise hyperbolic polyhedra

One also can consider polyhedra of constant curvature -1 (i.e., *piecewise hyperberolic polyhedra*). Each cell of such a piecewise hyperbolic cell complex is a convex polytope in some hyperbolic space \mathbb{H}^n . As before,

the link of each cell in this cell complex is piecewise spherical complex and by Theorem 2.4, X has curvature ≤ -1 if and only if each link is CAT(1).

Next we consider the possibility of deforming the metric on a piecewise euclidean cell complex X_e to a piecewise hyperbolic cell complex X_h . If X_e is NPC, then when will X_h be CAT(-1)? The problem is that since the angles change when we deform a euclidean polytope to one which is hyperbolic, the links in X_{e} might not remain CAT(1) in X_h . If P_e is a small euclidean polytope, say with all edge lengths $< \varepsilon$, then a small deformation will be small hyperbolic polytope P_h . The link of a face in P_h will be a small deformation of the link of the corresponding face of P_h . A precise formulation of this result can be found in [102]. However, after even a small deformation, the link of the cell in X_h may fail to be CAT(1). The reason is that if a closed geodesic in the link of a cell in a piecewise euclidean cell complex has length $= 2\pi$, then under a small deformation its length can become $< 2\pi$. For example, each angle of a regular convex 4-gon in \mathbb{H}^2 is $< \pi/2$. So, if a euclidean square complex has a vertex link with a circuit with 4-edges, then after deformation this circuit will become a closed geodesic in the link of length $< 2\pi$; hence, the hyperbolic square complex will fail to be locally CAT(-1). The condition that the link have no such 4-circuits is the "no \Box condition" discussed below. More generally, given a compact 2-dimensional piecewise euclidean cell complex X, it can deformed to a piecewise hyperbolic complex of curvature ≤ -1 provided that for each vertex v of X, each circuit of Lk(v,X) has length > 2π , cf. Example 2.2. (It may be necessary be necessary to first rescale the metric on X so that all its edge lengths are small.)

Definition 2.2. A piecewise spherical complex *L* is *large* if it is CAT(1). The cell complex *L* is *extra large* if the length of the shortest closed geodesic in *L* is strictly greater than 2π and if the same holds true for $Lk(\sigma, L)$ for any cell σ in *L*.

Lemma 2.6. (Moussong [102, Lemma 5.11]). Suppose L is a piecewise spherical cell complex with finitely many cells. If L is extra large, then any sufficiently small deformation of it will be CAT(1)

A corollary of this lemma is the following.

Proposition 2.1. (Moussong [102]). Suppose X is a NPC piecewise euclidean cell complex with finitely many shapes of cells so that for each cell σ of X, Lk(σ ,L) is extra large. Then, after rescaling the metric, X can be deformed to a piecewise hyperbolic cell complex with curvature ≤ -1 .

2.2.2 A few remarks concerning word hyperbolic groups

The most important chapter in geometric group theory begins with Gromov's work on hyperbolic groups in [76]. He first defines what it means for metric space to be hyperbolic - his definition encapsulates the notion of having negative curvature in the large. The definition is easiest to state for a geodesic metric spaces. A geodesic in triangle in a geodesic metric space X is δ -thin if the distance from any point x on one side of the triangle to the union of the other two sides is $\leq \delta$. The geodesic space X is *hyperbolic* if there is a constant $\delta > 0$ so that every triangle in X is δ -thin. (See [23, pp. 408–409], [76].) A finitely generated group G with a finite set of generators S is word hyperbolic if its Cayley graph Cay(G,S) is a hyperbolic metric space. The property of being hyperbolic is invariant under quasi-isometry. It follows that the property of a group being word hyperbolic is independent of the choice of generating set S. The action of a group on a metric space X by isometries is geometric if the action is proper and cocompact. The Fundamental Lemma of Geometric Group Theory states that if a group G acts geometrically on two metric spaces X and X' then they are quasi-isometric; in particular, any such X is quasi-isometric to Cay(G,S). Since the hyperbolic plane \mathbb{H}^2

insert reference

2.3 Flag complexes and Gromov's Lemma

is a hyperbolic, any CAT(-1)-space is a hyperbolic metric space; hence, any group which acts properly on a CAT(-1)-space is word hyperbolic. The above discussion is summarized in the next lemma.

Lemma 2.7. If a group G acts geometrically on a piecewise hyperbolic CAT(-1) cell complex X, then G is word hyperbolic.

2.2.3 Reshetnyak's Gluing Lemma

2.3 Flag complexes and Gromov's Lemma

A Δ -complex L is a *flag complex* if it is a simplicial complex and if it satisfies the following: any finite set of vertices that are pairwise connected by edges of L, spans a simplex of L. In other words, any clique in the 1-skeleton L^1 spans a simplex of L. (A *clique* in a simplicial graph Γ is a finite set of vertices of Γ such that the induced subgraph is a complete graph. Note that a complete graph is the 1-skeleton of a simplex.)

Remark 2.2. (Comments about terminology). Combinatorialists use the term "clique complex" instead of "flag complex". The *clique complex* of a simplicial graph Γ is the simplicial complex whose simplices are the cliques in Γ . So, a simplicial complex *L* is a flag complex if and only if *L* is equal to the clique complex determined by L^1 .

Here is a slick version of this definition. A simplicial complex L is a flag complex if and only if every non-simplex contains a non-edge (a set of vertices is a "non-simplex" if it is not the vertex set of a simplex).

A simplicial graph is a *circuit of length* m if it is isomorphic to the boundary complex of an m-gon). A circuit Γ of length m is a flag complex if and only if m > 3. More generally, a simplicial graph Γ is a flag complex if and only if it contains no circuits of length 3.

Instead of "flag complex" Gromov [76] used the terminology that L satisfies "the no Δ condition" (pronounced "the no triangle condition").

A natural way in which flag complexes arise is as order complexes of partially ordered sets.

Example 2.4. (Barycentric subdivisions). If X is a classical convex cell complex, then the order complex of its poset of cells is a simplicial complex. The geometric realization of the order complex is the barycentric subdivision of X. It follows that the requirement of being a flag complex puts no restriction on the topological type of X.

The next lemma is left as an exercise for the reader.

Lemma 2.8. (Some properties of flag complexes).

- (i) A full subcomplex of a flag complex is a flag complex.
- (ii) The join of two flag complexes is a flag complex.
- (iii) The link of a simplex in a flag complex is a flag complex.

Lemma 2.9. (Gromov's Lemma, cf. [76, p. 122], [23, p. 211] or [48, Lemma I.6.1]). Suppose L is a finite dimensional all right, piecewise spherical simplicial complex (or Δ -complex). Then L is CAT(1) if and only if it is a flag complex.

2 Polyhedral preliminaries

Example 2.5. (An all right simplicial complex that is not CAT(1)). Start with the triangulation of \mathbb{S}^2 as the boundary complex of an octahedron in which each 2-simplex is all right. Let L be the complement of the interior of one of the 2-simplices Δ . Of course, L is not a flag complex since it has an empty triangle, namely $\partial \Delta$. Note that $\partial \Delta$ is a short closed geodesic in L (its length is $3\pi/2$). (To see that $\partial \Delta$ is a closed geodesic, it suffices to show that it is a local geodesic. This follows from the fact that at each vertex of $\partial \Delta$ the link is a circular arc of length $3\pi/2$.)

Combining Gromov's Lemma with the observation in Example 2.4 about barycentric subdivisions, we get the following well-known result of Berestovskii.

Corollary 2.2. (Berestovskii [11]). Any finite dimensional polyhedron can be given a piecewise spherical CAT(1) metric.

The form in which Lemma 2.9 will most often be used is that in the following corollary. This corollary will also be referred to as "Gromov's Lemma."

Corollary 2.3. (Gromov's Lemma for cube complexes). A piecewise euclidean cube complex is NPC if and only if the link of each vertex is a flag complex.

Proof (of Gromov's Lemma). If L has curvature ≤ 1 and is not a flag complex, then an argument similar to that in Example 2.5 shows that L has a short geodesic. It follows by induction on dimension that if L is not a flag complex, then at least one of its links has a short closed geodesic of length $3\pi/2$.

In all right simplicial complex the closed star of a vertex v is the closed ball $B_{\pi/2}(v)$. An abbreviated version of Gromov's argument goes as follows. Suppose L is a flag complex and that γ is a closed geodesic which intersects the interior of $B_{\pi/2}(v)$ in a geodesic arc γ' . The claim is that $l(\gamma') = \pi$. (This is proved by developing the surface determined by v and γ' onto the northern hemisphere of \mathbb{S}^2 .) Hence, if $l(\gamma) < 2\pi$, then it cannot intersect two disjoint open stars about vertices. So, if V denotes the set of vertices v such that γ intersects the interior of $B_{\pi/2}(\nu)$, then any two element of V are connected by an edge. Since L is a flag complex V must span a simplex of L. But an all right spherical simplex does not contain a closed geodesic. So, $l(\gamma) \geq 2\pi$.

The no \Box condition. When is an all right flag complex L extra large as in Definition 2.2? Since all edge lengths are $= \pi/2$, all closed geodesic of length 2π must correspond to an "empty 4-circuit" in L, i.e., a 4-circuit in L so that there is no other edge of L connecting a pair of diagonally opposite edges in the 4-circuit. So, the answer is simple: L cannot contain any empty 4-circuit. (In [76] this is called the no \Box condition.) Proposition 2.1 then yields the following.

Lemma 2.10. A cube complex can be given a piecewise hyperbolic structure with curvature ≤ -1 if and only if the link of each vertex is a flag complex satisfying the no \Box condition.

2.4 Some first examples of NPC cube complexes

2.4.1 Products

Any graph is a 1-dimensional NPC cube complex. Hence, any tree is a CAT(0) cube complex (by Theorem 2.2 repeat from earlier (i)).

10

 $G = \Omega?$

2.4 Some first examples of NPC cube complexes

A product of graphs is naturally an NPC cube complex. To see this, suppose $X = G_1 \times \cdots \times G_n$ is a product of graphs. The cubes in X correspond to collections $\{c_1, \ldots, c_n\}$, where c_i is either an edge or a vertex of G_i . The corresponding cube $\Box(c_1, \ldots, c_n)$ is isometric to $c_1 \times \cdots \times c_n$. Its dimension is the number of c_i which are edges. If $v = (v_1, \ldots, v_n)$ is a vertex of X, then its link is the *n*-fold join:

$$Lk(v,X) = \underset{i=1}{\overset{i=n}{*}} Lk(v_i,G_i)$$
(2.6)

Noting that the link of a vertex in a graph is a discrete space, we see that Lk(v,X) is a simplicial complex in which each (k-1)-simplex is a join of k points. With the metric of a spherical join, each such simplex is all right. By Lemma 2.8 (ii), Lk(v,X) is a flag complex. By Gromov's Lemma 2.3, X is an NPC cube complex. A similar argument gives the following result.

Proposition 2.2. The product of cube complexes is a cube complex. If $X = X_1 \times \cdots \times X_n$ is such a product and each X_i is NPC, then X is NPC.

Remark 2.3. One can prove directly, without mentioning the link condition, that a product of NPC spaces is NPC. (See [23, Exercise 1.9 (c), p. 163 and Examples 1.15(3), p. 167].)

Example 2.6. (The cubical complex P_L). Given a finite simplicial complex L with vertex set I, in section 3.1 of Chapter 3 we define a subcomplex of the cube $[-1,1]^I$ such that the link at each vertex is L. It follows from Gromov's Lemma that if L is a flag complex, then P_L is NPC.

2.4.2 Alternating link complements

2.4.3 Square complexes

Higman groups. In Serre's book. Paper by Martin Burger-Mozes. Wise

2.4.4 Configuration spaces of a graph

Given a space Y and a natural number $n \in \mathbb{N}$, the *configuration space* of n ordered points in Y is the space of n-tuples $(y_1, \ldots, y_n) \in Y^n$ such that $y_i \neq y_j$ for $1 \leq i < j \leq n$, in other words, it is the complement in Y^n of the fat diagonal. When Y is a cell complex we shall often remove a neighborhood of the fat diagonal instead. For example, suppose Y is a connected graph G. Define $\mathcal{C}_n(G)$ to be the subcomplex of G^n consisting of all cubes of the form $\Box(c_1, \ldots, c_n)$, where $\{c_1, \ldots, c_n\}$ is a collection of pairwise disjoint cells of G. In other words, $\mathcal{C}_n(G)$ is the union of all cubes in G^n which do not meet the fat diagonal. After replacing G by a suitable subdivision, it is easy to see that \mathcal{C}_n is a deformation retract of the complement of the fat diagonal in G^n . So, $\mathcal{C}_n(G)$ is a model for the configuration of n ordered points in G.

The symmetric group Σ_n acts freely and cellularly on $\mathcal{C}_n(G)$. The orbit space $\mathcal{C}_n(G)/\Sigma_n$ is denoted by $\overline{\mathcal{C}}_n(G)$ and is called the *unordered configuration space of* G. Since Σ_n acts freely on $\mathcal{C}_n(G)$, the unordered configuration space is also a cube complex. A *k*-cube in $\overline{\mathcal{C}}_n(G)$ corresponds to a set $\{x_1, \ldots, x_k, u_1, \ldots, u_{n-k}\}$, where the x_i are edges of G and the u_i are vertices. The cube is denoted $\Box\{x_1, \ldots, x_k, u_1, \ldots, u_{n-k}\}$.

put in bibliography

 Γ instead of G?

reference [132] Świ

2 Polyhedral preliminaries

Proposition 2.3. The cube complexes $\mathcal{C}_n(G)$ and $\overline{\mathcal{C}}_n(G)$ are NPC.

Proof. After subdividing we may assume that *G* is a simplicial graph. Since $C_n(G)$ is a covering space of $\overline{C}_n(G)$, it suffices to consider $C_n(G)$. Put $\mathcal{C} = C_n(G)$. According to Gromov's Lemma 2.3, we only need show that the link of each vertex of \mathcal{C} is a flag complex. A vertex *x* of \mathcal{C} has the form, $x = (x_1, \ldots, x_n)$, where $\{x_1, \ldots, x_n\}$ is a collection of *n* disjoint vertices in *G*. By (2.6), $Lk(x, G^n)$ is the join $L_1 * \cdots * L_n$, where $L_i = Lk(x_i, G)$. since \mathcal{C} obtained from G^n by deleting certan open cells, $Lk(x, \mathcal{C})$ is similarly, obtained from $Lk(x, G^n)$. First, notice that certain vertices must be deleted from $Lk(x, G^n)$. This happens whenever x_i and x_j are adjacent vertices in *G*. Indeed, if $c = [x_i, x_j]$ denotes the edge from x_i to x_j , then all faces of the square $c \times c$, except for the vertices (x_i, x_j) and (x_j, x_i) , are deleted. Let u_{ij} be the vertex of L_i corresponding tot the direction from x_i to x_j . (Similarly, $u_{jl} \in L_j$ corresponds to the direction from x_j to x_i .) Let L'_i denote the complement of $\bigcup u_{ij}$ in L_i (a discrete set of vertices). Then $Lk(x, \mathcal{C})$ is a subcomplex of the subjoin,

$$L'_1 * \cdots * L'_n < L_1 * \cdots * L_n = \mathrm{Lk}(x, G^n).$$

To obtain $Lk(x, \mathbb{C})$ we must delete more simplices from $L'_1 * \cdots * L'_n$. For example, suppose the distance in *G* from x_i to x_j is 2. Let *v* be a vertex on the geodesic between them. The square $[x_i, v] \times [x_j, v]$ has vertices $(x_i, x_j), (x_i, v), (v, v), (v, x_j)$. Since interior of this square is deleted; however, the interiors of the edges $[x_i, v]$ and $[x_j, v]$ are not deleted. If $u_{iv} \in L'_i$ and $u_{jv} \in L'_j$ are the directions correspondin to these edges , then the interior of the 1-simplex $u_{iv} * u_{jv}$ is deleted from $L'_i * L'_j$. Similarly, if y_0, \ldots, y_k are distinct vertices in $\{x_1, \ldots, x_n\}$ and each is another vertex *v* adjacent to all of them (as in Figure ??), and if u_{iv} are corresponding vertices of $Lk(x, \mathbb{C})$ then the interior of the *k*-simplex $u_{0v} * \cdots * u_{kv}$ is deleted. However, the point is that the interior of each positive-dimensional subface of $u_{0v} * \cdots * u_{kv}$ is also deleted. Since $L'_1 * \cdots * L'_n$ is a join, it is a flag complex (by Lemma 2.8 (ii)). The simplicial complex $Lk(x, \mathbb{C})$ is obtained from this by deleting open simplices and each edge of such a simplex is also deleted, so there are no "empty simplices". It follows that $Lk(x, \mathbb{C})$ is a flag complex.

The fundamental groups of $\mathcal{C}_n(G)$ and $\overline{\mathcal{C}}_n(G)$ are called, respectively, the *pure braid group* and *braid group* of G and denoted by $PB_n(G)$ and $B_n(G)$, [39]. (See also [2], [72], [132].)

The symmetric product of a graph. For any space *Y*, the symmetric group Σ_n acts on the *n*-fold product Y^n . The orbit space Y^n/Σ_n is called the *n*-fold symmetric product of *Y*.

2.4.5 Graph braid groups embed in RAAGs

later, after RAAGs The discussion follows [39].

Given a graph G, define another graph $\Delta(G)$. The vertex set of $\Delta(G)$ is the edge set of G. Given $e \in Edge(G)$, denote by \overline{e} the corresponding vertex of $\Delta(G)$. Vertices \overline{e} and \overline{f} span an edge of $\Delta(G)$ if and only if e and f are disjoint.

Let $A_{\Delta(G)}$ be the RAAG associated to $\Delta(G)$ and let $BA_{\Delta(G)}$ be the standard model for its classifying space. The is a 1-cell of $BA_{\Delta(G)}$ for each vertex of $\Delta(G)$. Choose an orientation for edge of G. Define a cubical map $\Phi : \overline{\mathbb{C}}_n(G) \to BA_{\Delta(G)}$ by sending the *k*-cube $\Box \{x_1, \ldots, x_k, u_1, \ldots, u_{n-k}\}$ to the cube in $BA_{\Delta(G)}$ spanned by the edges corresponding to $\overline{x}_1, \ldots, \overline{x}_k$ while respecting the orientations of all edges. (The image of the *k*-cube in $BA_{\Delta(G)}$ is a *k*-torus.) It is proved in [39, Theorem 2] that Φ is a local isometry and hence, is π_1 -injective.) This gives the following. **Proposition 2.4.** (Crisp-Weist [39]). *The map* Φ , *defined above, induces an injection,* $B_n(G) \hookrightarrow A_{\Delta(G)}$. In other words, any graph braid group is a subgroup of a RAAG.

Chapter 3 Right-angled complexes and groups

Many examples of cubical complexes can be defined using the notion of a "polyhedral product" with respect to a simplicial complex L. Under suitable conditions the universal cover of such a polyhedral product often will be CAT(0).

The fundamental group of a polyhedral product is the "graph product" of the fundamental groups of its factors.

Dual cones, dual of a simplicial complex, strict fundamental domain

3.1 Polyhedral products

Let *L* be a simplicial complex with vertex set *I*. The poset of simplices (including the empty simplex) is denoted by S(L). If σ is a simplex of *L*, then $I(\sigma)$ denotes its vertex set.

Suppose that $\mathbf{A} = \{(A_i, B_i)\}_{i \in I}$ is a collection of pairs of spaces with base points $*_i$ chosen in B_i .

Definition 3.1. The *polyhedral product* of **A** with respect to *L* is the subset \mathbf{A}^L of the product $\prod_{i \in I} A_i$ consisting of the points $(x_i)_{i \in I}$ satisfying the following two conditions:

(a) $x_i = *_i$ for all but finitely many *i*, (b) $\{i \in I \mid x_i \notin B_i\}$ is a simplex of *L*.

For pairs M < M' of subcomplexes of L, the base points determine natural inclusions $\mathbf{A}^M \hookrightarrow \mathbf{A}^{M'}$. When L is finite, the polyhedral product inherits its topology from the product $\prod_{i \in I} A_i$. In general \mathbf{A}^L acquires the topology of the direct limit, $\lim \mathbf{A}^M$, as M runs through the finite subcomplexes.

One can also view \mathbf{A}^{L} as a union of products as follows. For each simplex σ in S(L) put

$$\mathbf{A}^{\sigma} = \prod_{i \in \operatorname{Vert} \sigma} A_i \times \prod_{i \notin \operatorname{Vert} \sigma} B_i$$

So, if σ is the empty simplex, then $A^{\sigma} = \prod B_i$. Also, note that if each $B_i = *_i$, then A^{σ} is just the product of the A_i , $i \in \text{Vert } \sigma$. Then

$$\mathbf{A}^{L} = \bigcup_{\sigma \in \mathcal{S}(L)} A^{\sigma} . \tag{3.1}$$

15

The topology on \mathbf{A}^{L} also can be explained using (3.1). If Δ denotes the full simplex on I, then \mathbf{A}^{Δ} is the subset of the product of the A_{i} satisfying (a). So, \mathbf{A}^{Δ} is the union of all \mathbf{A}^{σ} over all finite subsets of $\sigma \leq I$. Give \mathbf{A}^{Δ} the direct limit topology and $A^{L} \leq A^{\Delta}$ is the induced topology.

The main examples of this book are when:

- (1) each A_i is the interval $[0, 1], B_i = *_i = 1$,
- (2) each A_i is the interval $[-1, 1], B_i = \partial [-1, 1] = \{\pm 1\}$ and $*_i = 1$,
- (3) each B_i is a discrete space E_i and $A_i = \text{Cone } E_i$ and $*_i \in E_i$
- (4) each $A_i = S^1$ and $B_i = *_i = 1$.

In (3) Cone E_i means the cone on E_i , i.e., it is the quotient space $(E_i \times [0,1])/\sim$. For example, if $E_i = \{\pm 1\}$, then Cone E_i can be identified with [-1,1]. The *cone point* of E_i is the point 0_i corresponding to $0 \in [0,1]$ Choose an element $*_i \in E_I$ to be the base point. Let **Cone E** denote the *I*-tuple (Cone $E_i, E_i, *_i)_{i \in I}$.

Denote the polyhedral products in (1),(2), (3), (4) by K_L , P_L , Z_L , and \mathbb{T}^L , respectively, i.e.,

$$K_L = (0,1], 1)^L, (3.2)$$

$$P_L = ([-1,1], \{\pm 1\})^L, \tag{3.3}$$

$$Z_L = (\mathbf{Cone}\,\mathbf{E},\mathbf{E})^L,\tag{3.4}$$

$$\mathbb{T}_L = (S^1, 1)^L. \tag{3.5}$$

When the pairs (A_i, B_i) are independent of *i* so that $(A_i, B_i) = (A, B)$, we have writen simply $(A, B)^L$ instead of using boldface (for example, this is the case in (3.2), (3.3) and (3.5)).

3.1.1 The cube complexes K_L , P_L , Z_L , and \mathbb{T}_L

Since polyhedral products are subspaces of products, P_L is a cubical subcomplex of the cube $[-1,1]^I$ while \mathbb{T}_L is a subcomplex of the torus, $(S^1)^I$. Regarding S^1 as a 1-cube complex with one edge and one vertex, the product $(S^1)^I$ becomes a cube complex and \mathbb{T}_L is a cubical subcomplex. The cone, Cone E_i , is the union of intervals, one for each point of E_i , glued together at the cone point 0_i . So, Cone E_i is a 1-cube complex. Hence, the product $\prod_{i \in I} \text{Cone } E_i$ is a cube complex. It follows that the polyhedral product Z_L , defined by (3.4), is also a cube complex. Each of these cube complexes, K_L , P_L , Z_L and \mathbb{T}_L has the same dimension, namely, $1 + \dim(L)$.

The complex P_L . Let $\mathbb{C}_2 = \{\pm 1\}$ be the cyclic group of order 2 and denote by $(\mathbb{C}_2)^I$ (or by $\sum_{i \in I} \mathbb{C}_2$ the direct sum of I copies of \mathbb{C}_2 . Similarly, $[-1,1]^I$ is the set of all (t_i) in the product satisfying condition (a) in Definition 3.1. The group $(\mathbb{C}_2)^I \sim [-1,1]^I$ as a reflection group. Each element of I can be regarded as a standard basis vector for the euclidean space \mathbb{R}^I . So, each (k-1)-face σ of the full simplex on I determines a k-dimensional subspace \mathbb{R}^σ spanned by the vertex set $I(\sigma)$ of σ . The cube complex P_L is stable under the $(\mathbb{C}_2)^I$ -action on $[-1,1]^I$. Any k-dimensional face of $[-1,1]^I$ is parallel to some \mathbb{R}^σ . We denote such a face by \Box^σ . Moreover, given σ the corresponding set of k-cubes forms a single $(\mathbb{C}_2)^I$ -orbit which we denote by orb (\Box^σ) . In other words, orb $\Box^\sigma \cong (\mathbb{C}_2)^{I-I(\sigma)} \times [-1,1]^{I(\sigma)}$. Thus,

$$P_L = \bigcup_{\sigma \in \mathcal{S}(L)} \operatorname{orb} \Box^{\sigma}.$$
(3.6)

16

 Z_L ?

3.1 Polyhedral products

So, orb \Box^{\emptyset} means the $(\mathbb{C}_2)^I$ -orbit of a vertex, i.e., orb $\Box^{\emptyset} = \{\pm 1\}^I$.

Example 3.1.

(i) If L consists of n vertices and no higher dimensional cells, then P_L is the 1-skeleton of an n-cube.

- (ii) If $L = \partial \Delta^n$ is the boundary complex of an *n*-simplex, then P_L is the boundary complex of an (n+1)-cube, i.e., $P_L \cong S^n$
- (iii) Suppose *L* is an *m*-circuit (that is, a circle triangulated as the boundary of an *m*-gon). The square complex P_L is a closed 2-manifold. (since the link of each vertex is S^1 and the link of each edge is S^0). It is easily seen to be connected and orientable. Its Euler characteristic is $2^m m2^{m-1} + m2^{m-2} = 2^m(1 m/4)$. This example appeared already in Coxeter's 1938 paper [37, p. 57]. (Thanks to Alex Suciu for pointing out this reference.) If m = 4, P_L is a flat 2-torus.

If the 1-skeleton L^1 of L is not a complete graph, then P_L is not simply connected. In general, $\pi_1(P_L)$ depends only on L^1 (cf. [51]).

The complex Z_L . The polyhedral product, $Z_L := (\text{Cone E})^L$, defined by (3.4), is a generalization of the previous example P_L of a polyhedral product of intervals.

Example 3.2. Suppose, for i = 1, 2, that E_i is a finite set of m_i points. If L is a 1-simplex σ^1 , then the polyhedral product $Z_{\sigma^1} = \text{Cone } E_1 \times \text{Cone } E_2$ is a square complex. Each square has the form $\text{Cone } e_1 \times \text{Cone } e_2$, where $e_i \in E_i$. This square complex also has the structure of cone, namely the cone on $Z_{\partial\sigma^1} = (\text{Cone } E_1 \times E_2) \cup (E_1 \times \text{Cone } E_2)$. The space $Z_{\partial\sigma^1}$ can be identified with the join $E_1 * E_2$ (a complete bipartite graph). (See Figure ??.)

More generally, suppose σ is an (n-1)-simplex with vertex set $I = \{1, ..., n\}$ and that E_i is a finite set of cardinality m_i . Then Z_{σ} is the product, Cone $E_1 \times \cdots \times$ Cone E_n , which is a cube complex. Moreover, $Z_{\partial\sigma}$ is homeomorphic to the join, $E_1 * \cdots * E_n$. If $I(\sigma)$ is the vertex set of σ regarded as a simplicial complex with no edges, then $Z_{I(\sigma)}$ is a graph – the star of a vertex lying above $i \in I$ is isomorphic to Cone E_i .

The fundamental chamber K_L . Suppose each E_i is a singleton, necessarily the base point. The polyhedral product Z_L becomes

$$K_L := (\operatorname{Cone} *_i, *_i)^L. \tag{3.7}$$

Since $(\text{Cone}_{*_i}, *_i) \cong ([0, 1], 1)$, K_L is naturally a cube complex. Since each cube contains the base point $* = (*_i)_{i \in I}$, K_L is a cone – in fact it can be identified with the cone on the barycentric subdivision of L. Note that K_L is a subcomplex of Z_L . In particular, it is a subcomplex of P_L . We have that P_L is a subcomplex of $[-1,1]^I$ and K_L can be identified with the intersection $P_L \cap [0,1]^I$. So, it is a strict fundamental domain, in the sense defined below, for the $(\mathbb{C}_2)^I$ -action on P_L . For this reason, K_L is called the *fundamental chamber*.

To simplify notation, put $K = K_L$ and for each vertex $i \in I$, define the *mirror* K_i to be the intersection of K with the hyperplane $x_i = 0$. Similarly, for each simplex $\sigma \in S(L)$, put

$$K_{\sigma} = \bigcap_{i \in \operatorname{Vert} \sigma} K_i = \{ x \in K \mid x_i = 0 \text{ for all } i \in I(\sigma) \}.$$

later in links

Why say the last sentence?

 \mathbb{T}_L , the standard classifying space for a RAAG. The cube complex \mathbb{T}_L , defined by (3.5), is a polyhedral product of circles. If $L = \Delta$ is an (n-1)-simplex, then \mathbb{T}_Δ is the *n*-torus, while $\mathbb{T}_{\partial\Delta}$ is the (n-1)-skeleton of the *n*-torus. If *L* is a flag complex, then \mathbb{T}_L is the *Salvetti complex*. Thus, \mathbb{T}_L is a subcomplex of \mathbb{T}^Δ where

some kind of examples

 Δ denotess the full simplex on *I*. In other words, \mathbb{T}_L is the union of subtori \mathbb{T}_{σ} , where $\sigma \in S(L)$, cf. (3.1). (In some contexts, we might define the Salvetti complex to be the covering space of \mathbb{T}_L with group of deck transformations the RACG associated to *L*.)

Definition of a strict fundamental domain.

Definition 3.2. Suppose G is a discrete group acting on a space X. A closed subspace Y < X is a *strict fundamental domain* if it intersects each G-orbit in exactly one point.

do we need to assume proper? We leave the proof of the next lemma as an exercise for the reader.

Lemma 3.1. Suppose $G \curvearrowright X$ with strict fundamental domain Y. Let $\pi : X \to X/G$ be projection to the orbit space and let $\overline{\pi} = \pi|_Y$. Then $\overline{\pi} : Y \to X/G$ is a homeomorphism.

One can reconstruct the *G*-action on *X* from the group *G*, the strict fundamental domain *Y*, and the knowledge of the isotropy subgroups G_y at each point $y \in Y$ (where $G_y = \{g \in G \mid gy = y\}$). Define an equivalence relation \sim on $G \times Y$ by

$$(g,y) \sim (g',y') \iff y = y' \text{ and } g^{-1}g' \in G_y$$
. (3.8)

The *basic construction* is the *G*-space D(G,Y) defined by $D(G,Y) = (G \times Y) / \sim$. Let [g,y] denote the equivalence class of (g,y).

Lemma 3.2. (Properties of the basic construction).

(i) The natural map y → [1,y] from Y into D(G,Y) is an embedding (which we regard as an inclusion).
(ii) G ∩ D(G,Y) and Y is a strict fundamental domain.

Links of vertices. One of the main features of P_L is that the link of each of its vertices can be identified with L. We state this as follows.

Lemma 3.3. For each vertex v of P_L , $Lk(v, P_L) = L$.

Consider the 1-cube complex Cone E_i . Let 0_i denote the cone point. There are two types of vertices in Cone E_i : the cone point 0_i is a vertex as is each element $e \in E_i$. As for links, $Lk(0_i, Cone E_i) = E_i$, while for each $e \in E_i$, $Lk(e, Cone E_i)$ is a point. Next consider the general case, $Z_L := (Cone E)^L$. Any vertex v of Z_L has the form $v = (v_i)_{i \in I}$, where v_i is a vertex of Cone E_i , i.e., either $v_i \in E_i$ or $v_i = 0_i$. Put

$$\tau(v) = \{i \in | v_i = 0_i\}.$$
(3.9)

So, $\tau(v)$ is a simplex of *L* (possibly the empty simplex). This proves the following lemma which describes links in *Z*_L.

Lemma 3.4. For a vertex v of Z_L let $\tau(v)$ be defined by (3.9). The link of v in Z_L is the join:

$$Lk(v, Z_L) = Lk(\tau(v), L) * *_{i \in I(\tau(v))} E_i.$$
(3.10)

In particular, if each v_i lies in some E_i , then $Lk(v, Z_L) = L$.

earlier Remark 3.1. If each E_i consists of two points, then $Z_L = P_L$. However, the cubical structure on P_L is coarser then the one we get by identifying it with Z_L . The reason is that if we identify [-1, 1] with Cone $\{\pm 1\}$, then we should subdivide [-1, 1] into two intervals, [-1, 0] and [0, 1]. Thus, when thought of as Z_L , each *n*-cube in P_L is subdivided into 2^n cubes. 3.1 Polyhedral products

earlier **Definition 3.3.** (Compare Definition 3.1 and formula (3.1)). Suppose $\{X_i\}_{i \in I}$ is a collection of spaces indexed by the set *I* of verices of a simplicial complex *L*. For any simplex σ of *L*, let $X(\sigma)$ denote the join $*_{i \in I(\sigma)} X_i$. The *polyhedral join over L* of $\{X_i\}_{i \in I}$ is defined by

$$*_L X_i := \bigcup_{\sigma \in \mathcal{S}(L)} X(\sigma).$$
(3.11)

Lemma 3.5. The link of each vertex v in \mathbb{T}_L is a polyhedral join of 0-spheres, $Lk(v, \mathbb{T}_L) = *_L S^0$.

Lemma 3.6. The polyhedral join of flag complexes is a flag complex.

The next result follows from Lemma 2.3 (Gromov's Lemma).

Theorem 3.1. The cube complexes K_L , P_L , Z_L and \mathbb{T}_L are NPC if and only if L is a flag complex.

Proof. The issue is to show that links of vertices in P_L , Z_L and \mathbb{T}_L are flag complexes if and only if L is a flag complex. So, suppose L is a flag complex. Since the link of each vertex in P_L is L (by Lemma 3.3), this is immediate for P_L . By Lemma 3.4, the link of a vertex v of Z_L has the form $Lk(\tau(v), L) * *_{i \in I(\tau(v))} E_i$. By part (iii) of Lemma 2.8, $Lk(\tau(v), L)$ is a flag complex and by part (ii), so is $*_{i \in I(\tau(v))} E_i$. Another application of part (ii) gives that the join, $Lk(\tau(v), L) * *_{i \in I(\tau(v))} E_i$, is a flag. complex. The complex K_L is the special case of Z_L where each E_i is a singleton. As for \mathbb{T}_L , if L is a flag complex, then, by Lemma 3.6, so is $*_L S^0$; hence, by Lemma 3.5, the link of each vertex in \mathbb{T}_L is a flag complex.

Universal covers. More important for us than the cube complexes P_L , Z_L and \mathbb{T}_L will be their universal covers, denoted \tilde{P}_L , \tilde{Z}_L and \mathbb{T}_L , respectively. If L is a flag complex, then by the Cartan-Hadamard Theorem (Theorem 2.2 (i)), these universal covers are CAT(0) cube complexes. The complex \tilde{P}_L is the *Davis complex* for the right-angled Coxeter group associated to L, when each E_i has a least two points the complex \tilde{Z}_L is a *right-angled building*, and \mathbb{T}_L is the universal cover of the Salvetti complex for the right-angled Artin group associated to L. In order to describe the groups which act on these complexes we need the notion of a "graph product" of groups, developed in the next subsection.

Here is another straightforward application of Gromov's Lemma.

Lemma 3.7. For each $i \in I$ suppose A_i is an NPC cube complex and its basepoint $*_i \in A_i$ is a vertex of A_i . Put $\mathbf{A} = (A_i, *_i)$. Then \mathbf{A}^L is naturally a cube complex. Moreover, it is NPC if and only if L is a flag complex.

3.1.2 Polyhedral products of NPC cube complexes

Theorem 3.2. A polyhedral product over a flag complex L of NPC cube complexes is an NPC cube complex.

Examples? RACGs. Start with a RACG and replace each generator with another RACG, eg., the infinite dihedral group D_{∞} (cf..

3.1.3 Graph products of groups

Suppose $\{G_i\}_{i \in I}$ is a collection of groups indexed by the vertex set *I* of a simplicial graph L^1 . Their graph product, denoted $\prod_{I} G_i$, is the quotient of the free product of the G_i by the normal subgroup generated by all

commutators of the form $[g_i, g_j]$, where $\{i, j\} \in \text{Edge } L^1$, $g_i \in G_i$ and $g_j \in G_j$. (These were first studied by Droms [65].)

As examples, if $\text{Edge}L^1 = \emptyset$, then the graph product is the free product, while if L^1 is a complete graph, then the graph product is the direct sum: $\prod_{L^1} G_i = \sum_{i \in I} G_i$. (The *direct sum* $\sum_{i \in I} G_i$ is the subset of the direct product where only finitely many coordinates are not equal to the identity element of G_i .) There is a canonical epimorphism $p : \prod_{L^1} G_i \to \sum_{i \in I} G_i$.

Given a system of groups, the notion of their "direct limit" is defined, for example, on page 1 of [128]. The graph product alternatively can be described as the direct limit of certain system of groups indexed by the poset of cells in L^1 :

$$G_i, i \in I, \quad G_i \times G_j, \{i, j\} \in L^{(1)},$$

together with the natural inclusions $G_i \hookrightarrow G_i \times G_j$. The graph product $\prod_{L^1} G_i$ is then the direct limit of this system of groups. (In Chapter 5 this system of groups will be presented as a nice example of a "simple complex of groups".)

examples Example 3.3. (RACGs and RAAGs). If each G_i is the cyclic group of order two, then the graph product is called the *right-angled Coxeter group* associated to L^1 . (The term "right-angled Coxeter group" will often be abbreviated to RACG.) If each G_i is the infinite cyclic group, then $\prod_{L^1} \mathbb{Z}$ is the *right-angled Artin group* (abbreviated to RAAG) associated to L^1 .

Let $G = \sum_{i \in I} G_i$ denote the direct sum. Suppose $G_i \curvearrowright E_i$. Then $G \curvearrowright (\text{Cone E})^{\Delta}$, where Δ means the full simplex on I. The polyhedral product $Z_L = (\text{Cone E})^L$ is stable under the G-action. Choose base points $*_i \in E_i$ so that the I-tuple $* = (*_i)_{i \in I}$ is a base point for the polyhedral product Z_L . Let K_L be the subcomplex of Z_L corresponding to this choice of base points. The cone point 0_i of Cone E_i lies in K_L and its isotropy subgroup in G is G_i . Let $p : \tilde{Z}_L \to Z_L$ be the universal cover and let $\tilde{*}$ be a lift of * in \tilde{Z}_L . Since K_L is simply connected, each component of $p^{-1}(K_L)$ is mapped homeomorphically onto K_L . Let \tilde{K} denote the component containing $\tilde{*}$ and let $\tilde{0}_i$ denote the inverse image of 0_i in \tilde{K} .

Lemma 3.8. With notation as above, let $G = \sum G_i$ be the direct sum and let $\Gamma = \prod_{L^1} G_i$ be the graph product. Let $Z_L = (\text{Cone E})^L$ be the polyhedral product and let \tilde{Z}_L be its universal cover. For each $i \in I$, suppose the G_i -action on E_i is simply transitive. Then Γ is the group of all lifts of the G-action to \tilde{Z}_L . Hence, there is a short exact sequence,

$$1 \to \pi_1(Z_L) \to \Gamma \to G \to 1$$
.

Fix *Proof.* Let *H* be the group of all lifts of the *G*-action to \tilde{Z}_L and let $\varphi : H \to$ be the canonical projection. The subgroup H_i consisting of all lifts which fix the point \tilde{c}_i projects isomorphically onto G_i . If $\{i, j\} \in L^{(1)}$, then Cone $E_i \times \text{Cone } E_j$ is a subcomplex of Z_L and there is an isomorphic copy of this subcomplex in \tilde{K} containing $\tilde{*}$. In particular, this subcomplex contains the square $[\tilde{*}_i, \tilde{c}_i] \times [\tilde{*}_j, \tilde{c}_j]$. The subgroup $H_i \times H_j$ fixes the point $\tilde{c}_i \times \tilde{c}_j$ and projects isomorphically to $G_i \times G_j$. So, $H_i \times H_j$ is the group of all lifts of $G_i \times G_j$ which fix $\tilde{c}_i \times \tilde{c}_j$. In particular, this shows that whenever $\{i, j\}$ is an edge of L^1 , the subgroups H_i and H_j commute. It follows from this and the universal property of the direct limit (cf. **??**), that the natural inclusions, $G_i \to H_i \hookrightarrow H$ and $G_i \times G_j \to H_i \times H_j \hookrightarrow H$, with $|[i, j] \in \text{Edge } L^1$, extend to a homomorphism $\phi : G \to H$ from the graph product to the group of lifts. Since G_i^I acts simply transitively on $(E_i)^I$, H acts simply transitively on $p^{-1}((E_i)^I)$. A straighforward argument implies that $\phi : G \to H$ is an isomorphism.

Example 3.4. Let *L* be a finite dimensional flag complex on vertex set *I*. For each $i \in I$, suppose G_i is a nontrivial finite group. Let $Z_L = (\text{Cone } G_i, *_i)^L$ be the polyhedral product and \tilde{Z}_L the associated RAB. Let Γ be the graph product. Then Γ acts properly, each finite subgroup is conjugate to a finite subgroup of $G = \sum_{i \in I} G_i$. It follows that \tilde{Z}_L is the universal proper Γ -space $\underline{E}G$.

20

3.1 Polyhedral products

Proposition 3.1. Suppose *L* is a flag complex. Then the polyhedral product $(BG_i, *_i)^L$ is the classifying space for the graph product $\Gamma = \prod_{I=1}^{I} G_i$.

Proof. Let $p_i: EG_i \to BG_i$ be the universal cover. The induced map of polyhedral products, $(EG_i, p^{-1}(*_i))^L \to (BG_{i,i})^L$, is then a covering projection. Since $(EG_i, p_i^{-1}(*_i))$ is contractible, $(EG_i, p_i^{-1}(*_i))$ and $(\text{Cone } G_i, G_i)^L$ are homotopy equivalent as pairs; so, the polyhedral products $Z_L = (\text{Cone } G_i, G_i)^L$ and $Y_L = (EG_i, p_i^{-1}(*_i))^L$ also are homotopy equivalent. Hence, their universal covers \tilde{Z}_L and \tilde{Y}_L are homotopy equivalent. Moreover, the group of lifts of the $(\sum_{i \in I} G_i)$ -action on Y_L to \tilde{Y}_L is also identified with the graph product Γ . So, the composition $\tilde{Y}_L \to Y_L \to (BG_i, *_i)^L$ is the universal cover and Γ is the group of deck transformations. By Theorem 3.1, since L is a flag complex, Z_L is NPC and hence, by the Cartan-Hadamard Theorem, \tilde{Z}_L is contractible. (Alternatively, it is proved in [49] that \tilde{Z}_L is the standard realization of a right-angled building (a "RAB") and hence, is contractible (since the standard realization of any RAB is contractible, see ??). Since \tilde{Y}_L is homotopy equivalent to \tilde{Z}_L , it also is contractible. Hence, $B\Gamma \sim (BG_i, *_i)^L$.

Example 3.5. (*Standard classifying spaces for* RAAGs and RACGs, cf. Examples 3.3). Suppose each G_i is infinite cyclic so that $BG_i = S^1$. For each simplex $\sigma \in S(L)$, the polyhedral product $(S^1, *_i)^{\sigma}$ is a torus with basis Vert σ . By Proposition 3.1, when L is a flag complex, $\mathbb{T}_L = (S^1, *_i)^L$ is the classifying space BA_L for the RAAG associated to L^1 . (The NPC cube complex \mathbb{T}_L from (3.5) in Subsection '3.1.1 is the Salvetti complex of A_L .)

If each G_i is \mathbb{C}_2 , then $B\mathbb{C}_2 = \mathbb{R}P^{\infty}$. When *L* is a flag complex, $(\mathbb{R}P^{\infty}, *_i)^L$ is the classifying space BW_L for the RACG associated to L^1 .

Proposition 3.1 suggests that the fundamental group of an arbitrary polyhedral product with each $B_i = *_i$ is a graph product of the fundamental groups of the factors. In fact, this is proved in [51, Theorem 2.18, p. 248]. We state this as the following proposition without giving the proof.

Proposition 3.2. ([51]). Suppose $\{A_i\}_{i \in I}$ is a collection of path connected CW complexes indexed by the vertex set of a simplicial complex L. Let $*_i$ be a base point and put $\mathbf{A} = (A_i, *_i)$ and $G_i = \pi_1(A_i, *_i)$. Then $\pi_1(\mathbf{A}^L)$ is the graph product $\Gamma = \prod_{L^1} G_i$. (In particular, $\pi_1(\mathbf{A}^L)$ depends only on the 1-skeleton of L.)

Actually a more general fact is proved in [51]: if the B_i is only required to be a subspace of A_i , then $\pi_1(\mathbf{A}^L)$ is an appropriately defined "generalized graph product".

3.1.4 Graph wreath products

Suppose L^1 is a simple graph on vertex set I (which need not be finite) and $\Gamma = \prod_{L^1} G$ denotes the graph product. Let J be a group of automorphisms L^1 . The J-action on L^1 naturally induces a J-action on Γ via group automorphisms. The semidirect product $\Gamma \rtimes J$ is called the (restricted) graph wreath product of the G. (The terminology was introduced in [91].) When L^1 is the complete graph, Γ is the direct sum $\sum_{i \in I} G$ and $\Gamma \rtimes J$ is the ordinary restricted wreath product. ("Restricted" refers to the fact that we are using the direct sum rather than the direct product.)

The graph wreath product $\Gamma \rtimes J$ acts naturally on 1) the universal cover \tilde{Z}_L of the polyhedral product $Z_L = (\text{Cone } G, G)$; it also acts on 2) the classifying space $E\Gamma$ for Γ . To see the action in case 1), first consider

Examples?

Someplace else

the action of the ordinary wreath product $\sum G \rtimes J$ on the weak product, $(\operatorname{Cone} G, G)^{\Delta(I)}$, where $\Delta(I)$ is the full simplex on *I*. The subspace $(\operatorname{Cone} G, G)^L$ is stable under the action of $\sum G \rtimes J$. The group of lifts of this action to \tilde{Z}^L is the graph wreath product $\Gamma \rtimes J$. (To see this it is sufficient to note that since *J* fixes the base point of $(\operatorname{Cone} G, G)^L$, its group of lifts of *J* at a base point of \tilde{Z}_L projects isomorphically to *J*). Suppose $M = L^J$ denotes the subcomplex of *L* fixed by *J*. It follows that $(\tilde{Z}_L)^J = (\tilde{Z}_M)$: in particular, since this fixed point set contains a given lift of the base point, it is nonempty. It follows that the action of $\Gamma \rtimes J$ on \tilde{Z}_L is proper if and only *J* is a finite group.

One can check case 2) by following the construction in the proof of Proposition **??**. The covering space $(EG, p^{-1}(*))^L \to (BG, *)^L$ corresponds to the homomorphism $\pi_1((BG, *)^L) = \Gamma \to \sum G$. The wreath product $\sum G \rtimes J$ acts on $(EG, p^{-1}(*))^L$ and $\Gamma \rtimes J$ is the group of lifts to the universal cover $E\Gamma$. If J is finite, then action of $\Gamma \rtimes J$ on this model of $E\Gamma$ is proper. Since $E\Gamma^J$ is homotopy equivalent to \tilde{Z}_{L^J} which is contractible, $E\Gamma^J$ is also contractible. Hence, when J is finite, $E\Gamma$ is a model for $\underline{E}(\Gamma \rtimes J)$, the universal space for proper actions of the graph wreath product. We record this as the following.

Proposition 3.3. Suppose *J* is a finite group of automorphisms of *L* and that $\Gamma \rtimes J$ is the graph wreath product. Then the universal cover of $(BG,*)^L$ is a model for $\underline{E}(\Gamma \rtimes J)$.

3.1.5 Application: cohomological dimension versus geometric dimension

The geometric dimension of a group G, denoted gd G, is the smallest dimension of a model for BG by a CW complex. If EG denotes the universal cover of BG, then its cellular chin complex $C_* := \{C_i(EG)\}$ is a chain complex of free $\mathbb{Z}G$ -modules. In fact, $C_i(EG)$ is isomorphic to a direct sum of copies of $\mathbb{Z}G$ where there is one summand for each *i*-cell in BG. Since EG is contractible, the chain complex C_* is acyclic. In fact, if we equip it with the augmentation homorphism $C_0 \to \mathbb{Z}$ then C_* gives a free resolution of \mathbb{Z} .

The cohomological dimension of *G*, denoted cd *G* is the shortest length of a resolution of \mathbb{Z} by projective $\mathbb{Z}G$ -modules. (Here one can replace the word "projective" by "free".) It follows that $gd G \ge cd G$. In the 1950 s Eilenberg and Ganea [66] raised the question if these two numbers are equal and in almost all cases they proved that they are equal. More precisely, we have the following.

Theorem 3.3. (The Eilenberg-Ganea Theorem [66]).

- (1) If $\operatorname{cd} G = 1$, then $\operatorname{gd} G = 1$.
- (2) If $\operatorname{cd} G \ge 3$, then $\operatorname{gd} G = \operatorname{cd} G$.
- (3) If $\operatorname{cd} G = 2$, then either $\operatorname{gd} G = 2$ or $\operatorname{gd} G = 3$

Actually statement (1) follows from results of Stallings and Swan that show that groups of cohomological ! are free. The Eilenberg-Ganea Conjecture is the conjecture that we always have equality between gdG and cdG.

Acyclic versus contractible.

Probable counterxamples to the Eilenberg-Ganea Conjecture.

3.2 Reflection groups

There is a well-known, classical theory of groups generated by reflections on the constant curvature spaces, \mathbb{S}^n , \mathbb{E}^n and \mathbb{H}^n . Since much of the intuition for the constructions of polyhedral products in section 3.1, we briefly will give some highlights of this theory here. A fuller account can be found in **??**.

A *dihedral group* is a group generated by two involutions There are two basic examples of geometric actions of dihedral actions to have in mind. First, the group generated by the reflections across an interval in \mathbb{E}^1 gives an action of the infinite dihedral group D_{∞} with fundamental domain the interval. The resulting tessellation of \mathbb{E}^1 by intervals gives it the structure of a 1-cube complex. Second consider the group generated by reflections across two lines through the origin in \mathbb{R}^2 . If the angle between the two lines is π/m the group is the finite dihedral group D_m of order 2m. A strict fundamental domain for the action is the sector bounded by two rays making an angle of π/m . If we restrict the D_m -action to the unit circle \mathbb{S}^1 , a strict fundamental domain is a circular arc of length π/m .

Before considering more of the classical geometric pictures, let us discuss a fairly general situation of a reflection group acting on connected smooth manifold *X*. A *reflection* on *X* is a smooth involution $r: X \to X$ such the its fixed set X^r disconnects *X* into two connected components, the closure of one of these components is called a *half-space* and the fixed set X^r is a *wall*. One sees that each wall is a submanifold of codimension one and that each half-space is a manifold with boundary and a strict fundamental domain for the action of $C_2 (= \langle \rangle)$ on *X*. Suppose that *W* is a discrete group acting properly on *X* and that *W* is generated by the set *R* of all reflections in *W*. Choose a K° denote a connected component of $X - \bigcup_{r \in R} X^r$ and let *K* denote the closure of K° . Then *K* is a *chamber* for the reflection group *W*. One says say that X^r is a *wall of K* if $K \cap X^r$ is codimension zero in X^r . It follows that *K* is the intersection of half-spaces bounded by the walls of *K*. Let *S* be the set of reflections across the walls of *K*. Here are the main facts.

- (1) R is the set of all conjugates of elements of S by elements of W. In particular, since R generates W, S is a set of generators for W.
- (2) The group W is a *Coxeter group*. This means that W has a presentation of the form (S | R) where S is the above set of generators and the relations R consists of all words of the form: s², with s ∈ S and (st)^{m(s,t)}, where {s,t} ranges over all unordered pairs of distinct elements of S and where m(s,t) is the order of st in W (and if m(s,t) = ∞ we omit the relation.) The pair (W,S) is called a *Coxeter system*If all the m(s,t) are equal to 2 or ∞, then W is a right-angled Coxeter group.
- (3) *K* is a strict fundamental domain for the *W*-action (see Definition 3.2). It follows from considering the basic geometric examples on \mathbb{E}^1 and \mathbb{R}^2 that *K* has codimension one faces where the isotropy group is \mathbb{C}_2 . In fact it will follow from **??** below that *K* is a manifold with corners. Hence, *X* together with the *W*-action can be recovered via the basic construction D(W, K) (cf. (3.8) and the following definition).

Next we specialize to the geometric examples where $X = \mathbb{S}^n$, \mathbb{E}^n of \mathbb{H}^n , reflections are isometries and half-spaces are actual geodesically convex half-spaces. Since the fundamental chamber *K* is an intersection of half-spaces, it is a convex polytope. We require *K* to be compact.

- (a) $(X = \mathbb{S}^n)$. Assume *W* acts without global fixed points on \mathbb{S}^n . Then *K* is a spherical simplex. (Actually this is a result in linear algebra, cf. [48, Lemma 6.3.3, p. 78].) This result implies that when $X = \mathbb{E}^n$ or \mathbb{H}^n , the polytope *K* is *simple*, which means that the link of each nonempty face of *K* is a simplex. Similarly, in the case where *X* is a general manifold, it follows that *K* is a manifold with corners.
- (b) $(X = \mathbb{E}^n)$.
- (c) $(X = \mathbb{H}^n)$.

Where should this be in relation to ??? Maybe it stays here

3 Right-angled complexes and groups

3.2.1 Davis complex

• Geometric examples with simple polytopes

3.2.2 Semidirect products

subsumed in graph wreath products

3.2.3 The reflection group trick

This is method for showing any group of type F is a retract of the fundamental group of a closed aspherical manifold. In fact, any group of type FH is a retract of a Poincaré duality group.

Corollary 3.1. (Mess, Weinberger, Sapir)

- (i) The fundamental group of a closed aspherical manifold need not have solvable word problem.
- (ii) The fundamental group of a closed aspherical manifold need not be residually finite.
- (iii) [100] The fundamental group of a closed aspherical manifold need not have solvable word problem.

(iv) It can contain a "Tarski monster" ([123])

3.3 Right-angled buildings

A product of trees is a RAB (provided each vertex has degree ≥ 2). So, the universal cover of a product of graphs without degree 1 vertices is a RAB.

The RABs of form \tilde{Z}_{Δ} are very symmetric RABs. The \tilde{Z}_L are highly symmetric in that automorphism group is chamber-transitive.

The space Z_L is usually not simply connected. (In fact, if each E_i has at least two elements, then Z_L is simply connected if and only if L^1 is a complete graph.)

Definition 3.4. Suppose $\operatorname{Card}(E_i) \ge 2$ for each $i \in I$ and suppose L is a flag complex. Consider the polyhedral product $Z_L = (\operatorname{Cone} \mathbf{E}, \mathbf{E})^L$ from (3.4). Then \tilde{Z}_L , the universal cover of Z_L , is called (the standard realization of) a *right-angled building*, and abbreviated as RAB.

In fact, each RAB \tilde{Z}_L is the standard realization of a Tits building (see [?]); the associated Coxeter group is the RACG, W_L , associated to the 1-skeleton of L; each apartment is isomorphic to its Davis complex \tilde{P}_L .

3.4 Right-angled Artin groups

Salvetti complex

3.4 Right-angled Artin groups

3.4.1 The standard cube complex

3.4.2 Residually torsion-free nilpotent, left orderable

Chapter 4 Coxeter groups, Artin groups, buildings

Suppose L^1 is a simple graph with its edges labeled by integers ≥ 2 . Let $S = \text{Vert}L^1$. The edge labeling is a function $m : \text{Edge}(L^1) \to \{2, 3...\}$ written as $\{s, t\} \mapsto m(s, t)$. For s = t, put m(s, s) = 1. Then m(s, t) is a symmetric $(S \times S)$ -matrix called the *Coxeter matrix* associated to (L^1, m) . This data is enough to define two groups, a Coxeter group W and an Artin group A. Define a presentation of W so that its set of generators S coincides with the vertex set of L^1 . The relations are given by

$$s^2$$
, for all $s \in S$, and $(st)^{m(s,t)}$, for all $\{s,t\} \in \text{Edge}L^1$. (4.1)

That is to say, each generator of *S* is an involution and every other relation is a word involving only the two letters that correspond to the end points of an edge of L^1 . Then *W* is called the *Coxeter group* and the pair (W,S) is the *Coxeter system* associated to the labeled graph (L^1,m) . For any subset *T* of *S* the subgroup W_T generated by *T* is a *special subgroup*. It turns out that (W_T,T) is itself a Coxeter system. If W_T is finite, then the subset *T* and the subgroup W_T that it generates are said to be *spherical*. Denote by *S* (or *S*(*W*,*S*)) the set of spherical subsets of *S*. There is an associated abstract simplicial complex L(W,S) called the *nerve* of (W,S); its vertex set is *S* and its poset of simplices (including the empty simplex) is *S*.

To define the presentation for the associated Artin group A we introduce symbols $\{\sigma_s\}_{s\in S}$ for its generators. The relations are given by omitting the relations corresponding to the s^2 in (4.1) and by rewriting the relations corresponding to $(st)^{m(s,t)}$ as

$$\underbrace{\sigma_s \sigma_t \cdots}_{n(s,t) \text{ terms}} = \underbrace{\sigma_t \sigma_s \cdots}_{m(s,t) \text{ terms}}$$
(4.2)

In other words, the alternating word of length m(s,t) in σ_s and σ_t is equal to the alternating word of the same length written in the other order. The relation in (4.2) is called the *Artin relation* (or the *braid relation* when m(s,t) = 3). Since each *s* has order two, it follows that the Artin relation is simply a rewriting of $(st)^{m(s,t)}$. Hence, the map $\sigma_s \mapsto s$ gives a canonical epimorphism $A \to W$.

A spherical coset in W is a left coset in W/W_T , where W_T is a spherical special subgroup. Let

$$WS := \bigsqcup_{T \in S} W/W_T \tag{4.3}$$

denote the set of spherical cosets. The set WS can be given the structure of a poset. The order relation is inclusion of one coset in another. Similarly, a spherical coset in A is a left coset in A/A_T , where A_T denotes the spherical special subgroup of A generated by $\{\sigma_s\}_{s\in T}$. Let

4 Coxeter groups, Artin groups, buildings

$$AS := \bigsqcup_{T \in S} A/A_T \tag{4.4}$$

denote the poset of spherical cosets in A. The geometric realization of the order complex of WS is the barycentric subdivision of the Davis-Moussong complex $\Sigma(W,S)$ described in Subsection 4.2.3. The geometric realization of the order complex of AS is the barycentric subdivision of the Deligne complex $\Lambda(W,S)$ described in Subsection 4.3.3.

The poset of spherical cosets WS also plays an important role in the theory of buildings. As will be explained in Section 4.4, a building is a set C of "chambers" together with some extra structure. In particular, every building has an associated Coxeter system (W, S). The standard geometric realization of the building C is the order complex of its "poset of spherical residues." An "apartment" in C is a subposet isomorphic to WS.

Roughly speaking, the term "simple complex of groups" means a poset of groups, i.e., an assignment of a group G_{σ} to each element of a poset Q. The relevant definitions and the theory of simple complexes of groups are the subject of Chapter 5. Our goal in this chapter is to explain three basic examples of simple complexes of groups associated to a Coxeter system (W, S). For all three examples the underlying poset is the same: the poset S of spherical subsets of S. (Actually to be in compliance with the definitions in [23] we will eventually want to replace S by its opposite poset S^{op} where the order relations are reversed.) Here are the three types of examples.

- (1) The complex $\{W_T\}_{T \in S}$ of special spherical subgroups of W.
- (2) The complex $\{A_T\}_{T \in S}$ of special spherical subgroups of an Artin group A.
- (3) The complex $\{G_T\}_{T \in S}$ of residue stabilizers in building \mathcal{C} of type (W, S) with a chamber transitive automorphism group *G* (cf. Example **??**).

4.1 Some simple complexes of groups

The theory of simple complexes of groups is developed in [23]. Basic examples of simple complexes of groups are provided by Coxeter groups, Artin groups and buildings. So, we will make a few comments concerning the general theory here and postpone a fuller discussion to Appendix ??

Let \mathcal{P} be a poset. A *chain* in \mathcal{P} is a totally ordered subset. In other words, a finite subset $\{p_0, \dots, p_k\}$ is a chain if, possibly after reordering, $p_0 < \dots < p_k$. The *order complex* of \mathcal{P} is the set of all chains in \mathcal{P} . It is an abstract simplicial complex; a *k* simplex is a chain of length k + 1. Denote the geometric realization of the order complex by $|\mathcal{P}|$. Reversing the order relation on \mathcal{P} one obtains the *opposite poset* \mathcal{P}^{op} . Note that the order complexes of \mathcal{P} and \mathcal{P}^{op} are identical

Here is the definition from the book of Bridson-Haefliger.

Definition 4.1. (cf. Bridson-Haefliger [23, II.12.11, p. 375]). A simple complex of groups $G\Omega$ over a poset Ω is a collection of groups $\{G_{\sigma}\}_{\sigma\in\Omega}$ and monomorphisms $\phi_{\sigma\tau}: G_{\sigma} \to G_{\tau}$ defined whenever $\tau < \sigma$. The G_{σ} are the *local groups*. Furthermore, $G\Omega$ must be a cofunctor from Ω to the category of groups and monomorphisms in the sense that $\phi_{\sigma\tau}\phi_{\tau\mu} = \phi_{\sigma\mu}$ whenever $\mu < \tau < \sigma$. (A "cofunctor" is a contravariant functor, i.e., a functor on Ω^{op} .)

Simple complexes of groups arise from group actions with strict fundamental domains. Suppose $G \cap Y$ with strict fundamental domain $|\Omega|$. The space $|\Omega|$ has a stratification with strata indexed by Ω , where $|\Omega|_{\sigma} = |\Omega_{\leq \sigma}|$. (For example if Ω is the poset of cells in a cell complex, then each stratum is a cell.) The local groups are the stabilizers of the strata (one assumes the *G*-action is without inversions meaning that

Maybe not needed at this point

4.2 Coxeter groups

the setwise stabilizer of a stratum fixes each point of the stratum. So, G_{σ} is the isotropy subgroup of $|Q|_{\sigma}$. (The reason a simple complex of groups is defined as a cofunctor rather than a functor is that for group actions, a smaller stratum should correspond to a larger isotropy subgroup.) The simple complex of groups is *developable* if it is induced from a *G*-action.

The *direct limit*, $\lim GQ$, is defined in the usual fashion by taking the free product of the G_{σ} and then quotienting by relations to insure that G_{σ} is identified with its image under $\phi_{\sigma\tau}$ in G_{τ} and that $\phi_{\sigma\tau}\phi_{\tau\mu}(g) = \phi_{\sigma\mu}(g)$ whenever $\mu < \tau < \sigma$, for all $g \in G_{\mu}$. A *simple morphism* $\Psi = (\Psi_{\sigma})$ from GQ to a group G is a function which assigns to each $\sigma \in Q$ a homomorphism $\Psi_{\sigma} : G_{\sigma} \to G$ such that $\Psi_{\tau} = \Psi_{\sigma}\phi_{\sigma\tau}$ whenever $\tau < \sigma$. It turns out that GQ is developable if and only if the canonical simple morphism $G_{\sigma} \to \lim GQ$ is injective for each $\sigma \in Q$.

The next result uses the basic construction of Subsection 3.1.1 to show that every developable GQ arises from an action with strict fundamental on the basic construction.

Theorem 4.1. (The basic construction, cf. [23]). Suppose that GQ is developable and $G = \lim GQ$. Then there is a poset D(G,Q) such that $G \cap D(G,Q)$ with strict fundamental poset Q. Taking geometric realizations, we get a polyhedron D(G,|Q|) such that $G \cap D(G,|Q|)$ with strict fundamental domain |Q|.

The basic construction D(G, |Q|),

$$D(G, |\Omega|) = (G \times |\Omega|) / \sim \tag{4.5}$$

is defined as in Subsection 3.1.1. Given $x \in |\Omega|$, let $\sigma(x)$ be the index the smallest stratum $|\Omega|_{\sigma}$ such that $x \in |\Omega|_{\sigma}$. the equivalence relation \sim is then defined the same way as in (3.8):

$$(g,x) \sim (g',x') \iff x = x' \text{ and } g^{-1}g' \in G_{\sigma(x)}$$
, (4.6)

in other words, $gG_{\sigma(x)}$ and $g'G_{\sigma(x)}$ are the same coset in $G/G_{\sigma(x)}$.

It also turns out that when $|\Omega|$ is simply connected, then so is $D(G, |\Omega|)$ (see ??). So, when $|\Omega|$ is simply connected, $G = \lim G\Omega$ should be regarded as the "fundamental group of $G\Omega$ ".

Given a Coxeter system (W, S), we define two simple complexes of groups over S^{op} , where S = S(W, S), the poset of spherical subsets of *S*. Denote the order relation on S^{op} by \prec , so that $T' \prec T$ if *T* is a proper subset of *T'*. The *fundamental chamber K* is defined as $|S^{op}|$. The stratum K_T is the union of all simplices in $|S^{op}|$ with maximum vertex = *T*.

Example 4.1. (*The complex of spherical Coxeter subgroups of* W.) There is a simple complex of groups WS^{op} over S^{op} called the *complex of spherical subgroups of* W. It is defined as the cofunctor $T \mapsto W_T$, so that if $T' \prec T$, then W_T is a subgroup of $W_{T'}$. It is obvious that the direct limit, $\lim WS^{op}$, is equal toW. On the level of posets the basic construction $D(W, S^{op}$ is equal to the poset of spherical cosets WS defined in (4.3). On the level of spaces it is the cell complex $\Sigma(W, S)$ of Subsection 4.2.3 (actually, the strata of D(W, K) are dual cones to the cells in $\Sigma(W, S)$).

Example 4.2. (The complex of spherical Artin subgroups of A.) The *complex of spherical subgroups of A* is the simple complex of groups AS^{op} over S^{op} defined by $T \mapsto A_T$. If $T' \prec T$, then A_T is a subgroup of $A_{T'}$. We also have $\lim AS^{op} = A$. On the level of posets, $D(A, AS^{op}) = AS$ where AS is poset of spherical cosets from (4.4) and on the level of spaces, D(A, K) is the Deligne complex $\Lambda(W, S)$ of Subsection 4.3.3 below.

4.2 Coxeter groups

The material in this section can be found in [48].

4.2.1 Spherical Coxeter groups

Suppose (W, S) is a Coxeter system with $Card(W) < \infty$. The group W has a representation as an orthogonal linear reflection group on \mathbb{R}^n , with n = Card(S), so that each $s \in S$ corresponds to reflection across a supporting hyperplane of a simplicial cone. When (W, S) is irreducible this representation is unique up to homothety (e.g., see [15, p. 70]). It is called the *canonical representation*. The set of reflections R in W is precisely the set of conjugates of elements of S. A hyperplane H_r fixed by a reflection $r \in R$ is a *wall* and the set of all such hyperplanes, $\mathcal{A} := \{H_r\}_{r \in R}$ is the associated *reflection arrangement*. The arrangement \mathcal{A} cuts \mathbb{R}^n into simplicial cones; this collection of cones together and their faces is called the *fan*, Fan(\mathcal{A}), associated to \mathcal{A} . Each top-dimensional simplicial cone in Fan(\mathcal{A}) is called a *chamber*. There are two chambers (antipodal to each other) bounded by the walls indexed by S. Let C be one of them. Then C is a strict fundamental domain for the W-action on \mathbb{R}^n . A *codimension-one face* C_s of C is the intersection of C with the wall fixed by s. The intersection of Fan(\mathcal{A}) with \mathbb{S}^{n-1} is called the *spherical fan* and denoted \mathbb{S} Fan(\mathcal{A}). It is a tessellation of \mathbb{S}^{n-1} by spherical simplices. The underlying simplicial complex is the *Coxeter complex* of (W, S).

The intersection $C \cap \mathbb{S}^{n-1}$ is denoted by $\sigma(W, S)$ (or sometimes simply by σ) and is called the *fundamental* spherical simplex (or the *fundamental chamber*). The codimension-one faces of $\sigma (= \sigma(W, S))$ are the faces $\sigma_s := \sigma \cap H_s$ and for each proper subset T < S we have a face σ_T of σ defined by $\sigma_T := \bigcap_{s \in T}$. The isotropy subgroup at a point in the relative interior of σ_T is a special subgroup W_T . Since $W \cap \mathbb{S}^{n-1}$ with $\sigma(W, S)$ as a strict fundamental domain, the sphere \mathbb{S}^{n-1} can be identified with the basic construction $D(W, \sigma)$ defined Subsection 3.1.2, where, as in (3.8), the equivalence relation \sim on $W \times \sigma$ is defined by

$$(w,x) \sim (w',x') \iff x = x' \text{ and } wW_{S(x)} = w'W_{S(x)}.$$

$$(4.7)$$

Here $S(x) = \{s \in S \mid x \in \sigma_s\}.$

For each $s \in S$, let u_s be the outward-pointing unit normal vector to the face C_s of C (or equivalently, the outward-pointing normal vector to σ). Since the angle between u_s and u_t is the exterior dihedral angle along $C_s \cap C_t$, this angle is $\pi - \pi/m(s,t)$. Hence, the matrix of inner products $\langle u_s, u_t \rangle$ is equal to the *cosine matrix* c(s,t) of (W,S). This is the $(S \times S)$ -symmetric matrix defined by

$$c(s,t) = -\cos(\pi/m(s,t)).$$
 (4.8)

When s = t, m(s,s) = 1 so that c(s,s) = 1. (The formula in (5.6) for the cosine matrix c(s,t) makes sense for any Coxeter system (W,S), provided that when $m(s,t) = \infty$, we interpret $-\cos(\pi/\infty)$ to be $-\cos(0)$ (= -1).) Since $\{u_s\}_{s \in S}$ is a basis for \mathbb{R}^n when (W,S) is spherical we see that the cosine matrix is positive definite whenever W is finite. The converse is also true and this gives the following well-known characterization for a Coxeter group to be finite.

Lemma 4.1. (e.g. see [48, Theorem 6.12.9]). *The Coxeter group W is finite if and only if the cosine matrix from* (5.6) *is positive definite.*

The *fundamental dual simplex* $\sigma^*(W,S)$ is the spherical simplex in \mathbb{S}^{n-1} spanned by the unit vectors $\{u_s\}_{s\in S}$. The simplex $\sigma^*(W,S)$ actually is dual to $\sigma(W,S)$ in the sense that it consists of all points $y \in \mathbb{S}^{n-1}$ of distance $\geq \pi/2$ from $\sigma(W,S)$, i.e., $\sigma^*(W,S) = \{y \in \mathbb{S}^{n-1} \mid \langle y, x \rangle \leq 0$ for all $x \in \sigma(W,S)\}$. The length of the circular arc connecting the vertices u_s and u_t of $\sigma^*(W,S)$ is $\pi - \frac{\pi}{m(s,t)}$.

4.2 Coxeter groups

4.2.2 Coxeter zonotopes

Associated a spherical Coxeter system (W, S) there is a Coxeter zonotope Z (= Z(W, S)) that is dual to the reflection arrangement A. It is a simple convex polytope such that ∂Z is dual to the simplicial complex \mathbb{S} Fan(A). Here is an explicit description. Let C be the fundamental chamber bounded by the walls indexed by S. Choose a base point x_0 in the interior of C and consider its W-orbit, Wx_0 . Then Z(W, S) can be defined to be the convex hull of Wx_0 .

Lemma 4.2. (e.g. see [33, Lemma 2.1.3]). Suppose that (W, S) is a spherical Coxeter system and that Z(W, S) is the corresponding Coxeter zonotope. Then the face poset of Z(W,S) is isomorphic to the poset of spherical cosets WS defined in (4.3). The face corresponding to the coset wW_T has vertex set $(wW_T)x_0$ and this face is isomorphic to the zonotope $Z(W_T, T)$.

As examples, if (W, S) is the symmetric group on (n + 1) letters (i.e., if its Coxeter diagram is A_n), then Z is a permutohedron. When n = 2 a permutohedron is a hexagon. If $W = (\mathbb{C}_2)^n$, then Z is an *n*-cube. N.B. The action of W on Z, regarded as a cell complex Z "has inversions," e.g., the maximum cell Z is stabilized by the entire group W but is not pointwise fixed.

The link, $Lk(x_0, Z)$ of the vertex x_0 in Z is naturally identified with the dual spherical simplex $\sigma^*(W, S)$. The edges meeting at the vertex x_0 are parallel to the basis of unit normal vectors, $\{u_s\}_{s \in S}$. The metric on Z(W, S) depends on the position of x_0 in the interior of C. We can normalize this position by choosing x_0 to be the unique point of distance = 1/2 from each wall H_s , $s \in S$. This will have the effect of giving each edge of Z(W, S) a length of 1. The face corresponding to wW_T is said to be of *type T*; it is a cell of dimension = Card(T).

Figure of hexagon and pemutohedron

Zonotopes play the same role in constructing the standard CAT(0) complexes for general Coxeter groups as do cubes for RACGs. They are also used in the construction of the Salvetti complex for a general Artin groups.

4.2.3 The Davis-Moussong complex

We return to the situation where (W, S) is an arbitrary Coxeter system. Let S = S(W, S) be its poset of spherical subsets and let WS be the set of spherical cosets (cf. (4.3)). There is a poset structure on WS defined by inclusion of cosets, so that

$$uW_T < vW_{T'} \iff T < T' \text{ and } v^{-1}u \in W_{T'}.$$

$$(4.9)$$

The purpose of this subsection is to describe a cell complex $\Sigma = \Sigma(W, S)$ together with a proper action $W \cap \Sigma$. The poset of cells of $\Sigma(W, S)$ is WS; the cell corresponding to the spherical coset wW_T is isomorphic to the zonotope $Z(W_T, T)$ (cf. Lemma 4.2). It follows that the barycentric subdivision of $\Sigma(W, S)$ is the geometric realization |WS| of the order complex of WS. (The *order complex* of a poset \mathcal{P} is the simplicial complex whose *k*-simplices are the chains, $p_0 < \cdots < p_k$ of length k + 1.)

Since the complex $\Sigma(W, S)$ plays a central role in [48] and is described in detail there, we will content ourselves with giving a briefer description here. In the case where W is right-angled, $\Sigma(W, S)$ is the cubical complex \tilde{P}_L explained at the end of Subsection 3.1.1. Some properties of Σ are listed in the next proposition.

Proposition 4.1. Let $\Sigma = \Sigma(W, S)$.

Coxeter zonotope

- (1) The poset of cells of in the cell complex Σ is equal to WS.
- (2) Each cell of $\Sigma(W,S)$ is a Coxeter zonotope. The cell corresponding to the spherical coset wW_T is isomorphic to the Coxeter zonotope $Z(W_T,T)$; it is a convex polytope of dimesion Card T.
- (3) The group W acts properly on Σ . The stabilizer of the cell $wZ(W_T,T)$ is the subgroup wW_Tw^{-1} .
- (4) The W-action on Σ has a strict fundamental domain homeomorphic to the geometric realization |S| of the poset S. Hence, when S is finite, Σ/W is compact.
- (5) The natural piecewise euclidean metric on Σ is CAT(0). In particular, by Theorem 2.1, Σ is contractible.

In order to appreciate the naturalness and beauty of Σ , we spell out its construction in some detail. Its 0-skeleton, Σ^0 , is identified with W (i.e., with the spherical coset W/W_{\emptyset}). Its 1-skeleton is the Cayley graph Cay(W, S) – there is an edge connecting w to ws whenever $\{w, ws\}$ is a coset of $W_{\{s\}}$. So, there is an orbit of edges for each generator $s \in S$. Next we fill in the 2-cells. There is an orbit of 2-cells of the form $Z(W_{\{s,t\}}, \{s,t\})$, whenever Card $\{s,t\} = 2$ and $m(s,t) < \infty$. Each such 2-cell a 2-dimensional Coxeter zonotope, in this case, it is a polygon with 2m(s,t) sides. The orbits of these 2-cells correspond precisely to the relations in (4.1). Essentially, Σ^2 is the Cayley 2-complex of (W, S). In particular, Σ^2 is simply connected. We continue by filling in an orbit of 3-dimensional zonotopes of the form $Z(W_T, T)$ for each spherical subset $T \in S$ with 3 elements. Miraculously, after filling in all such 3-cells, the resulting 3-skeleton, Σ^3 is 2-connected. The complex Σ is formed by filling in an orbit of Coxeter zonotopes for each spherical coset in WS. It turns out that Σ is contractible. One way to to show this is to prove statement 5 in the above proposition: Σ is CAT(0). This is a theorem of Moussong which will be discussed in the next subsection. We note that when W is finite, $\Sigma(W, S)$ is equal to the single zonotope Z(W, S) (cf. Lemma'4.2).

The *nerve* of (W, S) is the simplicial complex L(W, S) with vertex set S such that a nonempty subset $T \leq S$ spans a simplex if and only if $T \in S(W, S)$. There is natural piecewise spherical metric on L(W, S) where the simplex corresponding to T is identified with the fundamental dual spherical simplex $\sigma^*(W_T, T)$ defined above. In particular, the edge $\{s,t\}$ that corresponding to the spherical subset $T = \{s,t\}$ has length $\pi - \pi/m(s,t)$.

A spherical simplex σ has size $\geq \pi/2$ if the distance from any vertex to the opposite face is $\geq \pi/2$.

Definition 4.2. (*Metric flag complexes*). Suppose *L* is a simplicial complex with a piecewise spherical metric so that whenever *e* is an edge of *L*, its length, l(e), lies in the interval $[\pi/2, \pi]$ (this follows from the assumption that each simplex of *L* has size $\geq \pi/2$). Then *L* is a *metric flag complex* if whenever Γ is a subcomplex of *L* isometric to the 1-skeleton of a spherical *k*-simplex σ , then there must actually exist a *k*-simplex σ in *L* with edge lengths specified by the l(e).

The condition in this definition also can be expressed as follows. Suppose *T* is the set of vertices of a subcomplex isomorphic to a complete graph Γ and that for distinct vertices *t*, *t'* of Γ , the length l(t,t') of the edge $\{t,t'\}$ lies $[\pi/2,\pi)$. Let c(t,t') be the "cosine matrix" defined by

$$c(t,t') = \begin{cases} 1, & \text{if } t = t';\\ \cos(t,t'), & \text{if } t \neq t'. \end{cases}$$
(4.10)

The condition that eac Then *L* is a metric flag complex if and only if the $(T \times T)$ -symmetric matrix c(t,t') is positive definite whenever *T* is vertex set of a simplex of *L*. In other words, the simplices of *L* are determined by the metric on L^1 . By Lemma 4.1, for each $T \in S$, the corresponding cosine matrix is positive definite; so, L(W,S), with its natural piecewise spherical structure L(W,S) is a metric flag complex.

Example 4.3. (*Metric flag complexes of dimension* 1). When is the graph L^1 a CAT(1) cell complex? The answer was alluded to in Examples 2.2: the length of each circuit in L^1 must be $\geq 2\pi$. If all edge lengths lie
4.2 Coxeter groups

in $[\pi/2, \pi]$ and if L^1 is a simplicial graph, then this condition is vacuous except for circuits with 3 edges, say e_1, e_2, e_3 , in which case the condition reads:

$$l(e_1) + l(e_2) + l(e_3) \ge 2\pi. \tag{4.11}$$

This is precisely the condition that the cosine matrix for the 3-circuit fails to be positive definite. Thus, L^1 is a metric flag complex if and only if (4.11) holds for each 3-circuit in L^1 and this is equivalent to the condition that L^1 be CAT(1). Moussong's Lemma, which is stated in the next subsection, is a generalization of this to higher dimensions.

Example 4.4. (Σ *is a 2-dimensional* CAT(0) *cell complex*, cf. Example 2.3). Suppose (L^1, m) is a labeled simplicial graph with edge labeling m: Edge $L^1 \rightarrow \{2, 3, ...\}$ with associated Coxeter system (W, S). As in the previous example suppose each spherical subset $T \in S$ has cardinality ≤ 2 so that $L(W, S) = L^1$. Then dim $\Sigma = 2$. The 2-dimensional zonotope corresponding to an edge e is a regular euclidean 2m(e)-gon. The interior angle at a vertex of such a 2m(e)-gon is $\pi - \pi/m(e)$. If $\{e_1, e_2, e_3\}$ is a 3-circuit of L^1 , then the condition in (4.11) is equivalent to

$$\sum_{i=1}^{3} \frac{1}{m(e_i)} \le 1.$$
(4.12)

So, dim $\Sigma = 2$ exactly when (4.12) holds for all 3-circuits in L^1 . Thus, Σ is a 2-dimensional CAT(0) cell with an orbit of 2m(e)-gons for each $e \in L^1$.

Next, consider the question of when does Σ admit a piecewise hyperbolic structure that is CAT(-1). First, there is the question of how to make the zonotopes hyperbolic. If W_T is a spherical Coxeter group, then we can represent it as a finite reflection group on \mathbb{H}^m , where m = Card T. Let C^h be a fundamental cone for the W_T -action on \mathbb{H}^m and choose a point x_0 in the interior of C of distance $\frac{1}{2\varepsilon}$ from each wall. Let $Z^h_{\varepsilon}(W_T, T)$ ($= Z^h_{\varepsilon}$) denote the convex hull of the orbit of x_0 . Then $Z^h_{\varepsilon}(W_T, T)$ is the hyperbolic zonotope of with all edge lengths $= \varepsilon$. As before, we can glue together the $Z^h_{\varepsilon}(W_T, T)$ to obtain a piecewise hyperbolic model $\Sigma^h(W, S)$ of the Davis-Moussong complex. When ε is small, the effect on Lk($x_0, Z^h_{\varepsilon}(W_T, T)$ will be to make a small change in the dual spherical simplex $\sigma^*(W_T, T)$ Hence, there also will only be a small change in the metric on Lk($v, \Sigma^h_{\varepsilon}$) (cf. [102, Lemma 5.11] or [48, Lemma I.6.7]).

Example 4.5. (2-dimensional CAT(-1) cell complexes). Suppose dim $\Sigma = 2$. Then a small deformation of the metric on L^1 is guaranteed to be CAT(1) if and only if the sum of the edge lengths in each circuit is strictly greater than 2π . As in Example 4.4 we only consider circuits with ≤ 4 edges. If the number of edges is 3 we must modify (4.12) by requiring the inequality to be strict. If the number of edges is 4, the only case which leads to a closed geodesic of length 2π is when each edge has length exactly $\pi/2$. We exclude that possibility by requiring that for 4-circuits at least one edge has $m(e) \neq 2$. (This is a more general version of the no \Box condition of Section 2.3.) Following the discussion in Subsection 2.2.1 we see that a small deformation of the piecewise euclidean structure on Σ to a piecewise hyperbolic Σ_{ϵ}^h is CAT(-1) if and only if the above two conditions hold for for L^1 , i.e., if L^1 is extra large in the sense of Definition 2.2, in which case each new vertex links will be CAT(1). Hence, for small enough ε , Σ_{ϵ}^h will be CAT(-1) if and only if L^1 is extra large.

Example 4.6. (*Gromov polyhedra*). Here is a corollary to the previous example. Suppose *m* is an integer ≥ 2 and $n \geq 3$. Then there is a simply connected, 2-dimensional polyhedron $X_{n,2m}$ so that each 2-cell is a 2m-gon and so that the link of each vertex is a complete graph on *n*-vertices. (Apply the construction in the previous example to the case where L^1 the complete graph and each edge is labelled by the same integer *m*, and $X_{n,m} = \Sigma(W,S)$, or its 2-skeleton when m = 2.) If $m \geq 3$, then $X_{n,m}$ is CAT(0). If $m \geq 4$, it can be given a CAT(-1) metric where each polygon is hyperbolic.

4.2.4 Moussong's Lemma

Lemma 4.3. (Moussong's Lemma, cf. [102] or [48]). A piecewise spherical simplicial complex L with all simplices of size $\geq \pi/2$ is CAT(1) if and only if it is a metric flag complex.

Moussong's Theorem is a corollary of this lemma.

Theorem 4.2. (Moussong [102], also cf. [62] or [48, Lemma I.7.4]). *The piecewise euclidean cell complex* $\Sigma(W, S)$ *is* CAT(0).

Proof (Comments on). As explained in Subsection 4.2.2, the link of a vertex in the zonotope $Z(W_T, T)$ is isometric to the dual spherical simplex $\sigma^*(W_T, T)$. It follows that the link of each vertex in $\Sigma(W, S)$ is isometric to L(W, S). By Moussong's Lemma, this link is CAT(1); hence, $\Sigma(W, S)$ is NPC. Since the 2-skeleton of Σ is the Cayley 2-complex for (W, S), we see that Σ is simply connected; hence, by Theorem 2.2, Σ is CAT(0).

When (W, S) is right-angled, its nerve L(W, S) is a flag complex and it has an all-right, piecewise spherical structure. By Lemma 2.10, L(W, S) is extra large if and only if it satisfies the no \Box condition. So, the cubical metric on $\Sigma(W, S)$ can be deformed to a piecewise hyperbolic metric that is CAT(-1) if and only if L(W, S) satisfies the no \Box condition. Basically, this establishes the following proposition. We will give details of its proof later in Theorem 4.3 below.

Proposition 4.2. Suppose (W,S) is a right-angled Coxeter system (a "RACS"). Then (W,S) is word hyperbolic if and only if L(W,S) satisfies the no \Box condition.

Moussong continued along the line begun in Lemma 4.3 by determining when a piecewise spherical, metric flag complex *L* of size $\geq \pi/2$ is extra large (cf. Definition 2.2). As usual S = Vert L. For T < S let L_T denote the full subcomplex spanned by *T*. Then L_T is a totally geodesic subspace of *L*. There are two obvious types of subcomplexes L_T that can lead to a closed geodesic of length 2π in *L*. They are the following.

- (i) The complex L_T is isometric to the polar dual of a euclidean simplex so that L_T is isometric to the round sphere of dimension Card(T) - 2 and LT combinatorially isomorphic to the boundary complex of a simplex of dimension Card(T) - 1.
- (ii) The complex L_T decomposes as a spherical join $L_{T_1} * L_{T_2}$, where neither L_{T_1} nor L_{T_2} is a spherical simplex.

To see that (ii) yields a closed geodesic of length 2π , for i = 1, 2, choose points x_i, y_i in L_{T_1} of distance at least π . Then the join $\{x_1, y_1\} * \{x_2, y_2\}$ is a 4-circuit where each edge has length $\pi/2$. Moussong proved in [102, Lemma 10.3] that the metric flag complex L has no closed geodesics of length 2π if and only if it contains no subcomplexes L_T of type (i) or (ii). Applying this in the special case L = L(W, S) we get the following.

Lemma 4.4. The piecewise spherical complex L(W,S) is extra large if for each subset T of S neither of the following conditions holds for W_T .

- (i) The special subgroup W_T is an irreducible euclidean reflection group (generated by the reflections across the faces of a euclidean simplex of dimension ≥ 2 .
- (ii) (W_T, T) decomposes as $(W_{T_1} \times W_{T_2}, T_1 \sqcup T_2)$, where both W_{T_1} and W_{T_2} are infinite.

If both conditions (i) and (ii) of this lemma fail to hold, we say that (W, S) satisfies *Moussong's Condition*.

4.2 Coxeter groups

Theorem 4.3. (Moussong's characterization of word hyperbolic Coxeter groups). *The following conditions* on a (finitely generated) Coxeter system (W, S) are equivalent.

(a) The group W is word hyperbolic.

(b) The group W does not contain $\mathbb{Z} \times \mathbb{Z}$ (the product of two infinite cyclic groups.

- (c) The Coxeter system (W,S) satisfies Moussong's Condition.
- (d) The CAT(0) metric on $\Sigma(W,S)$ can be deformed to a piecewise hyperbolic, CAT(-1) metric.

Proof. Since a word hyperbolic group cannot contain $\mathbb{Z} \times \mathbb{Z}$, (a) \implies (b). If a special subgroup W_T is a euclidean reflection group, then its translation subgroup is free abelian of rank $\operatorname{Card}(T) - 1$ and this rank is ≥ 2 when (i) holds. Similarly, if $W_T = W_{T_1} \times W_{T_2}$, where both factors are infinite, then $\mathbb{Z} \times \mathbb{Z} < W_T$. So, (b) \implies (c). By Lemma 4.4, Moussong's Condition implies that L(W,S) is extra large. As in Example 4.5, Proposition 2.1 implies that $\Sigma(W,S)$ can be given a piecewise hyperbolic $\operatorname{CAT}(-1)$ metric. So, (c) \implies (d). By Lemma 2.7, (d) \implies (a).

Examples of word hyperbolic Coxeter groups. Moussong, Benoist.

4.2.5 Piecewise euclidean metrics on K

Let $K = |S^{op}| (= |S|)$ be the fundamental chamber defined in Section 4.1 and put $K_T = S_{\preceq T}^{op}$. In other words, K_T is the union of simplices in the order complex of S^{op} with maxmum vertex T. Since $W \curvearrowright \Sigma$ with strict fundamental domain K (cf. Definition 3.2), we get a different description of Σ in terms of the basic construction of (5.2): $\Sigma = D(W, K)$. Here $D(W, K) = (W \times K) / \sim$, where \sim is defined by: $(u, x) \sim (v, x')$ if and only if x = x' and $uW_{S(x)} = vW_{S(x)}$, where S(x) is the index of the smallest stratum containing x. This description is dual to the cellulation of Σ by Coxeter zonotopes described in Proposition 4.1.

The fundamental domain *K* is a cube complex. For any $T \in S$, the poset $S_{\leq T} (= S_{\geq T}^{op})$ is the power set of *T*. This means that the order complex of $S_{\geq T}^{op}$ can be identified with a standard subdivision of a cube \Box^T of dimension $\operatorname{Card}(T)$. More generally, for any subset *T'* of *T* consider the interval [T, T'] in S^{op} defined by $[T, T'] = \{J \in S \mid T' \leq J \leq T\}$. So, [T, T'] is a standard subdivision of a cube $\Box^{T-T'} (\Box^{[T,T']})$ of dimension $\operatorname{Card}(T - T')$. Explicitly, $\Box^{[T,T']}$ is the subcomplex of $|S^{op}$ consisting of all simplices in $|S^{op}|$ with minimum vertex *T* and maximum vertex *T'*. In other words, the simplices in $|S^{op}| (=K)$ can be amalgamated into cubes, giving *K* the structure of a cube complex. By taking the union of all translates of such cubes in $|WS| (= \Sigma)$ we see that Σ is also a cube complex. We state this as follows.

Lemma 4.5. The fundamental chamber K has the structure of a cube complex as does the Davis-Moussong complex Σ .

The subcomplexes $\Box^{[T,T']}$ are only combinatorially equivalent to cubes. What is the relationship between these combinatorial cubes and the Coxeter zonotopes in Σ . Suppose $Z(W_T, T)$ is the Coxeter zonotope in \mathbb{R}^T corresponding to W_T and let Fan (W_T) be the fan cut out by the reflecting hyperplanes. A top dimensional simplicial cone in Fan (W_T) is called a *sector*. The intersection $B_T := Z(W_T, T) \cap \text{Fan}(W_T)$ is called a *Coxeter block*; it is combinatorially isomorphic to the cube $\Box^{[T,\emptyset]}$. However, B_T need not be isometric to to a unit cube. (The link of the vertex the Coxeter block corresponding to $T \in S_{\leq T}$ (the cone point in the fan) is isometric to the spherical simplex $\sigma(W_T, T)$ while the link at the opposite vertex corresponding to $\emptyset \in S_{\leq T}$ is the dual spherical simplex $\sigma^*(W_T, T)$ defined near the end of Subsection 4.2.1.) The *natural piecewise euclidean metric on* Σ is the path metric defined by giving each zonotopal cell in Σ its natural metric as a Coxeter zonotope (say, with each edge of length = 2). Similarly, the *natural piecewise euclidean metric on K* is the path metric defined by giving each Coxeter block its natural metric as a convex subset of the Coxeter zonotope. We just showed in Theorem 4.2, that with its natural piecewise euclidean metric Σ is CAT(0). On the other hand, we could give *K* the piecewise euclidean metric where each subcomplex $\Box^{[T,T']}$ is isometric to the unit cube. Denote *K* with this cubical metric by K^{\Box} . Similarly, the cubical metric on Σ is denoted Σ^{\Box} . The cube complexes K^{\Box} and Σ^{\Box} may fail to be CAT(0) since links need not be flag complexes. the condition that we need to insure that all klinks are flag complexes is given in the following definition.

Figure of pentagon with square, Figure of hexagon with square

Definition 4.3. A Coxeter system (W,S) is *type FC* if its nerve L(W,S) is a flag complex. Similarly, an Artin group is type FC if its associated Coxeter system is FC.

The cubical structures on these polyhedra induce the piecewise euclidean cubical metrics, denoted by K^{\Box} and Σ^{\Box} , respectively. N.B. In general, K^{\Box} and Σ^{\Box} need not be CAT(0). On the other hand, we just showed in Theorem 4.2, that Σ is CAT(0) when given the metric induced from its zonotopal cells. We shall see in Proposition **??** below, the condition that we need to insure that the cube complex Σ^{\Box} is CAT(0) is that the nerve L(W, S) of the Coxeter system is a flag complex.

Proposition 4.3. The cube complex Σ^{\Box} is CAT(0) if and only if the Coxeter system (W,S) is type FC.

Proof. Of course, this should follow from the version of Gromov's Lemma stated as Corollary 2.3; however, it does not quite immediately follow from the fact the fact that L(W,S) (=L) is a flag complex. Although it is not true that the link of every cube in Σ is isomorphic to L or a sublink of the form $Lk(\sigma,L)$, it is true that the link of every 0-cube in Σ is isomorphic to a join $SFan(W_T,T) * Lk(\sigma_T,L)$, where $SFan(W_T,T)$ is the spherical Coxeter complex and σ_T is the simplex corresponding to T. Since any spherical Coxeter complex is a triangulation of a sphere as a flag complex (cf. [33]), both factors of the join are flag complexes and hence, so is the join.

This shows that when (W, S) is type FC one can prove, without using Moussong's Lemma, that $\Sigma(W, S)$ has a CAT(0) structure. In Subsection 4.3.2 essentially the same observation will be used to show that if an Artin group is FC, then its Salvetti complex has a CAT(0) cubical structure.

4.2.6 The Tits representation

When *W* is infinite, it also has a representation on \mathbb{R}^n , with $n = \operatorname{Card} S$, as a group generated by linear reflections (which need not be orthogonal linear transformations). Let c(s,t) be the cosine matrix defined as in (5.6) with the proviso that c(s,t) = -1 when $m(s,t) = \infty$. Then there is a symmetric bilinear form $B : \mathbb{R}^S \times \mathbb{R}^S \to \mathbb{R}$ defined by $B(e_s, e_t) = c(s, t)$. For each $s \in S$, let ρ_s be the linear reflection on \mathbb{R}^S defined by

$$\rho_s(x) := x - 2B(e_s, x)e_s. \tag{4.13}$$

(The eigenvalue -1 of ρ_s has eigenvector e_s .) The map $s \mapsto \rho_s$ extends to a homomorphism $\rho : W \to GL(\mathbb{R}^S)$, called the *canonical representation*. More interesting to us is the dual of the canonical representation, $\rho^* : W \to GL((\mathbb{R}^S)^*)$, which we call the *geometric representation* or the *Tits representation* of W. The main fact about the geometric representation is that it defines a *W*-action on a certain open convex set Ω with a strict fundamental domain. This allows us to conclude that the representation ρ^* is discrete and faithful.

36

4.3 Artin groups

To see this, first define a simplicial cone C in $(\mathbb{R}^S)^*$ by the inequalities: $e_s(v) \leq 0$, for all $s \in S$. (Each basis vector e_s gives linear form, $v \mapsto e_s(v)$ on $(\mathbb{R}^S)^*$.) For each subset $T \leq S$, let C_T be the face of C defined by $e_s(v) = 0$, for all $s \in T$ and let C^f denote the complement in C of the union of all faces C_T with $T \notin S$. That is to say, $C^f = \{x \in C \mid S(x) \in S\}$, where $S(x) = \{s \in S \mid x \in C_s\}$. The *Tits cone* $\overline{\Omega}$ is defined to be $\bigcup_{w \in W} wC$. Let Ω denote the interior of $\overline{\Omega}$. Tits established the following facts (e.g., see [15, Prop. 6, p. 102]).

- (a) For any nontrivial $w \in W$, $w(int C) \cap (int C) = \emptyset$.
- (b) $\overline{\Omega}$ is a convex cone. So, its interior Ω is an open *W*-stable subset in \mathbb{R}^n (= (\mathbb{R}^S)*).
- (c) $\Omega = \bigcup_{w \in W} wC^f$.
- (d) The isotropy subgroup at a point $x \in C^f$ is W_T where C_T is the smallest face such that $x \in C_T$.

It follows from (a) that $\rho^* : W \to GL(\mathbb{R}^n)$ is injective and that its image is a discrete subgroup of $GL(\mathbb{R}^n)$. By (d) each isotropy subgroup is finite; hence, W acts properly on Ω .

Theorem 4.4. The group W acts on Ω with strict fundamental domain C^f . Thus, Ω is equivariantly homeomorphic to the basic construction $D(W, C^f)$ defined by formula (3.8) in Section 3.1. It follows that there is a W-equivariant, piecewise linear embedding $\Sigma(W, S) \hookrightarrow \Omega$ whose image is a spine for Ω .

So, $W \curvearrowright \Omega$ as a reflection group.

4.3 Artin groups

The material in this section comes from [64], [32, 33].

4.3.1 The complement of the complexification of a reflection arrangement

Let (W, S) be a spherical Coxeter system so that W acts on \mathbb{R}^n (= \mathbb{R}^S) via the geometric representation. Let \mathcal{A} be the corresponding reflection arrangement of hyperplanes in \mathbb{R}^n . Complexifying, we get $W \curvearrowright \mathbb{C}^n$ as well as a reflection arrangement of complexified hyperplanes $\mathcal{A} \otimes \mathbb{C}$ in \mathbb{C}^n . The complement of this arrangement is denoted by

$$M(\mathcal{A} \otimes \mathbb{C}) := \mathbb{C}^n - \bigcup_{H \in \mathcal{A} \otimes \mathbb{C}} H.$$
(4.14)

It is well-known that $\pi_1(M(\mathcal{A} \otimes \mathbb{C})) = PA_W$, where PA_W is the *pure Artin group* is defined as the kernel of the canonical epimorphism $\varphi : A_W \to W$. The group W acts freely on $M = M(\mathcal{A} \otimes \mathbb{C})$. So, $\pi_1(M/W)$ is the Artin group A_W . For example, if W is the symmetric group S_{n+1} acting on \mathbb{R}^{n+1} by permuting the coordinates, then the complement M is identified with the configuration space of n+1 distinct ordered points in \mathbb{C} and M/S_{n+1} is the unordered configuration space. It follows from the definition of the braid group, that $\pi_1(M)$ is the *pure braid group* on n+1 strands, PB_{n+1} , while $\pi_1(M/S_{n+1})$ is the *braid group*, B_{n+1} .

Theorem 4.5. (Deligne's Theorem [64]). *The arrangement complement M is aspherical. Hence, M is a model for BPA*_W and M/W is a model for BA_W

In fact, Deligne proved that whenever \mathcal{A} is a real simplicial arrangement in \mathbb{R}^n , the complexified complement $M(\mathcal{A} \otimes \mathbb{C})$ is aspherical.

When *W* is not required to be finite, we can realize it as a group generated by reflections on $\Omega < \mathbb{R}^n$ via the Tits representation $\rho^* : W \to GL(\mathbb{R}^n)$, as in Subsection 4.2.6. The group *W* acts properly on the open subset $\mathbb{R}^n + i\Omega$ of \mathbb{C}^n and freely on the arrangement complement, $M = M(A \otimes \mathbb{C})$, defined by

$$M = (\mathbb{R}^n + i\Omega) - \bigcup_{H \in \mathcal{A} \otimes \mathbb{C}} H , \qquad (4.15)$$

where \mathcal{A} is the collection of all reflecting hyperplanes in \mathbb{R}^n .

The following conjecture, which was made independently by Arnold, Phan, and Thom, asserts that a generalization of Theorem 5.3 holds for any Artin group. It is the most important open question about Artin groups. the conjecture is explained in detail in [32, 33] and [96].

Conjecture 4.1. (The $K(\pi, 1)$ -Conjecture for Artin groups). The space M/W is a model for BA_W .

There is an order relation on $W \times S$ that is closely related to the one given in (4.9). it is defined by (u,T) < (v,T') if and only if the following two conditions hold:

(a) $uW_T < vW_{T'}$ as in (4.9), i.e., $v^{-1}uW_T < W_{T'}$, (b) $v^{-1}u$ is the shortest element in the coset $v^{-1}uW_T$.

After comparing (a) and (4.9), we see that the projection $p: W \times S \to WS$ defined by $(w, T) \mapsto wW_T$ is order-preserving.

Theorem 4.6. (Salvetti [122] and Charney-Davis [33]). *The arrangement complement M is homotopy equivalent to* $|W \times S|$.

The proof is based on the following standard result which combinatorialists call the "Nerve Lemma".

Lemma 4.6. (The Nerve Lemma). Suppose \mathcal{U} is a locally finite, open cover of a paracompact space Y such that each $U \in \mathcal{U}$, as well as, each finite, nonempty intersection of such U is contractible. Then Y is homotopy equivalent to the nerve of the open cover, Nerve(\mathcal{U}).

Theorem 4.6 is then proved by applying the Nerve Lemma to a certain open cover \mathcal{U} of M, the nerve of which is the order complex of $W \times S$. Although M is defined as a subset of $\mathbb{R}^n + i\Omega$, we can replace it by the homotopy equivalent space $M \cap (b\Sigma \times b\Sigma)$, where the barycentric subdivision $b\Sigma$ of Σ is identified with the order complex of WS. For each $(w, T) \in W \times S$, define $\operatorname{Star}(w, T)$ to be the open star of vertex corresponding to wW_T in $b\Sigma$, let $\operatorname{Sec}(w, T)$ be the open "sector" in Σ which is bounded by the walls indexed by wTw^{-1} and which contains the open chamber $\operatorname{Star}(w, \emptyset)$. Put

$$U(w,T) := \operatorname{Star}(w,T) \times \operatorname{Sec}(w,T).$$

It is clear that both $\operatorname{Star}(w, T)$ and $\operatorname{Sec}(w, T)$ are contractible; hence, so is U(w, T) If $x \in \operatorname{Star}(w, T)$ lies in a wall, then that wall is indexed by a reflection in wW_Tw^{-1} . Since the open sector $\operatorname{Sec}(w, T)$ intersects no such wall, we see that U(w, T) < M. We see that $\mathcal{U} = \{U(w, T))\}_{(w, T) \in W \times S}$ is an open cover of M. It is proved in [33, Lemma 1.5.2 (ii)] that $U(w_0, T_0) \cap \cdots \cap U(w_k, T_k)$ is nonempty precisely when $\{(w_0, T_0), \dots, (w_k, T_k)\}$ is a chain in $W \times S$. Thus, Nerve($\mathcal{U} = |W \times S|$. Theorem 4.6 follows.

Combining Salvetti's Theorem 4.6 with Deligne's Theorem 5.3, we see that when W is spherical, the finite CW complex $X = |W \times S|$ is a model for BPA_W (and so, X/W is a model for BA_W). A corollary is that each spherical Artin group is type F.

4.3 Artin groups

4.3.2 The Salvetti complex and its universal cover

Given a locally finite arrangement \mathcal{A} of affine hyperplanes in \mathbb{R}^n , Salvetti [122] constructed a cell complex $X(\mathcal{A})$ homotopy equivalent to the complement in \mathbb{C}^n of the complexification of the arrangement. If there are finitely many hyperplanes in the arrangement, then $X(\mathcal{A})$ is a finite cell complex. The cells of $X(\mathcal{A})$ are zonotopes – each zonotope is the dual of a central arrangement normal to an intersection of hyperplanes from \mathcal{A} . When \mathcal{A} is the arrangement associated to the action of W on Ω , the cell complex $X (= X(\mathcal{A}))$ is homotopy equivalent to the complement M defined in (5.11).

The barycentric subdivision *bX* of the Salvetti complex *X* can be defined as the geometric realization of the order complex of $W \times S$. A *k*-simplex in the order complex is a chain, $(w_0, T_0) < \cdots < (w_k, T_k)$. We note that $(W \times S)_{\leq (w,T)} \cong WS_{\leq wW_T} \cong W_T(S_{\leq T})$. So, $(W \times S)_{\leq (w,T)}$ is isomorphic to the poset of faces of the Coxeter zonotope $Z(W_T, T)$. There is a cell structure on *X* defined by identifying the union of all in simplices of *bX* with maximum vertex (w, T) with the cell $Z(W_T, T)$. Let us also use the notation X(W, S) for X (= X(A)) (= X) and call it the *Salvetti complex* of (W, S).

Using Theorem 4.6, we get the following analog of Proposition 4.1.

Proposition 4.4. (cf. [122], [33]). Let X(W,S) be the Salvetti complex as defined above.

- (1) The poset of cells in X(W,S) is equal to $W \times S$. The cell corresponding to (w,T) is the Coxeter zonotope $wZ(W_T,T)$; its dimension is equal to Card T.
- (2) The fundamental group of X(W,S) is the pure Artin group PA_W .
- (3) There is a free action $W \curvearrowright X(W,S)$. The quotient space X(W,S)/W is a finite CW complex (provided the set of generators S is finite).

Here is another way to understand the cell structure on X (= X(W, S)). Each cell of X can be represented as a pair (v, F), where F is a zonotopal cell in $\Sigma (= \Sigma(W, S))$ and $v \in \text{Vert } F$. The partial order on the set of such (v, F) is defined by (v, E) < (v', F') if and only if F < F' and the shortest edge path from v' to Fterminates at v'. The geometric realization of the projection $p : W \times S \to WS$ is also denoted $p : X \to \Sigma$. If Fis a cell of Σ of type (W_T, T) , then the set of connected components of $p^{-1}(\text{int } F)$ is naturally bijective with Vert F. The closure of such a component is isomorphic to F. Just as we did for $\Sigma(W, S)$ in Subsection ??, let us spell out the cell structure on X(W, S) in more detail.

- (a) A vertex of X has the form (v, v) where $v \in \text{Vert } \Sigma$. Hence, the 0-skeleton of X is naturally isomorphic to the 0-skeleton of Σ and both correspond bijectively to W.
- (b) If *E* is an edge of Σ with end points *u* and *v*, then there are two edges in *X* lying above it: (u, E) and (v, E). The edge (u, E) is directed so that its initial vertex is (u, u) and its terminal vertex is (v, v). If *E* is labeled by the element *s*, then label the two directed edges lying above it by the Artin generator σ_s . Thus, each edge of Σ is doubled to get a circuit consisting of two directed edges in *X*.
- (c) Similarly, if *F* is a 2*m*-gon in Σ corresponding to an edge {*s*,*t*} ∈ Edge L¹ with *m*(*s*,*t*) = *m*, then *p*⁻¹(*F*) consists of 2*m* copies of *F*. The face (*v*,*F*) is glued to the 1-skeleton as follows. The two edge paths from *v* to the antipodal vertex −*v* in *F* are labeled by alternating words of length *m*, *st* ··· and *ts* ···. If we direct these edge paths as traveling from *v* to −*v*, then corresponding to an edge labeled by *s* (or *t*) there is a directed doubled edge labeled by *σ_s* (or *σ_t*). Then (*v*,*F*) is glued onto *X*¹ so that the two edge paths of *F* correspond to the positively oriented edge paths *σ_s σ_t* ··· and *σ_t σ_s* ··· in *X*¹.
- (d) The projection p: W×S → WS has a section f: WS → W×S defined by wW_T → (u,T) where u is the shortest element in the coset wW_T. This induces a continuous map f: Σ → X that is a section of p: X → Σ.

THIS IS WRONG

The 1-skeleton of X is isomorphic to the Cayley graph of (W, S) except that each edge is doubled and each of the new edges is then assigned a direction. Each doubled edge corresponds to a conjugate of the square of an Artin generator by an element of A and these conjugates give a set of generators for PA_W . By (c) the 2-cells in X correspond to conjugates of the Artin relations in (4.1). It follows that $\pi_1(X^2) = PA_W$. (This also follows from the fact that X is homotopy equivalent to the arrangement complement, $M(\mathcal{A} \otimes \mathbb{C})$.)

Definition 4.4. Let \tilde{X} be the universal cover of X.

Here are some more properties of X and \tilde{X} .

- (e) The group *W* acts freely on *X* and $\pi_1(X/W) = A_W$ (cf. Theorem 4.6).
- (f) There is an equivalence relation \sim on Σ so that $X/W = \Sigma / \sim$. (By (d) there is a surjection $\Sigma \hookrightarrow X \to X/W$ which induces the equivalence relation on Σ .
- (g) The space X/W is a finite CW complex (provided *S* is finite) with a single vertex and with one cell of dimension *k* for each spherical subset $T \in S$ with Card T = k.
- (h) The 1-skeleton \tilde{X}^1 is the Cayley graph of A_W with respect to the standard generating set $\{\sigma_s\}_{s\in S}$. Similarly, \tilde{X}^2 is the Cayley 2-complex of A_W .
- (i) The cell complex \tilde{X} satisfies the classical conditions for a convex cell complex in Section 2.1: the intersection of two cells is either empty or a common face of both.

The $K(\pi, 1)$ -Conjecture can be rewritten as follows.

Conjecture 4.2. (The $K(\pi, 1)$ -Conjecture for Artin groups, second version). The finite CW complex X/W is a model for BA_W . In other words, \tilde{X} is contractible.

4.3.3 The Deligne complex

Given an Artin group $A = A_W$, let $\mathcal{G} = \mathcal{AS}^{op}$, the complex of spherical Artin groups defined in Example 5.2. The basic constructions of Theorem 5.1 give a poset $D(A, S^{op})$ and a space D(A, K), with $K = |S^{op}|$, see (5.1). The poset $D(A, S^{op})$ is equal to the poset AS from (4.4) of spherical cosets in A. The space D(A, K) is the *Deligne complex* for A; it will be denoted by $\Lambda (= \Lambda(W, S))$. So, $\Lambda(W, S)$ is analogous to the definition of $\Sigma(W, S)$ given in Subsection 4.2.5 using the basic construction and fundamental chamber K as of Subsection 4.2.5.

The facts which are summarized in the next paragraph can be found in [23] as well as in [32].

Since *K* is simply connected, Λ can be identified with the "universal cover of the complex of groups" \mathcal{G} (see ??). As is true for any complex of groups, there is a classifying space $B\mathcal{G}$ for \mathcal{G} , together with a projection map $p: B\mathcal{G} \to K = |\mathcal{S}^{\text{op}}|$. In our case, $B\mathcal{G}$ can be identified with $\Lambda \times_A EA$ where EA denotes the universal cover of the aspherical space BA and where the projection map p is induced by projection onto the first factor $\Lambda \times EA \to \Lambda$ after dividing by A to get $\Lambda \times_A EA \to \Lambda/A = K$. The space $B\mathcal{G}$ is characterized by the fact that it is an "aspherical realization of the local groups". In our context this means that if $v_T \in |\mathcal{S}^{\text{op}}|$ denotes the vertex corresponding to T, then $p^{-1}(v_T)$ is homotopy equivalent to BA_T .

Proposition 4.5. The universal cover \tilde{X} of the Salvetti complex X(W,S) is contractible if and only if the Deligne complex $\Lambda(W,S)$ is contractible.

Proof. As in Subsection 4.3.2, the natural projection $W \times S \to WS$ induces $X \to \Sigma$. Dividing out by W we get a projection map $q: X/W \to K$. One checks that $q^{-1}(v_T) \cong X(W_T, T)/W_T$. By Deligne's Theorem 5.3 and Salvetti's Theorem 4.6, $X(W_T, T)/W_T$ is homotopy equivalent to BA_T . So, X/W also is an aspherical

4.3 Artin groups

realization of \mathcal{G} and hence, $X/W \sim B\mathcal{G}$ (where the symbol \sim stands for "homotopy equivalent". It follows that $\tilde{X} \sim (\Lambda \times EA)$. So, \tilde{X} and Λ are simultaneously contractible or not.

A corollary is that the $K(\pi, 1)$ -Conjecture holds for A if and only if A is contractible.

As explained in Subsection 4.2.5, each copy of the fundamental chamber K is cellulated by Coxeter blocks, each of which is the intersection of a Coxeter zonotope and a sector in a fan for a reflection arrangement. This induces a natural piecewise Euclidean metric on Λ analogous to the natural piecewise euclidean metric on the zonotopal complex Σ .

Conjecture 4.3. (cf. [32, Conjecture 4.4.4, p. 622]). With its natural piecewise euclidean metric Λ is CAT(0).

Using Theorem 2.1 this conjecture implies that $\Lambda(W,S)$ is contractible, Hence, by Proposition 4.5, Conjecture 4.3 implies Conjecture 4.2. The issue in Conjecture 4.3 is showing that the link of each cell of Λ satisfies the Link Condition of Definition 2.1.

When (W, S) is spherical, we define its *spherical Deligne complex* $\Lambda'(W, S)$ by using $|S_{<S}|$ as fundamental chamber instead of |S|. In other words, the fundamental domain for $A_W \curvearrowright \Lambda'(W, S)$ is the fundamental simplex $\sigma(W, S)$ for the *W*-action on the unit sphere \mathbb{S}^{n-1} in the reflection representation, $W \curvearrowright \mathbb{R}^n$. This induces the *natural piecewise spherical metric* on $\Lambda'(W, S)$. As we will explain at the end of this subsection, when (W, S) is spherical $\Lambda'(W, S)$ is a union of subcomplexes, called *apartments*, each of which is isometric to the round sphere \mathbb{S}^{n-1} . (When (W, S) is spherical, Deligne's actual definition in [64] was of a complex $\Lambda''(W, S)$ obtained by filling in each apartment of Λ' with a round ball.)

To prove Conjecture 4.3 there are three types of links of cells in Λ which must be shown to be CAT(1).

- (a) The link of a Coxeter block B_T . Such a link corresponds to the order complex of S_T , i.e., it is a sublink of L(W,S).
- (b) A link isomorphic to $\Lambda'(W_T, T)$.
- (c) A join of a link of type (a) with one of type (b)

Links of type (a) are metric flag complexes; hence, they are CAT(1) by Moussong's Lemma 4.2.4. The spherical join of two CAT(1) piecewise spherical complexes is CAT(1) (see [23, Proposition 5.15, p.,64]] or [?cd93, Appendix, Theorem A9]). So, it suffices to consider links of type (b). This leads to the following conjecture in [33, Conjecture 3, p.,6]).

Conjecture 4.4. For any spherical (W, S), the spherical Deligne complex $\Lambda'(W, S)$ is CAT(1).

So, Conjecture 4.4 implies Conjecture 4.3. (When proving the natural metric on Σ is CAT(0), the links of type (b) are replaced by Coxeter complexes of the form $D(W_T, \sigma(W_T, T))$ which are round spheres and hence, CAT(1). So, in the case of Σ , it was unnecessary to consider links of type (b) and (c).)

Nonpositive curvature and the Deligne complex was used successfully in [32] in proving the $K(\pi, 1)$ -Conjecture [32] in two special cases described below. First, Conjecture ?? is true for any (W, S) with Card $S \leq 2$.

Proposition 4.6. (cf. [32, Proposition 4.4.5]). *The spherical Deligne complex* $\Lambda'(W, S)$ *is* CAT(1) *whenever* Card $S \leq 2$.

This is proved in [32] by observing that it follows from a lemma of Appel-Schupp that for $S = \{s, t\}$, the shortest loop in $\Lambda'(W, S)$ has length 2m(s, t) corresponding to an Artin relation. A corollary is the following.

Corollary 4.1. The $K(\pi, 1)$ -Conjecture holds for A_W whenever dim $\Lambda \leq 2$.

The cubical structure on *K* defined in Subsection 4.2.5 gives Λ the structure of a cube complex denoted Λ^{\Box} . Once one shows that links of type (b) are flag complex (by [32]), the proof of Proposition 4.3 shows that cube complex Λ^{\Box} is CAT(0) if and only if L(W, S) is a flag complex.

Proposition 4.7. (cf. [32]). The cube complex Λ^{\Box} is CAT(0) if and only if (W,S) is type FC (see Definition 4.3).

First, one could define its barycentric subdivision $b\Lambda$ to be the geometric realization of the order complex of the poset AS of spherical Artin cosets (cf. (4.3)):

$$AS := \bigsqcup_{T \in S} A/A_T . \tag{4.16}$$

In the same way as Σ was defined in Subsection 4.2.5 we also could define Λ by using the basic construction with fundamental chamber K (= |S|) as a strict fundamental domain, i.e., $\Lambda = D(A, K) = (A \times K)/ \sim$, where the local group at $x \in K$ is $A_{S(x)}$. When (W, S) is spherical, one defines its *spherical Deligne complex* Λ' by using $S_{<S}$ in place of S. (When (W, S) is spherical, Deligne's actual definition in [64] was of a complex $\Lambda''(W, S)$ obtained by filling in certain round spheres in Λ' with balls, see below.) The *natural piecewise euclidean metric on* Λ is induced from the natural piecewise euclidean metric on K. This metric on K also induces the natural metric on Σ . Similarly, since the fundamental spherical simplex $\sigma(W_T, T)$ is a fundamental domain for the W_T -action on the round sphere \mathbb{S}^{k-1} (= $\mathbb{S}Fan(W_T)$), the natural piecewise spherical metric on the spherical Coxeter complex $\mathbb{S}Fan(W_T)$ is isometric to \mathbb{S}^{k-1} . In the same way, there is a *natural piecewise spherical metric on the spherical Deligne complex* $\Lambda'(W_T, T)$ induced from the fundamental spherical simplex $\sigma(W_T, T)$.

Remark 4.1. (*Apartments in* $\Lambda(W, S)$). If $s_1 \dots s_k$ is a reduced decomposition of an element $w \in W$, then the element $a_w = \sigma_{s_1} \dots \sigma_{s_k}$ is a well-defined element of A depending only on w and not on the choice of reduced decomposition for it. (This follows from the solution to the word problem for Coxeter groups by Matsumoto and Tits.) Hence, $q : w \mapsto a_w$ is a section of the canonical epimorphism $A \to W$ (The function q is only a map of sets; it is not a homomorphism). The canonical epimorphism $A \to W$ induces a simplicial projection $b\Lambda \to b\Sigma$ and q induces a simplicial section $q : b\Sigma \to b\Lambda$. Moreover, for any $a \in A$ by composing with translation with a we get another section $aq : b\Sigma \to b\Lambda$. The image of such a section is called an *apartment* of Λ (cf. [64]).

4.4 Buildings

 $d_{atte, reference}$ In the 1960s, J. Tits [133,134] developed the theory of buildings and their automorphism groups in connection with his work on incidence geometries and algebraic groups. Each building has an associated Coxeter system (W, S). For buildings that arise from algebraic groups, only spherical and euclidean Coxeter groups play a role. Nevertheless, Tits was careful to develop the theory for arbitrary Coxeter systems. In fact, it seems that Tits motivation for introducing the notion of a Coxeter system was for its use in developing the theory of buildings. Another remarkable aspect of this work is that the symmetries of a building were not involved in its definition; in particular, a building need not have the structure of a complex of groups.

References for this material include [1], [133–136], [43,49], and [48, Ch. 18]

4.4 Buildings

4.4.1 The combinatorial theory of buildings

A *chamber system* is a set \mathbb{C} together with a family of equivalence relations on \mathbb{C} indexed by another set *S*. Chambers $C, D \in \mathbb{C}$ are *s*-*adjacent* if they are *s*-equivalent and not equal. A *gallery* in \mathbb{C} is a finite sequence of chambers (C_0, \ldots, C_k) such that C_{j-1} is adjacent to C_j , for $1 \le j \le k$. The *type* of this gallery is the word $\mathbf{s} = (s_1, \ldots, s_k)$ where C_{j-1} is s_j -adjacent to C_j . If each s_j belongs to a given subset *T* of *S*, then the gallery is a *T*-gallery. Two chambers are in the *same T*-connected component if they can be connected by a *T*-gallery. The *T*-connected components of a chamber system \mathbb{C} are its *residues* of *type T*. An *s*-equivalence class is the same thing as a residue of type $\{s\}$. If $C \in \mathbb{C}$ and $T \le S$, then $\text{Res}_T(C)$ denotes the residue of type *T* containing *C*.

Example 4.7. (*The chamber system associated to a family of subgroups*). Suppose that *G* is group, that *B* is a subgroup, and that $(G_s)_{s \in S}$ a family of subgroups of *G* such that each G_s contains *B*. Define a chamber system $\mathcal{C} = \mathcal{C}(G, B, (G_s)_{s \in S})$ as follows: $\mathcal{C} = G/B$; chambers *gB* and *g'B* are *s*-adjacent if they have the same image in G/G_s .

For each subset T of S, let G_T be the subgroup generated by $\{G_s\}_{s \in T}$. As we will see in Chapter 5, if \mathcal{P} denotes the power set of S, then $\{G_T\}_{T \in \mathcal{P}^{\text{op}}}$ is a simple complex of groups over the opposite poset to \mathcal{P} .

A building is a chamber system with some extra structure that depends on a Coxeter system (W, S).

Definition 4.5. (cf. [1]). Suppose (W,S) is a Coxeter system. A *building of type* (W,S) is a pair (\mathcal{C}, δ) consisting of a nonempty set \mathcal{C} (the elements of which are called *chambers*), and a function $\delta : \mathcal{C} \times \mathcal{C} \to W$ (called the *Weyl distance*) so that the following conditions hold for all $C, D \in \mathcal{C}$

(WD1) $\delta(C,D) = 1$ if and only if C = D. **(WD2)** If $\delta(C,D) = w$ and $C' \in \mathbb{C}$ satisfies $\delta(C',C) = s \in S$, then $\delta(C',D) = sw$ or w. If, in addition, l(sw) = l(w) + 1, then $\delta(C',D) = sw$. **(WD3)** If $\delta(C,D) = w$, then for any $s \in S$ there is a chamber $C' \in \mathbb{C}$ such that $\delta(C',C) = s$ and $\delta(C',D) = sw$.

Example 4.8. The group *W* itself has the structure of a building: $\delta : W \times W \to W$ is defined by $\delta(v, w) = v^{-1}w$. It is called the *thin building* of type (W, S).

Now suppose C is a building of type (W, S). Chambers $C, D \in C$ are *s*-adjacent if $\delta(C, D) = s$; they are *s*-equivalent if they are either *s*-adjacent or equal. This gives C the structure of a chamber system. By (**WD3**) every *s*-equivalence class has at least two elements. A building C is *thick* if each *s*-equivalence class contains at least 3 elements.

When $\mathcal{C} = W$, a *T*-residue is just a left coset of W_T . A residue of type *T* in a building is itself a building; its type is (W_T, T) . A building is *spherical* if its type is a spherical Coxeter system (cf. Subsection 4.2.1). It is a *right-angled* building (a RAB) if its type is a right-angled Coxeter system (cf. Subsection 3.3). Let $R_T(\mathcal{C})$ denote the set of all residues in \mathcal{C} of type (W_T, T) . In practice the set of chambers \mathcal{C} in a building will usually be associated to a family of subgroups $\mathcal{C}(G, B, (G_s)_{s \in S})$ as in Example 4.7. When this is the case, $\mathcal{C} = G/B$ and a residue of type *T* is the image of a left coset gG_T in G/B.

Let $R(\mathcal{C})$ denote the disjoint union of all $R_T(\mathcal{C})$ with $T \in S$. The set $R(\mathcal{C})$ is called the poset of *spherical residues*. The order relation is inclusion of residues: $\operatorname{Res}_T(C) < \operatorname{Res}_{T'}(C')$ if T < T' and the first residue is a subset of the second. The geometric realization of the order complex of $R(\mathcal{C})$ is the *standard realization* of \mathcal{C} . In this realization each apartment is isomorphic to $b\Sigma(W, S)$, the barycentric subdivision of the Davis-Moussong complex in Subsection 4.2.3.

Definition 4.6. An *apartment* in \mathcal{C} is a subset which is *W*-isometric to the thin building *W*. In other words, if $\rho_C : \mathcal{C} \to W$ denotes the function $D \mapsto \delta(C, D)$, then a subset \mathcal{A} of \mathcal{C} is an apartment if $(\rho_C)|_{\mathcal{A}} : \mathcal{A} \to W$ is an isomorphism. The function $\mathcal{C} \to \mathcal{A}$, defined as the composition of ρ_C with the inverse of $(\rho_C)|_{\mathcal{A}}$, is called the *retraction onto* \mathcal{A} *centered at* C.

Definition 4.7. The *thickness* of a locally finite building \mathcal{C} of type (W, S) (at a chamber C) is the *s*-tuple $(q_s)_{s \in S}$, where each $\{s\}$ -residue containing C has exactly $q_s + 1$ elements.

If \mathbb{C} is the building for an algebraic group over a finite field \mathbb{F}_q of order q, then each $\{s\}$ -residue has q + 1 elements since it can be identified with the projective line over \mathbb{F}_q . Hence, each q_s is equal to q.

The *rank* of a building of type (W, S) is Card(S).

Example 4.9. (*Rank one buildings*). The only Coxeter system of rank one is $(\langle s \rangle, \{s\})$, where $\langle s \rangle$ means the cyclic group of order two. So, a rank one building is any set with more than two elements (the chambers) and $\delta : \mathbb{C} \times \mathbb{C} \to \langle s \rangle$ is defined by

$$\delta(C,D) = \begin{cases} 1, & \text{if } C = D, \\ s, & \text{if } C \neq D. \end{cases}$$

Example 4.10. (Rank two buildings). If $(W, \{s,t\})$ is a Coxeter system of rank two, then W is a dihedral group D_m of order 2m, where m = m(s,t), and where D_∞ means the infinite dihedral group. If $m < \infty$, then a building C of type $(D_m, \{s,t\})$ is a *generalized m-gon.* A generalized 3-gon is a *projective plane.* The spherical realization of C is a bipartite graph of girth 2m and diameter m, the edges of the graph are the chambers C. Each vertex of an edge determines a residue of type $\{s\}$ or $\{t\}$. An apartment in C is a subgraph which is a circuit of length 2m. A building C of type $(D_\infty, \{s,t\})$ is the same thing as a tree without terminal vertices, that is to say, C is the set of edges in such a tree. An apartment is a subtree isomorphic to the real line, that is, to the Coxeter complex of $(D_\infty, \{s,t\})$.

Example 4.11. (*Products*). For i = 1, 2, suppose $(\mathcal{C}_i, \delta_i)$ is a building of type (W_i, S_i) . Then $(W, S) = (W_1 \times W_2, S_1 \sqcup S_2)$ is Coxeter system and $(\mathcal{C}, \delta) = (\mathcal{C}_1 \times \mathcal{C}_2, \delta_1 \times \delta_2)$ is a building of type (W, S). For example, if each $(\mathcal{C}_i, \delta_i)$ is the building corresponding to a tree T_i , then $(\mathcal{C}_1 \times \mathcal{C}_2, \delta_1 \times \delta_2)$ is a building corresponding to $T_1 \times T_2$; its associated Coxeter group is $D_{\infty} \times D_{\infty}$. Each apartment in $T_1 \times T_2$ is isomorphic the euclidean plane with its square tiling.

If $\mathbf{s} = (s_1, \dots, s_k)$ is a word in *S*, then its *value* $w(\mathbf{s}) = s_1 \cdots s_k$ is the corresponding element of *W*. A gallery (C_0, \dots, C_k) of type \mathbf{s} is *reduced* if \mathbf{s} is a reduced expression for $w(\mathbf{s})$.

It is proved in [1, Prop. 5.23] that the conditions (WD1),(WD2), (WD3) in Definition 4.5 are equivalent to the following two conditions on a chamber system \mathcal{C} , equipped with a function $\delta : \mathcal{C} \times \mathcal{C} \to W$.

- Each *s*-equivalence class has at least two elements.
- Given a reduced expression s for an element w ∈ W, there is a gallery of type s from C to D if and only if δ(C,D) = w.

(This is the definition in [117].)

previous 3 paragraphs could be omitted or put later

An *automorphism* of C is a self-bijection which preserves *s*-equivalence classes for each $s \in S$. Equivalently, it is a self-bijection which preserves Weyl distance.

Given $C \in \mathbb{C}$, the *combinatorial ball of radius n about C* is the set $B_C(n) := \{D \in \mathbb{C} \mid l(\delta(C,D)) \le n\}$. There is a natural topology on the group Aut(\mathbb{C}) of automorphisms of \mathbb{C} : an open neighborhood of $1 \in Aut(\mathbb{C})$ is the set of automorphisms which fix each element of $B_C(n)$ for some $n \in \mathbb{N}$ and $C \in \mathbb{C}$. (The neighborhood is small if *n* is large.) Since \mathbb{C} is locally finite, Aut(\mathbb{C}) is a locally compact, totally disconnected topological 4.4 Buildings

group. As such, it has a Haar measure. A closed subgroup $G \subset \operatorname{Aut}(\mathcal{C})$ inherits a topology and a Haar measure. A subgroup $\Gamma \subset G$ is a *lattice* if it is discrete and G/Γ has finite volume. It is a *uniform* lattice if G/Γ is compact. A discrete subgroup $\Gamma \subset \operatorname{Aut}(\mathcal{C})$ is a uniform lattice if and only if \mathcal{C}/Γ is finite. (See [?thomas] for a discussion of lattices in $\operatorname{Aut}(\mathcal{C})$, when \mathcal{C} is a right angled building.)

Definition 4.8. A subgroup $G \subset Aut(\mathcal{C})$ is *chamber-transitive* if it is transitive on \mathcal{C} . It is *strongly transitive* if it is transitive on the set of pairs $(\mathcal{A}, \mathcal{C})$, where \mathcal{A} is an apartment in \mathcal{C} and $\mathcal{C} \in \mathcal{A}$ (cf. [1, §6.1.1]). (In fact, it is not necessary use all apartments in this definition, \mathcal{A} need only belong to a certain "system of apartments" satisfying the classical axioms for a building, cf. [1, §6.1]).

It turns out that if G is strongly transitive on a thick building, then it inherits the structure of a BN pair (also called a "Tits system"), cf. [1, Thm. 6.56].

4.4.2 Geometric realizations of buildings

Given a Coxeter system (W, S), let $K = |S^{op}|$ be its fundamental chamber. Even though C need not be a group, the definition of the basic construction can be modified to work for C. The definition of D(C, K) is essentially the same as that for the standard complex $\Sigma(W, S)$ in Subsection 4.2.3 or for the Deligne complex $\Lambda(W, S)$ in Subsection 4.3.2. To wit, $D(C, K) = (C \times K) / \sim$, where the equivalence relation \sim is given by

 $(C,x) \sim (C',x') \iff x = x'$, and the chambers *C*, *C'* belong to the same *S*(*x*)-residue.

(Here, as before, $S(x) = \{s \in S | x \in K_s\}$).) The polyhedron $D(\mathcal{C}, K)$ is the *standard realization of* \mathcal{C} . Since *K* is the geometric realization of \mathcal{S} , $D(\mathcal{C}, K)$ can be identified with $|R(\mathcal{C})|$, the geometric realization of the poset of spherical residues.

As in Subsection 4.2.5, *K* is partitioned into Coxeter blocks B_T , with $T \in S$, where B_T is the union of all simplices in the order complex of the subposet $[T, \emptyset]$ of S^{op} ; furthermore, each B_T has a natural metric as a convex polytope in some euclidean space. (Recall B_T is the intersection of a Coxeter zonotope with a W_T -sector.) This induces the natural piecewise euclidean metric on *K*, as well as, the *natural piecewise euclidean metric on* $D(\mathcal{C}, K)$. Moreover, each apartment in $D(\mathcal{C}, K)$ is isometric to $D(W, K) = \Sigma(W, S)$, with its CAT(0) metric. To simplify notation denote the standard realization $D(\mathcal{C}, K)$ with its natural piecewise euclidean metric by $|\mathcal{C}|$.

Theorem 4.7. (cf. [43]). For any building C, the natural piecewise Euclidean metric on its standard realization is CAT(0).

Suppose (W_T, T) is a spherical Coxeter system as in Subsection 4.2.1. The group W_T acts on the unit sphere \mathbb{S}^{m-1} in the geometric representation on \mathbb{R}^m . A fundamental domain for the action is the fundamental spherical simplex $\sigma(T)$; so, $\mathbb{S}^{m-1} = D(W_T, \sigma(T)$ (see Subsection 4.2.1). Now suppose $\mathcal{C}(T)$ is a spherical building of type (W_T, T) . The *spherical realization* of $\mathcal{C}(T)$ is the basic construction applied to $\mathcal{C}(T)$ with fundamental domain $\sigma(T)$:

$$D(\mathcal{C}(T), \sigma(T)) = (\mathcal{C}(T) \times \sigma(T)) / \sim$$
,

with its natural piecewise spherical metric induced from $\sigma(T)$. We shall also write $\Delta(\mathcal{C}(T) \text{ for } D(\mathcal{C}(T), \sigma(T))$. In this spherical realization, each apartment is isometric to the round sphere $D(W_T, \sigma(T)) = \mathbb{S}^{m-1}$. More precisely, if \mathcal{A} is an apartment in $\mathcal{C}(T)$, then the map induced by $(\rho_C)|_{\mathcal{A}}$ from Definition 4.6 induces an isometry from the spherical realization of \mathcal{A} to \mathbb{S}^{m-1} . (N.B. The standard realization of $\mathcal{C}(T)$ uses for fundamental domain the Coxeter block B_T so that each apartment becomes isometric to the Coxeter zonotope $Z(W_T, T)$.)

Consider the standard cellulation of $\Sigma(W, S)$ by Coxeter blocks. Let v_T be a vertex corresponding to the center of a Coxeter zonotope $Z(W_T, T)$, then the link of v_T in $\Sigma = \Sigma(W, S)$ decomposes as a spherical join:

$$Lk(v_T, \Sigma) = \mathbb{S}^{m-1} * Lk(T, |S|).$$
(4.17)

Here \mathbb{S}^{m-1} is the link of v_T in the zonotope and $Lk(T, |S|) = |S_{>T}| = Lk(v_T, K_T)$ is the the link of v_T in the stratum of a chamber corresponding to *T*. Since $|S_{>T}|$ is a link in the metric flag L(W,S), it is also a metric flag complex; so, by Moussong's Lemma 4.3, it is CAT(1). A similar analysis holds for the link of a vertex v_T in the geometric realization $|\mathcal{C}| = D(\mathcal{C}, K)$ except that in (4.17), the round sphere \mathbb{S}^{m-1} must be replaced by the spherical realization of a building of the form $\operatorname{Res}_T(C)$. In other words, the formula in (4.17) should be replaced by

$$Lk(v_T, |\mathcal{C}|) = \Delta(\mathcal{C}(T)) * Lk(T, |\mathcal{S}|), \qquad (4.18)$$

where $\Delta(\mathcal{C}(T)) = D(\operatorname{Res}_T(C), \sigma(T))$. So, to show that $\operatorname{Lk}(v_T, |\mathcal{C}|)$ is CAT(1) it suffices to prove the following lemma which states that the spherical realization of any spherical building is CAT(1).

Lemma 4.7. (cf. [43]). If C(T) is a spherical building of type (W_T, T) , then its spherical realization $\Delta(C(T))$ is CAT(1).

Lemma 4.7 is a corollary of Theorem 4.7. Indeed, a neighborhood of the vertex v_T in $|\mathcal{C}(T)|$ is isometric to a neighborhood of the cone point in Cone($\Delta(\mathcal{C}(T))$). By Lemma 2.2, if v_T has a CAT(0) neighborhood, then $\Delta(\mathcal{C}(T))$ is CAT(1).

Remark 4.2. If Lemma 4.7 holds for all spherical building, then, by the Link Condition (Definition 2.1), $|\mathcal{C}|$ is NPC. Since it is also not hard to see that $|\mathcal{C}|$ is simply connected, it might seem that the way to prove Theorem 4.7 is to first establish Lemma 4.7. However, the proof in [43] is by a more direct argument which we shall explain below.

Proof (Sketch of proof of Theorem 4.7). The proof in [43] follows a standard argument in [?brbook, Ch. VI §3]. An important consequence of Definition 4.5 is one of the classical axioms for a building: any two chambers are contained in a common apartment. Passing to the geometric realization, this that any two points $x, y \in |\mathcal{C}|$ are contained in the realization of $|\mathcal{A}|$ of an apartment. If *C* is a chamber in an apartment \mathcal{A} , we have $\rho_{C,\mathcal{A}} : \mathcal{C} \to \mathcal{A}$ the retraction onto \mathcal{A} centered at *C* (cf. Definition 4.6. Its geometric realization is denoted $\overline{\rho}_{C,\mathcal{A}} : |\mathcal{C}| \to |\mathcal{A}|$ (or simply by $\overline{\rho}$). If $x, y \in |\mathcal{A}|$, then $\overline{\rho}$ takes a geodesic segment connecting them in $|\mathcal{C}|$ to a geodesic segment in $|\mathcal{A}|$. Using the fact that $|\mathcal{A}|$ is CAT(0) one argues that the geodesic segment in $|\mathcal{C}|$ from x to y is equal to the geodesic segment in $|\mathcal{A}|$. Moreover, $\overline{\rho}$ is distance decreasing.

If *x*, *y*, *z* are vertices of a triangle in \mathbb{E}^2 and $p_t = tx + (1-t)y$ is a point on the segment from *x* to *y*, then it is an easy exercise to see that $d^2(z, p_t) = (1-t)d^2(z,x) + td^2(z,y) - t(1-t)d^2(x,y)$, where $d^2(x,y)$ denotes the square of the euclidean distance. It follows that the CAT(0) inequality for a triangle in a geodesic space with vertices *x*, *y*, *z* is equivalent to the same formula with the equality replaced by an inequality:

$$d^{2}(z, p_{t}) \leq (1-t)d^{2}(z, x) + td^{2}(z, y) - t(1-t)d^{2}(x, y)$$
(4.19)

Now choose an apartment \mathcal{A} so that $x, y \in |\mathcal{A}|$ and a chamber C so that $p_t \in |C|$ and let $\overline{\rho} : |\mathcal{C}| \to |\mathcal{A}|$ be the geometric realization of $\rho_{C,\mathcal{A}}$. Then

4.4 Buildings

$$d^{2}(z, p_{t}) = d^{2}(\overline{\rho}(z), p_{t})$$

$$\leq (1-t)d^{2}(\overline{\rho}(z), x) + td^{2}(\overline{\rho}(z), y) - t(1-t)d^{2}(x, y)$$

$$\leq (1-t)d^{2}(z, x) + td^{2}(z, y) - t(1-t)d^{2}(x, y) .$$

Hence, the CAT(0)-inequality holds for the triangle [x, y, z].

One can also use this argument to decide when the standard realization of a building has a piecewise hyperbolic, CAT(-1) metric. This gives the following analog of Theorem 4.3.

Theorem 4.8. (cf. [43, Remark 11.8]). Suppose C is a building of type (W,S), with W word hyperbolic. Then its standard realization D(C,K) can be given a piecewise hyperbolic, CAT(-1) metric.

For example, if W is a reflection group on \mathbb{H}^2 generated by reflections across the faces of a hyperbolic polygon, then C is a *Fuchsian bulding*. In the right-angled case these have been constructed and studied by Bourdon [20,21].

4.4.3 Using covering spaces to construct buildings

In Section 3.1 we gave a technique for constructing RACGs, RAAGs and RAB, that involved first taking a polyhedral product and then passing to the universal cover. Here we discuss a generalization of [49] that works for arbitrary Coxeter systems not just the right-angled ones. Suppose (W', S') is a Coxeter system with nerve L' and that L is another simplicial complex with vertex set S. Suppose that $f : S \to S'$ defines a simplicial map $L \to L'$ whose restriction to each simplex is injective. Let (m'(s', t')) denote the Coxeter matrix of (W', S'). Define a new $(S \times S)$ Coxeter matrix (m(s,t)) by

$$m(s,t) := \begin{cases} 1 & \text{if } s = t, \\ m'(f(s), f(t)) & \text{if } \{s,t\} \in \text{Edge}(L), \\ \infty & \text{otherwise,} \end{cases}$$
(4.20)

and let (W, S) be the corresponding Coxeter system. in other words, if m': Edge $(L') \rightarrow \{2.3,...\}$ is the edge labeling defining (W', S'), then (W, S) is defined by the edge labeling m which is the composition m of m' with $f|_{\text{Edge}(L')}$, i.e., m = m'f: Edge $(L) \rightarrow \{2,3,...\}$. The map of generating sets $S \rightarrow S'$ extends to a homomorphism $\varphi_f : W \rightarrow W'$; moreover, φ_f is surjective if and only if f is surjective.

For example, $f: L \to L'$ could be

- (a) a covering projection,
- (b) an embedding onto a subcomplex, or
- (c) a covering space of an embedded subcomplex.

Let K(L') denote the fundamental chamber for (W', S'), i.e., K(L') is the geometric realization of the order complex of S(L'). Similarly, if S(L) denotes the poset of simplices of L, then K(L) = |S(L)| is the fundamental chamber for a W-action on D(W, K(L)). The space D(W, K(L)) need not be contractible: a necessary and sufficient condition for this to be true is that for every spherical subset $T \in S(W, S)$, T is the vertex set of a simplex of L. The map f can be used to get a new stratification of K(L) indexed by S(L'):

4 Coxeter groups, Artin groups, buildings

$$K(L)_{s'} := \bigsqcup_{s \in f^{-1}(s')} K(L)_s \quad \text{and} \quad K(L)_{T'} = \bigsqcup_{T \in f^{-1}(T')} K(L)_T.$$
(4.21)

Let D(W', K(L)) be the result of applying the basic construction to this new stratification of K(L). If $W \neq W'$, then D(W', K(L)) will not be simply connected. Indeed, a necessary condition for it to be so is that $K(L)_{s'}$ is connected for each $s' \in S'$ and that $K(L)_{\{s',t'\}} \neq \emptyset$ whenever $m'(s',t') \neq \infty$ (see [48, Prop. 8.2.11]). However, the natural φ_f -equivariant map, $D(W, K(L)) \rightarrow D(W', K(L))$, is projection map of the universal covering. Similarly, for A' the Artin group associated to (W', S'), we get a version of the Deligne complex D(A', K(L)). More interesting is the case of buildings. Given a building C' we get a building of type (W', S'), we get a nonstandard realization D(C', K(L)). The main result of [49] is the following.

Theorem 4.9. ([49]). With notation as above, the universal cover of $D(\mathcal{C}', K(L))$ is the geometric realization of a building \mathcal{C} of type (W, S).

Example 4.12. (RACG*s*, RAAG*s*, RAB*s*). Here are the examples of type (b) which were considered previously in Section 3.1. Suppose that *L* is a flag complex with vertex set *S*, that S' = S and that $\Delta(S)$ is the simplex on *S*. Then *L* is a subcomplex of $\Delta(S)$ and K(L) is a cubical subcomplex of the cube $[0, 1]^S$. Let $W' = (\mathbf{C}_2)^S$ be the sum of cyclic 2-groups. Then $D((\mathbf{C}_2)^S, K(L))$ is the polyhedral product P_L of (3.3), $W = \prod_{L^1} \mathbf{C}_2$ is the RACG associated to L^1 in Example 3.3 and $D(W, K(L)) = \tilde{P}_L = \Sigma(W, S)$ is the Davis-Moussong complex of Subsection 4.2.3. If $A' = \sum_S \mathbb{Z}$ (= $(\mathbb{Z})^S$), then $D(\sum \mathbb{Z}, K(L)$ is the polyhedral product $Z_L = (\text{Cone } \mathbb{Z}, \mathbb{Z})^L$ of (3.4) and if $A = \prod_{L^1} \mathbb{Z}$ denotes the associated RAAG of Example 3.3, then $D(A, K(L)) = \tilde{Z}_L = \Lambda(W, S)$, the Deligne complex of Subsection 4.3.3 ($\Lambda(W, S)$ is also the RAB for *A*). Finally, if $\mathbf{E} = \{E_s\}_{s \in S}$ is a family of discrete sets, each having more than two elements, then $C_s = E_s$ is a rank one building and the product $C' = \prod C_s$ is a building of type (($\mathbf{C}_2)^S$, S). The thickness (q_s) is the S-tuple defined by $q_s + 1 = \text{Card}(E_s)$. Moreover, D(C', K(L)) is the polyhedral product ($\text{Cone } \mathbf{E}, \mathbf{E})^L$ of (3.4) and its universal cover is the geometric realization a RAB C of type (W, S).

Definition 4.9. A *coloring* of a simplicial complex *L* is a simplicial map $f : L \to \Delta(S')$ whose restriction to each simplex is injective. The elements of *S'* are the *colors*. (The map *f* gives a coloring of the vertices of *L* by elements of *S'*.)

Example 4.13. (*Coxeter systems of type FC*). Here is a generalization of the previous example. Suppose (W',S') is spherical so that $L' = \Delta(S')$ is the simplex on S'. Suppose L is a flag complex with vertex set S and that $f: L \to \Delta(S'')$ is a coloring. The resulting Coxeter system (W,S) is type FC (see Definition 4.3). Conversely, if (W,S) is type FC, then there is an inclusion $f: L(W,S) \hookrightarrow \Delta(S)$ (in particular, f is a coloring) and one can realize $\Delta(S)$ as the nerve of a spherical Coxeter system (W',S) by the simple expedient of labeling by 2 all edges of $\Delta(S)$ which are missing from L, i.e., if $m(s,t) = \infty$, then change it to m'(s,t) = 2. Thus, every Coxeter system of type FC is the pullback of a spherical Coxeter system via some coloring. The complex D(W', K(L)) plays the role of the polyhedral product P_L and the natural map $D(W, K(L)) \to D(W', K(L))$ is the universal cover. Thus, $\pi_1(D(W', K(L)))$ is a torsion-free subgroup of finite index in W. As before, (W', S') and (W, S) have respective Artin groups A' and A. Not all Coxeter systems occur as the type of a locally finite, thick spherical building; however, the irreducible Coxeter systems with diagrams A_n , B_n , D_n , G_2 , F_4 , E_6 , E_7 , and E_8 all can occur. For spherical buildings of rank two whose Coxeter group is a dihedral group of order 2m (= a generalized m-gon), by a theorem of Feit-Higman [?fh] there are locally finite examples only for $m \in \{2,3,4,6,8\}$

Example 4.14. (The case dim L = 1, *Fuchsian buildings).* If C' is a generalized *m*-gon, then $(W', S') = (D_m, \{s', t'\})$ and L' is the 1-simplex $\Delta(S')$. If L is any bipartite simplicial graph, then it admits a coloring

4.4 Buildings

 $f: L \to \Delta(S')$. For example, *L* could be a 2*k*-gon or a finite tree. So, for any bipartite graph *L*, there is a labeling of Edge(*L*) where each edge is labeled by the same integer *m*. If $m \in \{2, 3, 4, 6, 8\}$, we can choose a thick rank two spherical building C' of this type. Hence, if *L* is any bipartite graph, there is a building C such that each rank two spherical residue is isomorphic to C'. In other words, the link of each vertex of K(L) is isomorphic to the 1-dimensional piecewise spherical complex $D(C', \sigma(S'))$. (A vertex of K(L) corresponds to the midpoint of an edge of *L*.) If *L* is a 2*k*-gon and if $m \ge 3$ when 2k = 4, then K(L) can be realized as a 2*k*-gon in \mathbb{H}^2 ; so, C is a Fuchsian building.

We can get many more examples with dim L = 1 by starting with (W', S') a spherical Coxeter system (in which case $L' = \Delta(S')$) or possibly a euclidean Coxeter system (in which case $L' = \partial \Delta(S')$). To simplify the discussion let us suppose L is an n-gon. The simplicial map $f : L \to \Delta(S')$ is just a closed edge path, possibly with backtracking. If (W', S') has diagram \widetilde{A}_{n-1} , then $\partial \Delta(S')$ contains an n-circuit with each edge labeled 3. Let $f : L \to \partial \Delta(S')$ be an embedding with image this n-circuit. If C' is a building of type (W', S'), then over each edge of f(L) we have a rank two spherical residue with diagram A_2 , i.e., a projective plane. In particular, L can be an n-gon with n odd and each edge labeled 3. The resulting building C over K(L) will be Fuchsian provided n > 3. In this fashion we can find examples of buildings with L an n-gon whose edges are labeled fairly randomly by elements of $\{2, 3, 4\}$. We note that when C' is an irreducible spherical building corresponding to discrete valuation ring with residue field \mathbb{F}_q , then C' as well as the resulting building C will have constant thickness q.

Example 4.15. (*Branched covers*). Suppose L' is the nerve of a Coxeter system (W', S') and that $f : L \to L'$ is a regular covering space with group of deck transformations Γ . The simplicial structure on L' lifts to a simplicial structure on L - a simplex of L is a component of the inverse image of a simplex of L'. We can pull back the labeling on Edge(L') to get a labeling on Edge(L) and hence, a Coxeter system (W, S) where $S = f^{-1}(S')$. By construction, $L \leq L(W, S)$ and an easy argument shows that $L(W, S) \leq L$ (use the fact that each triangle in L projects to a triangle in L'). Since K(L') = Cone L' and K(L) = Cone L, the natural projection $K(L) \to K(L')$ is a branched cover (branched at the cone point). The stratification of K(L') indexed by $\Im(W', S')$ lifts to one on K(L) with the same index set; K(L) also has a natural stratification indexed by $\Im(W, S)$.

Example 4.16. (Fuchsian buildings).

Example 4.17. (Hyperbolic buildings).

4.4.4 Constructions, lattices

automorphism groups, lattices, Algebraic groups, Simple complexes of groups, (B,N) pairs

I doubt it

Chapter 5 Simple complexes of groups

Suppose a group G acts on a space X. Two points of X belong to the same *orbit type* if their isotropy groups are conjugate subgroups of G. The points of a given orbit type form a *pure stratum* of X. (In fact we will want to define a pure stratum to be a connected component of a set of points of a given orbit type.) The closure of a pure stratum is a *stratum*. So, X is a stratified space as in [23, II.12.1].

A *strict fundamental domain* for $G \cap X$ is a closed subspace $C \subset X$ such that *C* intersects each orbit in exactly one point. It follows that restriction of the orbit projection $p: X \to X/G$ to the subspace *C* is a homeomorphism and hence, that *C* provides a section $s: X/G \to X$ of *p* defined by $s = (p|_C)^{-1}$. The stratification of *X* induces a stratification of *C*. This means that two points of *C* belong to the same pure stratum if and only if their isotropy subgroups are equal. Similarly, if *G* acts on a poset \mathcal{P} by order-preserving automorphisms, then a *strict fundamental domain* for *G* on \mathcal{P} is a subposet Ω which intersects each *G*-orbit in exactly one element. Simple complexes of groups are designed to deal with group actions on cell complexes which admit a strict fundamental domain.

The basic reference for the following material is [23]. Other references include [32], [60], [68], [116].

The theory of simple complexes of groups is developed in [23]. Basic examples of simple complexes of groups are provided by Coxeter groups, Artin groups and buildings. So, we will make a few comments concerning the general theory here and postpone a fuller discussion to Appendix **??**

Let \mathcal{P} be a poset. A *chain* in \mathcal{P} is a totally ordered subset. In other words, a finite subset $\{p_0, \dots, p_k\}$ is a chain if, possibly after reordering, $p_0 < \dots < p_k$. The *order complex* of \mathcal{P} is the set of all chains in \mathcal{P} . It is an abstract simplicial complex; a *k* simplex is a chain of length k + 1. Denote the geometric realization of the order complex by $|\mathcal{P}|$. Reversing the order relation on \mathcal{P} one obtains the *opposite poset* \mathcal{P}^{op} . Note that the order complexes of \mathcal{P} and \mathcal{P}^{op} are identical

Simple complexes of groups arise from group actions with strict fundamental domains. Suppose $G \cap Y$ with strict fundamental domain $|\Omega|$. The space $|\Omega|$ has a stratification with strata indexed by Ω , where $|\Omega|_{\sigma} = |\Omega_{\leq \sigma}|$. (For example if Ω is the poset of cells in a cell complex, then each stratum is a cell.) The local groups are the stabilizers of the strata (one assumes the *G*-action is without inversions meaning that the setwise stabilizer of a stratum fixes each point of the stratum. So, G_{σ} is the isotropy subgroup of $|\Omega|_{\sigma}$. (The reason a simple complex of groups is defined as a cofunctor rather than a functor is that for group actions, a smaller stratum should correspond to a larger isotropy subgroup.) The simple complex of groups is *developable* if it is induced from a *G*-action.

The next result uses the basic construction of Subsection 3.1.1 to show that every developable GQ arises from an action with strict fundamental on the basic construction.

5 Simple complexes of groups

Theorem 5.1. (The basic construction, cf. [23]). Suppose that GQ is developable and $G = \lim GQ$. Then there is a poset D(G,Q) such that $G \curvearrowright D(G,Q)$ with strict fundamental poset Q. Taking geometric realizations, we get a polyhedron D(G,|Q|) such that $G \curvearrowright D(G,|Q|)$ with strict fundamental domain |Q|.

The basic construction D(G, |Q|),

$$D(G, |\Omega|) = (G \times |\Omega|) / \sim$$
(5.1)

is defined as in Subsection 3.1.1. Given $x \in |\Omega|$, let $\sigma(x)$ be the index the smallest stratum $|\Omega|_{\sigma}$ such that $x \in |\Omega|_{\sigma}$. the equivalence relation \sim is then defined the same way as in (3.8):

$$(g,x) \sim (g',x') \iff x = x' \text{ and } g^{-1}g' \in G_{\sigma(x)}$$
, (5.2)

in other words, $gG_{\sigma(x)}$ and $g'G_{\sigma(x)}$ are the same coset in $G/G_{\sigma(x)}$.

It also turns out that when $|\Omega|$ is simply connected, then so is $D(G, |\Omega|)$ (see ??). So, when $|\Omega|$ is simply connected, $G = \lim G\Omega$ should be regarded as the "fundamental group of $G\Omega$ ".

Given a Coxeter system (W, S), we define two simple complexes of groups over S^{op} , where S = S(W, S), the poset of spherical subsets of *S*. Denote the order relation on S^{op} by \prec , so that $T' \prec T$ if *T* is a proper subset of *T'*. The *fundamental chamber K* is defined as $|S^{op}|$. The stratum K_T is the union of all simplices in $|S^{op}|$ with maximum vertex = *T*.

Example 5.1. (*The complex of spherical Coxeter subgroups of* W.) There is a simple complex of groups WS^{op} over S^{op} called the *complex of spherical subgroups of* W. It is defined as the cofunctor $T \mapsto W_T$, so that if $T' \prec T$, then W_T is a subgroup of $W_{T'}$. It is obvious that the direct limit, $\lim WS^{op}$, is equal toW. On the level of posets the basic construction $D(W, S^{op}$ is equal to the poset of spherical cosets WS defined in (4.3). On the level of spaces it is the cell complex $\Sigma(W, S)$ of Subsection 4.2.3 (actually, the strata of D(W, K) are dual cones to the cells in $\Sigma(W, S)$).

Example 5.2. (The complex of spherical Artin subgroups of A.) The *complex of spherical subgroups of A* is the simple complex of groups \mathcal{AS}^{op} over \mathcal{S}^{op} defined by $T \mapsto A_T$. If $T' \prec T$, then A_T is a subgroup of $A_{T'}$. We also have $\lim \mathcal{AS}^{op} = A$. On the level of posets, $D(A, \mathcal{AS}^{op}) = AS$ where AS is poset of spherical cosets from (4.4) and on the level of spaces, D(A, K) is the Deligne complex $\Lambda(W, S)$ of Subsection 4.3.3 below.

5.1 Definitions

Definition 5.1. (cf. Bridson-Haefliger [23, II.12.11, p. 375]). A simple complex of groups $G\Omega$ over a poset Ω is a collection of groups $\{G_{\sigma}\}_{\sigma\in\Omega}$ and monomorphisms $\phi_{\tau\sigma}: G_{\sigma} \to G_{\tau}$ defined whenever $\tau < \sigma$. The G_{σ} are the *local groups*. Furthermore, $G\Omega$ must be a cofunctor from Ω to the category of groups and monomorphisms in the sense that $\phi_{\mu\tau}\phi_{\tau\sigma} = \phi_{\mu\sigma}$ whenever $\mu < \tau < \sigma$. (A "cofunctor" means a contravariant functor i.e., a functor on Ω^{op} .)

A simple morphism $\Psi = (\Psi_{\sigma})$ from $G\Omega$ to a group H is a function which assigns to each $\sigma \in \Omega$ a homomorphism $\Psi_{\sigma} : G_{\sigma} \to H$ such that $\Psi_{\sigma} = \Psi_{\tau} \phi_{\tau\sigma}$ whenever $\tau < \sigma$. The simple morphism Ψ is *injective* on local groups if each Ψ_{σ} is injective. If H acts on a poset \mathcal{P} with strict fundamental domain Ω , we get a simple complex of groups $G\Omega$ by setting G_{σ} equal to the isotropy subgroup at σ for each $\sigma \in \Omega$ as well as, a simple morphism $\Psi' : G\Omega \to H$ such that $\Psi' : G_{\sigma} \to H$ is the inclusion. Similarly, if H acts on a polyhedron D with strict fundamental domain the geometric realization of Ω , we get a simple complex of groups $G\Omega$

5.1 Definitions

where G_{σ} is the isotropy subgroup of H at a point in the pure stratum of the fundamental domain $|\Omega|$. If the simple complex of groups $G\Omega$ arises from an action on a poset \mathcal{P} or a polyhedron D, then we say that $G\Omega$ is *developable* and that D is a *development*. Conversely, as we shall see below in Theorem 5.2 that $G\Omega$ is developable if there is a simple morphism $G\Omega$ to some group H that is injective on local groups (in fact we can restrict H to being the direct limit of the system of groups $G\Omega$, as defined below.

The *direct limit* of GQ is a group $G (= \lim GQ)$ together with a simple morphism $\psi : GQ \to G$ with the following universal property: if $\psi' : GQ \to H$ is any simple morphism, then there exists a unique homomorphism $\theta : G \to H$ such that $\theta \psi_{\sigma} = \psi' : G_{\sigma} \to H$ for all $\sigma \in Q$. It follows that if it exists, the direct limit is unique up to a canonical isomorphism. To establish existence we note that he direct limit can be constructed in a standard fashion by taking the free product of the G_{σ} and then quotienting by relations to insure that G_{σ} is identified with its image under $\phi_{\tau}\sigma$ in G_{τ} and that $\phi_{\mu\tau}\phi_{\tau\sigma}(g) = \phi_{\mu\sigma}(g)$ whenever $\mu < \tau < \sigma$, for all $g \in G_{\sigma}$ (cf. [128, p. ?].

Remark 5.1. The usual way to regard a poset Ω as a category is have an arrow, $\tau \to \sigma$, whenever $\tau \leq \sigma$. When Ω is a set of subsets (e.g. a set of cells in a cell complex), it also is usual for the partial order to be given by inclusion. Suppose *G* acts on a cell complex *X*, that \mathcal{P} is the poset of cells in *X*, and that for each $\sigma \in \mathcal{P}$, G_{σ} is the (pointwise) stabilizer of σ . If $\tau < \sigma$ is a face of σ , then $G_{\sigma} < G_{\tau}$, i.e., "smaller cells have larger isotropy groups." For example, in the theory of graphs of groups [128], edge stabilizers are included in vertex stabilizers. If a cell complex *Y* is a strict fundamental domain for the *G*-action on *X* and Ω is its poset of cells, then we get a simple complex of groups $G\Omega = \{G_{\sigma}\}_{\sigma \in \Omega}$ where the $\phi_{\sigma\tau} : G_{\sigma} \to G_{\tau}$ are inclusions. This explains why $G\Omega$ is defined to be a cofunctor.

On the other hand, in some earlier papers, e.g., [32, 60], the author has used the reverse convention that a simple complex of groups is a functor from a poset to groups. The reason for this reversed convention is that for the complexes of groups associated to Coxeter groups and Artin groups in Section 5.3 the poset which naturally indexes the local groups is the poset of simplices in a certain simplicial complex *L* (so that smaller simplices correspond to smaller local groups). Here we will dispense with this reversed convention by using as index set, $S(L)^{\text{op}}$, the opposite poset to the face poset of *L*.

A simple morphism $\Psi = (\Psi_{\sigma})$ from GQ to a group G is a function which assigns to each $\sigma \in Q$ a homomorphism $\Psi_{\sigma} : G_{\sigma} \to G$ such that $\Psi_{\tau} = \Psi_{\sigma}\phi_{\sigma\tau}$ whenever $\tau < \sigma$. The simple morphism Ψ is *injective* on local groups if each Ψ_{σ} is injective. If there is a simple morphism Ψ from GQ to G, which is injective on local groups, then, as we shall see in Section 5.2, we get a G-action on a polyhedron D with GQ the associated complex of groups. If this is the case, GQ is said to be "developable" and D is a "development" of it. As in [128, p.?], one can define the direct limit of any system of groups and homomorphisms. So, given a simple complex of groups GQ we get a group, $\lim GQ$, called its *direct limit*. It comes equipped with a canonical morphism $\Psi : GQ \to \lim GQ$. The direct limit is characterized by a universal property: if $\Psi : GQ \to H$ is any simple morphism to a group H, then there is a unique homomorphism $\lim GQ \to H$ compatible with the canonical simple morphism. The simple complex of groups GQ is *developable* if the canonical homomorphism $GQ \to \lim GQ$ is injective on local groups.

The *order complex* of a poset \mathcal{P} is the simplicial complex whose simplices are the totally ordered finite subsets $\{\tau_0, \ldots, \tau_k\}$ in \mathcal{P} (where $\tau_0 < \cdots < \tau_k$). The order complex of a poset \mathcal{P} is a simplicial complex whose underlying topological space is denoted $|\mathcal{P}|$ and is called the *geometric realization of* \mathcal{P} . If \mathcal{P}^{opp} denotes the opposite poset where the order relations are reversed, then $|\mathcal{P}^{opp}|$ is isomorphic to $|\mathcal{P}|$. There are two natural stratifications of $|\mathcal{P}|$ both indexed by \mathcal{P} :

$$|\mathcal{P}|_{\sigma} := |\mathcal{P}_{\geq \sigma}| \quad \text{and} \quad |\mathcal{P}|^{\sigma} := |\mathcal{P}_{\leq \sigma}|.$$
 (5.3)

before simple morphism

Put some place else? Change (5.3)

In the second case, inclusion of one stratum into another corresponds to the original order relation on \mathcal{P} ; in the first case, the order relation is reversed. So, we should regard $\{|\mathcal{P}|_{\sigma}\}$ as being indexed by \mathcal{P}^{op} . Given $x \in |\mathcal{P}|$, let $\sigma(x)$ be the index of the smallest stratum $|\mathcal{P}|_{\sigma}$ containing *x*.

If *L* is a simplicial complex, then S(L) denotes its poset of simplices together with \emptyset (the empty simplex). The dual of a simplicial polytope is a simple polytope.

This seems wrong. Maybe ok This seems wrong. Maybe ok Definition 5.2. (Simplices of groups). Suppose σ is a simplex and $\partial \sigma$ is its boundary complex. Put $\Omega = S(\partial \sigma)$. The geometric realization of Ω can then be identified with the dual simplex σ^* . A simple complex of groups $G\Omega$ over Ω is called a *simplex of groups*. If dim $\sigma = 1$, then σ^* is a 1-simplex and $G\Omega$ is an *edge of groups*. If dim $\sigma = 2$, then $G\Omega$ is a *triangle of groups*.

switch σ , σ^* and P, P^* ?

Did I reverse something?

If GQ is an edge of groups, then $\lim GQ$ is the amalgamated product of G_{v_0} and G_{v_1} along G_{\emptyset} . Here v_0 and v_1 are the vertices of $\partial \sigma$ (= the vertices of σ^*) and G_{\emptyset} corresponds to the 1-simplex σ^* .

Nice examples of simplices of groups come from the theory Coxeter groups (see section 5.3). For example, when dim $\sigma = 2$, the triangle of groups pictured below the direct limit is a Coxeter group *W*. (In this figure, D_m stands for the dihedral group of order 2m. The group *W* is finite if and only if 1/p + 1/q + 1/r > 1.) Other examples of simplices of groups come from spherical buildings with chamber transitive automorphism groups (see Example 5.6.

Definition 5.3. (*Polytopes of groups*). Suppose *P* is a simplicial polytope and ∂P is its boundary complex (so ∂P is a simplicial complex). Put $\Omega = S(\partial P)$. Let P^* be the dual simple polytope (meaning that the poset of nonempty faces of P^* isomorphic to $S(\partial P)^{op}$). A simple complex of groups $G\Omega$ over $\Omega (= S(\partial P))$ is a *polytope of groups*. If dim P = 2, then $G\Omega$ is a *polygon of groups*.

Definition 5.4. (*Trees of groups*). Given a tree X, let $Q = \text{Vert} X \bigsqcup \text{Edge} X$ be its set of cells, partially ordered by reverse inclusion. A *tree of groups* is a simple complex of groups over Q. In fact, exactly the same definition works for any connected graph X without loop, yielding the notion of a graph of groups.

Suppose GQ is a graph of groups as in the above definition. Put $G = \lim GQ$. It follows from a result in [128] on the structure of amalgams that the natural simple morphism from GQ to G is injective on local groups, i.e., GQ is developable. (Even when X is allowed to have loops, one can still define a "graph of groups" as in [128, 4.4]; it is still a complex of groups as in [23, ?]; however, it is not a simple complex of groups. The difference is that when X is a loop there are two inclusions from the edge group into the vertex group and the local groups need not inject into the direct limit. However, there is still a notion of a fundamental group of a graph of groups is developable. For example, If X is a single loop, then the fundamental group is an HNN extension, cf. Example 5.11 in section 5.6.

5.2 The basic construction

move earlier We begin by reviewing material on simple complexes of groups from [23, II.12], [32], [60].

Theorem 5.2. (The basic construction, cf. [23]). Suppose $\psi' = (\psi'_{\sigma})$ is a simple morphism from GQ to a group H that is injective on local groups. Also, let G be the direct limit of GQ and $\psi : GQ \to G$ the canonical simple morphism.

(i) There is a H-action on a poset $D(\psi', Q)$ with Q as strict fundamental domain. (The poset $D(\psi', Q)$ is called the development of GQ with respect to ψ .)

5.2 The basic construction

- (ii) There is a H-action on a polyhedron $D(\Psi', \Omega)$ with strict fundamental domain $|\Omega|$.
- (iii) (Compare [?vinberg].) The polyhedron $D(\psi', \Omega)$ has the following universal property: suppose $H \cap Z$ and $f : |\Omega| \to Z$ is a map such that for all $\sigma \in \Omega$, $f(|\Omega|_{\sigma})$ is contained in Fix (G_{σ}, Z) , the fixed point set of G_{σ} on Z. Then there is a unique extension of f to a H-equivariant map $\tilde{f} : D(\psi', \Omega) \to Z$.
- (iv) The orbit projection $p: D(\psi', \Omega) \to |\Omega|$ is a retraction.
- (v) The space $D(\psi', \Omega)$ is connected if and only if $|\Omega|$ is connected and the subgroups $\{G_{\sigma}\}_{\sigma \in \Omega}$ generate *H*.
- (vi) The space $D(\Psi', \Omega)$ is simply connected if and only if $|\Omega|$ is simply connected and H = G.
- (vii) If $D(\psi, Q)$ is contractible, then so is |Q|.

Proof. Since ψ' is injective on local groups, we may identify each local group G_{σ} with its image $\psi'_{\sigma}(G_{\sigma})$.

(i) The poset $D(\psi', \Omega)$ is defined to be the disjoint union $\bigsqcup_{\sigma \in \Omega} H/G_{\sigma}$ consisting of all pairs (gG_{σ}, σ) , with $g \in H$ and $\sigma \in \Omega$. The partial order is the natural one, defined by $(hG_{\tau}, \tau) < (gG_{\sigma}, \sigma) \iff \tau < \sigma$ and $h^{-1}g \in G_{\tau}$.

(ii) For any $x \in |\Omega|$, let $\sigma(x)$ be the least element $\sigma \in \Omega$ such that x belongs to the stratum $|\Omega|_{\sigma}$. The space $D(\psi', |\Omega|)$ is defined by

$$D(\psi', |\Omega|) := (H \times |\Omega|) / \sim, \qquad (5.4)$$

where the equivalence relation \sim is defined by $(g,x) \sim (g',x') \iff x = x'$ and $gG_{\sigma(x)} = g'G_{\sigma(x)}$.

(iii) Suppose [g,x] denotes the equivalence class of (g,x). The map $\tilde{f}: D(\psi', |\Omega|) \to Z$ is then defined by $\tilde{f}([g,x]) = gf(x)$.

(iv) This is precisely what is meant by the statement that |Q| is a strict fundamental domain.

(v) and (vii) Since $p: D(\psi', |\Omega|) \to |\Omega|$ is a retraction, it induces a surjection on π_0 (in fact, on the homotopy groups π_i for all *i*). So, if $D(\psi', |\Omega|)$ is path connected, then so is $|\Omega|$. Let H' be the subgroup of H generated by $\{G_{\sigma}\}$. Since Im ψ' is contained in H', we have a simple morphism $\psi'': G\Omega \to H'$ and an embedding $D(\psi'', |\Omega|) \hookrightarrow D(\psi', |\Omega|)$. Clearly, $D(\psi'', |\Omega|)$ is both open and closed in $D(\psi', |G\Omega|)$. Statement (v) follows. Similarly, (vii) is true.

(vi) The universal property of the direct limits gives a homomorphism $\theta : G \to H$. Put $D = D(\psi, |\Omega|)$ and $D' = D(\psi', |\Omega|, \psi')$, where $\psi : G\Omega \to H$ is the canonical simple morphism. By (iii) there is a θ -equivariant map $\pi : D \to D'$ inducing the identity on $|\Omega|$; the map π is easily seen to be a covering projection. First suppose that D' is simply connected. Then π is a homeomorphism, θ is an isomorphism and by (iv), $|\Omega|$ is simply connected. Conversely, suppose that $|\Omega|$ is simply connected and that G = H. Then we can assume D = D'. Let $E \to D$ be any connected covering space. We will show that D is simply connected by showing that $E \to D$ is a homeomorphism. Since $|\Omega|$ is simply connected the inclusion $|\Omega| \hookrightarrow D$ lifts to E and gives a fundamental domain for the group \tilde{G} of all lifts to E of G-action on D (this is the key point). Each local group G_{σ} lifts isomorphically to the isotropy subgroup of \tilde{G} at a point of $|\Omega|$. Since E is connected, these local groups generate \tilde{G} ; hence, $\tilde{G} \to G$ is onto. By (iii), the lift of the fundamental domain defines a section $D \to E$ of the covering projection $E \to D$. Therefore, $E \to D$ is a homeomorphism.

Remark 5.2. In view of this theorem from now on we shall interpret the statement that "GQ is developable" as being synonymous to the statement that the canonical morphism $\psi : Q \to \lim GQ$ is injective on local groups.

Remark 5.3. When $G = \lim GQ$, we write D(G, |Q|) instead of $D(\psi, |Q|)$. By part (vi) of the above theorem, if |Q| is simply connected, then so is D(|Q|, G). So, when this holds, the direct limit should be thought of as "fundamental group of GQ" and D(|Q|, G) should be thought of as the "universal cover of GQ." This means that in some sense the notion of a simple complex of groups should only be considered when |Q| is simply connected. Indeed, it is only in this case that the *G*-action on the universal cover has |Q| as a strict fundamental domain.

vi is key

The polyhedron $D(|\Omega|, \psi)$ is locally finite if and only if each local group G_{σ} is finite. When this is the case the *G*-action on $D(|\Omega|, \psi)$ is proper.

Example 5.3. (*Graphs of groups*). Suppose Y is a connected graph without loop. (This means that each edge of Y has two distinct endpoints; in other words Y is a regular 1-dimensional CW complex.) its vertex set and edge set are denoted Vert Y and Edge Y. Let $\Omega = \text{Vert } Y \bigsqcup \text{Edge } Y$ be its set of cells, partially ordered by reverse inclusion. A *graph of groups* on Y is a simple complex of groups over Ω . (The standard definition of a graph of groups in [128, 4.4] allows the possibility of loops: one must use directed edges and for each edge τ one must give monomorphisms from G_{τ} to the local groups at its initial and terminal vertices.) If Y is a tree, then $G\Omega$ is a *tree of groups*.

For any system of groups and homomorphisms $\{G_{\sigma}, \phi_{\sigma\tau}\}$ one can form the direct limit (cf. [23, p.]), denoted by $\lim_{\sigma} G_{\sigma}$ In particular, one can take the direct limit of the system formed by $G\Omega$ denoted $\lim_{\sigma} G\Omega$. The direct limit has the universal property that for any group H and simple morphism $\psi : G\Omega \to H$, there is a unique homomorphism $\hat{\psi} : \lim_{\sigma} G\Omega \to H$ such that $\psi_{\sigma} = \hat{\psi}\iota_{\sigma}$ (see [23, II.12.13, p. 376]). A *simple morphism* $\psi = (\psi_{\sigma})$ from $G\Omega$ to a group G is a function which assigns to each $\sigma \in \Omega$ a homomorphism $\psi_{\sigma} : G_{\sigma} \to G$ such that $\psi_{\tau} = \psi_{\sigma}\phi_{\sigma\tau}$ whenever $\tau < \sigma$. For each $\sigma \in \Omega$, there is a canonical homomorphism $\iota_{\sigma} : G_{\sigma} \to \lim_{\sigma} G\Omega$, hence, a *canonical simple morphism* $\iota : G\Omega \to \lim_{\sigma} G\Omega$. The simple complex of groups $G\Omega$ is *developable* if ι is injective on local groups.

A *strict fundamental domain* for the action of a group *G* on a space *Y* is a closed subspace $C \subset Y$ such that *C* intersects each orbit in exactly one point. It follows that restriction of the orbit projection $p: Y \to Y/G$ to *C* is a homeomorphism. Note that *C* provides a section $s: Y/G \to Y$ of the orbit projection *p* defined by $s = (p|_C)^{-1}$. Similarly, if *G* acts on a poset \mathcal{P} by order-preserving automorphisms, then a *strict fundamental domain* for *G* on \mathcal{P} is a subposet Ω which intersects each *G*-orbit in exactly one element. Simple complexes of groups are designed to deal with group actions on cell complexes which admit a strict fundamental domain.

Suppose for example, that $G\Omega$ is a tree of groups, where Ω is the poset of cells in the tree Y. Put $G = \lim G\Omega$. It follows from a result in [128] on the structure of amalgams that that natural maps from the local groups to G are injective, i.e., $G\Omega$ is developable. As in [128], one can construct a G-action on a tree T with fundamental domain Y. The T tree is defined by

Vert
$$T = \bigsqcup_{v \in \text{Vert}Y} G/G_v$$
 and Edge $T = \bigsqcup_{\sigma \in \text{Edge}Y} G/G_\sigma$

The monomorhisms $\phi_{\nu\sigma}: G_{\sigma} \to G_{\nu}$ induce maps $G/G_{\sigma} \to G/G_{\nu}$, hence, the map Edge $T \to \text{Vert } T$ which defines the order relation on the 0- and 1-cells of T. The group G acts on T and one can then check that T is a tree. The isotropy subgroup at any vertex of T is conjugate to one of G_{ν} and an isotropy subgroup at an edge is conjugate to one of the G_{σ} . Moreover, there is a natural inclusion $Y \hookrightarrow T$ under which Y becomes a strict fundamental domain.

Conversely, suppose a group G acts on a tree T with strict fundamental domain a subtree Y. By associating to each vertex or edge of Y the corresponding isotropy subgroup of G one gives Y the structure of a tree of groups.

56

omi

omit

omit these paragraphs

5.3 Examples where Q is the dual face poset of a simplicial complex

5.3 Examples where Ω is the dual face poset of a simplicial complex

Given a simplicial complex *L*, its poset of simplices (including the empty simplex) is denoted by S(L). This section concerns simple complexes of groups over posets of the form S(L). The geometric realization |S(L)| of this poset is the cone on the barycentric subdivision of *L*. (The vertex corresponding to the empty simplex is the cone point.) Given a simple complex of groups GS(L) and a morphism to a group *G*, the fundamental domain for the *G*-action on on its development *D* is |S(L)|. To simplify notation, put

$$K = |\mathcal{S}(L)| \tag{5.5}$$

and call it a *fundamental chamber* (for the *G*-action on *D*). As in (5.3), for each $\sigma \in S(L)$, put $K_{\sigma} = |S(L)|_{\sigma} := |S(L)|_{\geq \sigma}|$.

Example 5.4. If *L* is a PL triangulation of S^{n-1} , then *K* is an *n*-disk and K_{σ} is the dual cell to the simplex σ . In particular, if *L* is the boundary complex of a simplicial convex polytope, then *K* is the dual simple convex polytope. To be even more particular, *L* is the barycentric subdivision of $\partial \Delta$, where Δ is an *n*-simplex, then *K* is an *n*-dimensional permutohedron.

Let *S* be a set. A *Coxeter matrix* on *S* is a symmetric $(S \times S)$ -matrix $M = (m_{st})_{(s,t) \in S \times S}$ with entries in $\mathbb{N} \cup \{\infty\}$, where the diagonal entries all equal 1 and each off-diagonal entry is an integer ≥ 2 or the symbol ∞ . The Coxeter matrix is *right angled* if each of its off-diagonal entries is 2 or the symbol ∞ . There is a *Coxeter group W* associated to the Coxeter matrix *M*; its set of generators is *S* and its relations are: $(st)^{m_{st}} = 1$, where $(s,t) \in S \times S$. Since $m_{ss} = 1$, we have $s^2 = 1$; so, each *s* has order ≤ 2 (in fact, it must = 2). The *cosine matrix* of *M* is the matrix $\cos M = (c_{st})_{(s,t) \in S \times S}$ defined by

$$c_{st} = -\cos(\pi/m_{st}),\tag{5.6}$$

where when $m_{st} = \infty$, c_{st} is interpreted to be $-\cos(\pi/\infty) = -\cos(0) = -1$. There is an extensive literature on Coxeter groups, for example, [15], [48]. The pair (W,S) is a *Coxeter system*. Given a subset $J \leq S$, let $M_J = (m_{st})_{(s,t) \in J \times J}$ denote the restriction of M to J. The subgroup generated by J is denoted W_J and called the *special subgroup* corresponding to J. It is a standard fact that (W_J, J) also is a Coxeter system (cf. [15, pp.12-13]). It turns out that the subgroup W_J generated by J is finite if and only if $\cos M_J$ is positive definite. If W_J is finite, then it can be represented as a group generated by orthogonal linear reflection groups on \mathbb{R}^n , where $n = \operatorname{Card}(J)$. So, W_J is a finite reflection group on the unit sphere $S^{n-1} < \mathbb{R}^n$. For this reason a finite Coxeter group W_J is said to be *spherical*; its set of fundamental generators $J \leq S$ is said to be a *spherical subset* of S. It turns out that a fundamental chamber for W_J on S^{n-1} is a spherical (n-1)-simplex. The poset of nonempty spherical subsets of S is an abstract simplicial complex L(W,S) (or simply L) called the *nerve* of (W,S). Thus, the vertex set of L is S and a nonempty subset of S is a simplex of L if and only if it is spherical. If we include the empty simplex in the face poset of L, we get S(L), the poset of all spherical subsets of S.

Suppose $\sigma(S)$ is the full simplex on *S*. The face poset of the dual simplex $\sigma^*(S)$ is isomorphic to $S(\partial \sigma(S))^{op}$. If (W,S) is a spherical Coxeter system, then $L(W,S) = \sigma(S)$. The poset of subgroups $\{W_J\}_{J < S}$ generated by the proper subsets of *S* is a simplex of groups over $\sigma^* (= \sigma^*(S))$; the development $D(\sigma^*, W)$ is a sphere; and *W* acts properly of $D(\sigma^*, W)$ with strict fundamental domain σ^* . Most of this works for an arbitrary Coxeter system (W, S): we get a simple complex of groups over $S(\partial \sigma(S), i.e., a \text{ simplex of groups over } \sigma^*$, and *W* is the direct limit (provided dim $\sigma(S) \ge 2$). this is the classical definition of the "Coxeter complex". The trouble with this definition is that, in general, the *W*-action on *D* need not be proper, in general. (Also, *D* will not be a sphere.) The remedy is explained in Example 5.5 below: one uses instead the natural simple complex of groups over the poset of spherical subsets of *S*.

Remark 5.4. (*Simplicial Coxeter groups*). A Coxeter system (W, S) is *simplicial* if W_J is finite for each proper subset J < S. If, in addition, W is infinite, then it is called a *Lannér group*. So, if (W, S) is Lannér, then its nerve is $\partial \sigma(S)$ and we get a simple complex of groups over $S(\partial \sigma(S))$. Thus, the W-action on $D(W, \sigma^*)$ is proper exactly in the case when (W, S) is simplicial. When W is Lannér it turns out that $D(W, \sigma^*)$ can be identified with either with euclidean space \mathbb{E}^{n-1} or hyperbolic space \mathbb{H}^{n-1} depending on whether the matrix cos M defined by (5.6) is positive semidefinite or indefinite of signature (n-1,1), cf. [48, §6.9].

There is another way to associate a group to a Coxeter matrix (m_{st}) . For each $s \in S$ introduce the symbol maybe later? a_s . The Artin group A associated to M has generating set $\{a_s\}_{s\in S}$ and relations:

$$\underbrace{a_s a_t \cdots}_{m_{st} \text{ terms}} = \underbrace{a_t a_s \cdots}_{m_{st} \text{ terms}},\tag{5.7}$$

where both sides of the equation are alternating words in a_s and a_t , where $\{s,t\}$ ranges over the edges of the nerve *L*. Notice that the relation (5.7) can be rewritten as $(a_s a_t \cdots)(\cdots (a_s)^{-1}(a_t)^{-1}) = 1$. In the Coxeter group $s = s^{-1}$, so after replacing a_s by s, (5.7) becomes $(st)^{m_{st}} = 1$. Hence, the function $a_s \mapsto s$ extends to an epimorphism $p : A \to W$. The kernel of p is denoted by *PA* and called the *pure Artin group*.

Given a subset $J \leq S$, let A_J denote the subgroup generated by $\{a_s\}_{s \in J}$. It is known that A_J is the Artin group associated to (W_J, J) . Moreover, A is the direct limit of the A_J where J ranges over the simplices in the 1-skeleton of L. The subgroup A_J is a *spherical Artin group* if J is a spherical subset of S.

Example 5.5. (*The complex* WS(L) of spherical subgroups in a Coxeter group). Suppose M is a Coxeter matrix, (W,S) is the associated Coxeter system, L is its nerve and S(L) is the poset of spherical subsets. For each $\sigma \in S(L)$, let M_{σ} be the restriction of M to σ and let (W_{σ}, σ) be the associated Coxeter system.

- (i) For each $\sigma \in S(L)$, there is a natural homomorphism $\psi_{\sigma} : W_{\sigma} \to W$ induced by the inclusion $\sigma \hookrightarrow S$ and similarly, for each $\tau < \sigma$, there is $\phi_{\sigma\tau} : W_{\tau} \to W_{\sigma}$. It is immediate from the definition that *W* is the direct limit of the family $\{W_{\sigma}\}_{\sigma \in S(L)}$.
- (ii) It is proved in [15] that the homomorphisms $\phi_{\sigma\tau}$ and ψ_{σ} are injective. Since each of the $\phi_{\sigma\tau}$ is injective, the functor $\sigma \mapsto W_{\sigma}$ defines a simple complex of groups over S(L). We call this simple complex of groups the *Coxeter complex* and denote it WS(L). Since $\psi_{\sigma} : W_{\sigma} \to W$ is injective, WS(L) is developable.

Let *K* be the fundamental domain defined by (5.5) and let $D = D(K, W) = (W \times K) / \sim$ be the result of applying the basic construction (5.4). Often, *D* is called the *Davis complex*. The focus of the book [48] is the study of *D*.

- (iii) It is shown in [48] and [102] that D(K,W) admits a cell structure so that the resulting piecewise euclidean metric is CAT(0); in particular, D(K,W) is contractible. In this cell structure the cells are the "Coxeter zonotopes" described in **??**. With this cell structure it is more appropriate to call D(K,W) the *Davis-Moussong complex* (Moussong proved in his PhD thesis [102] that this cell structure is CAT(0)). In the right-angled case this is the cubical structure described in Section **??**.
- (iv) Since each of the local groups W_{σ} is finite, the complex D(K, W) is locally finite and the *W*-action on it is proper. Since D(K, W) has a CAT(0)-metric, it is a model for <u>E</u>W (the classifying space for proper *W*-actions).

curvature space RACGs Coxe contractible subcomplexes of the Davis complex SDaC6

eflection gr

Coxeter groups first arose in the study of discrete groups generated by reflections on the constant curvature spaces \mathbb{S}^n , \mathbb{E}^n or \mathbb{H}^n . Let \mathbb{X}^n stand for one of these. The following facts are true (e.g. see [48]).

(a) Any such reflection group is a Coxeter group W.

58

- 5.3 Examples where Q is the dual face poset of a simplicial complex
- (b) The *W*-action on \mathbb{X}^n has a strict fundamental domain *P*, which is geodesically convex in \mathbb{X}^n . (The interior of *P* is a component of the complement of the union of the reflecting hyperplanes in \mathbb{X}^n .)
- (c) Let $D(P,W) = W \times P) / \sim$ be defined as in (5.4). The natural map $D(P,W) \to \mathbb{X}^n$ (defined by Theorem 5.2 (iii)) is a homeomorphism. Thus, \mathbb{X}^n is tessellated by the *W*-translates of *P*.
- (d) A generating set *S* for *W* can be taken to be the set of reflections across the codimension one faces of *P*. Then (W,S) is a Coxeter system and the nerve L(W,S) is dual to the boundary complex, ∂P . The set of reflections of \mathbb{X}^n generated by *S* is precisely the set of conjugates of *S* in *W*. (For this reason, in a general Coxeter system (W,S) a *reflection* in *W* is defined as any conjugate of an element of *S* in *W*.)
- (e) If *P* is compact, then it is a compact polyhedron homeomorphic to an *n*-disk and L(W, S) is the triangulation of S^{n-1} dual to ∂P . Hence, the polyhedron *K* (defined as the dual of Cone *L*) is equal to *P*.
- (f) If *W* is finite, then it has a representation as an orthogonal reflection group on \mathbb{S}^n (where n + 1 = Card(S)). This extends to a linear representation on \mathbb{R}^{n+1} in which each $s \in S$ acts as a linear reflection (cf. [48]). The fundamental chamber is a simplicial cone in \mathbb{R}^{n+1} , bounded by hyperplanes which are fixed point sets of elements of *S*.

These facts indicate the way in which one should think of the Coxeter complex WS(L): *K* should be thought of as a simple convex polytope, *L* as the dual triangulation of S^{n-1} , and the local group W_{σ} is the isotropy group of the cell of *K* which is dual to $\sigma \in S(L)$. The general definitions of a Coxeter system, the fundamental chamber *K* and the basic construction D(K, W) gives a much broader class of examples than occur in constant curvature. Even when *K* is required to combinatorially equivalent to a simple convex polytope very few of these examples are equivalent to tessellations of constant curvature space.

If *L* is a flag triangulation of S^{n-1} , then *K* is an *n*-disk and D(K, W) is a contractible *n*-manifold. If *L* is PL homeomorphic to S^{n-1} , then D(K, W) is PL homeomorphic to euclidean *n*-space. (For further information see [48, Ch. 10]).)

There are many classical beautiful tessellations of constant curvature space.

?K(*L*) is an *m*-gon. Haglund [23, pp. 393–394]

Example 5.6. (*Buildings*). The theory of buildings gives an important way to construct a simple complexes of groups with the same fundamental chamber as a Coxeter system. The *type* of a building is a Coxeter matrix *M*. (Any *M* can occur at least when all m_{st} , with $s \neq t$ lie in $\{2,3,4,6\} \cup \{\infty\}$.) The underlying poset is the same as in Example 5.5: it is S(L) where *L* is the nerve of the Coxeter system (*W*, *S*) of type *M*. Actually, the notion of building can be defined without specifying a local group for each $\sigma \in S(L)$. This goes as follows. A *building of type M* is a pair (\mathcal{C}, δ) where \mathcal{C} is a set (whose elements are called *chambers*) and where $\delta : \mathcal{C} \times \mathcal{C} \to W$ is a function (called a *W-valued distance function*) where δ satisfies certain axioms listed in [1, p.]. Let *J* be a subset of *S*. Two chambers $c, d \in \mathcal{C}$ belong to the same *J-residue* if $\delta(c, d) \in W_J$. (This is an equivalence relation on \mathcal{C} ; a *J*-residue is a nequivalence class.) A residue of type *J* is spherical if *J* is a spherical subset of *S*. For example, an \emptyset -residue is a singleton $\{c\}$. The Coxeter group *W* itself is a building where $\delta : W \times W \to W$ is given by $\delta(v, w) = v^{-1}$. Thus, *v* and *w* belong to the same *J*-residue if and only if they determine the same coset $vW_J = wW_J$. An *apartment* in a building \mathcal{C} is the image of an embedding $W \hookrightarrow \mathcal{C}$ which is isometric with respect to the *W*-valued distance functions. The basic construction on \mathcal{C} is defined using the same fundamental chamber *K* as for the Coxeter complex (cf. (5.5)), that is,

$$D(K,\mathcal{C}) = (\mathcal{C} \times K) / \sim, \tag{5.8}$$

where $(c,x) \sim (d,y)$ if and only if x = y and c, d belong to the same $\sigma(x)$ -residue. Thus, if $p: D(K, \mathcal{C}) \to K$ is the natural projection, then the fiber $p^{-1}(x)$ can be identified with the set of $\sigma(x)$ -residues. It follows that if $\mathcal{P}(\mathcal{C})$ denotes the poset of spherical residues in \mathcal{C} , then its geometric realization $|\mathcal{P}(\mathcal{C})|$ is naturally identified

with $D(K, \mathbb{C})$. The polyhedron $D(K, \mathbb{C})$ defined by (5.8) is also called as the *standard realization of* \mathbb{C} . The building \mathbb{C} has *finite thickness* if each $\{s\}$ -residue is finite for each $s \in S$. This implies that each spherical residue is finite and hence, that the complex $D(K, \mathbb{C})$ is locally finite. One should think of the geometric realization $D(K, \mathbb{C})$ as follows. For each $s \in S$, let K_s denote the codimension one face $|S(L)_{\geq \{s\}}|$. For a given copy of K_s , the set of chambers containing K_s is an $\{s\}$ -residue, so the number of chambers meeting along a codimension one face is ≥ 2 (since in any building each $\{s\}$ -residue must contain at least two elements). On the other hand, each apartment is a Coxeter complex in which exactly two chambers meet along each codimension one face. If the fundamental chamber K happens to be a convex polytope (e.g. a simplex) then $D(K, \mathbb{C})$ is tessellated by copies of K and the link of each codimension one face is a finite set of points while the link a each codimension two face is the realization of rank two spherical building as a graph. To be even more specific, suppose W is a cocompact reflection group on the hyperbolic plane with fundamental domain K a convex polygon with all face angles equal to, say $\pi/3$. Let \mathbb{F} be a field with q elements. Then there is a building \mathbb{C} with realization $D(K, \mathbb{C})$, so that q + 1 copies of K meet along each edge. (Here q + 1 is the number of points in the projective line over \mathbb{F}) and the link at each vertex is the projective plane over \mathbb{F} (a rank two building).

By using the piecewise Euclidean metric on the geometric realizations of its apartments, it is proved in [43] that $D(K, \mathcal{C})$ also has a CAT(0) cell structure; in particular, it is contractible.

A building \mathcal{C} is *spherical* if the corresponding Coxeter system (W, S) is spherical. The *spherical realization* of \mathcal{C} is $D(\mathcal{C}, \sigma^*)$ where σ^* is the simplex dual to the full simplex on *S*. Each apartment in $D(\mathcal{C}, \sigma^*)$ is a round sphere, namely, the spherical realization of $D(\mathcal{C}, \sigma^*)$. So, the spherical realization of any spherical building is a piecewise spherical complex. The standard realization, $D(\mathcal{C}, K)$, of the spherical building \mathcal{C} is isomorphic to the cone on $D(\mathcal{C}, \sigma^*)$. It follows that its spherical realization $D(\mathcal{C}, \sigma^*)$ is CAT(1). (See [43].)

A residue in a building is itself a building. In particular, a spherical residue is a spherical building. This observation provides insight about the local structure of the standard realization of any building \mathcal{C} . Suppose $\mathcal{D} \in \mathcal{P}$ is a spherical residue of type J, where $J \leq S$ is the vertex set of a simplex $\tau \in \mathcal{S}(W,S)$. The geometric realization of any chamber $c \in \mathcal{D}$ is homeomorphic to K and \mathcal{D} corresponds to a stratum K_{τ} of K. The normal link of K_{τ} in K is isomorphic to the dual simplex τ^* . So, the normal link of the stratum corresponding to \mathcal{D} is isomorphic to $D(\mathcal{D}, \tau^*)$, which is the the spherical realization of the spherical building \mathcal{D} . Hence, the link of each stratum in the standard realization of a building is the spherical realization of a spherical building.

terminology

A building \mathcal{C} is *right-angled* (a RAB) if its associated Coxeter system is right-angled. For example, the edge set of any tree without terminal vertices is the set of chambers of a RAB with associated Coxeter group the infinite dihedral group on two generators $S = \{s, t\}$. Since any tree is bipartite the vertices can be assigned type $\{s\}$ or type $\{t\}$. Given a vertex of type $\{s\}$ (or $\{t\}$) the corresponding residue is the set of edges which share that vertex. One need not require any regularity of the tree: two residues of type $\{s\}$ can have different cardinality. An apartment in a tree is an embedded copy of the real line with its usual subdivision into edges. Similarly, a product of trees is a RAB. In general, a RAB \mathcal{C} is *regular* if for each $s \in S$ each residue of type $\{s\}$ has the same cardinality.

Suppose a group *G* acts on \mathcal{C} through isometries of the *W*-valued distance function. Then *G* acts on the poset \mathcal{P} of all spherical residues in \mathcal{C} . If the *G*-action on \mathcal{C} is transitive, then by choosing an element $c \in \mathcal{C}$ as a fundamental chamber, we get a strict fundamental poset for the *G*-action and this poset can be identified with $\mathcal{S}(L)$. This gives the data for a simple complex of groups over $\mathcal{S}(L)$: for each $\sigma \in \mathcal{S}(L)$, G_{σ} is the isotropy subgroup of the residue of type σ which contains *c*. Thus, \mathcal{P} is the disjoint union of all cosets of the form G/G_{σ} , where σ ranges over $\mathcal{S}(L)$. In other words, *every chamber-transitive action of a group on a building* \mathcal{C} of type (W,S) gives rise to a simple complex of groups over *K*. When *G* is chamber transitive, the standard realization of the building, $D(K, \mathcal{C})$, is the same complex as D(K, G), the development of the complex of groups $G\mathcal{S}(L)$. A good example to keep in mind is where \mathcal{C} is a spherical building over a spherical Coxeter

5.3 Examples where Q is the dual face poset of a simplicial complex

system (*W*, *S*). If *G* is chamber transitive on \mathcal{C} we get a simplex of groups $GS(\partial \sigma)$ with fundamental domain σ^* . In other words, any spherical building with a chamber transitive automorphism group gives a simplex of groups (cf. Definition 5.2). More generally, we get a simplex of groups whenever the type of the building is a simplicial Coxeter system. If \mathcal{C} has finite thickness, then $GS(\partial \sigma)$ is a simplex of finite groups and its development $D(\sigma^*, G)$ is a locally finite simplicial complex.

The building \mathcal{C} has *finite thickness* if each rank one residue is finite (i.e., if for each $s \in S$ each $\{s\}$ -residue is finite). It can be shown that this condition is equivalent to the condition that the complex $D(K, \mathcal{C})$ is locally finite. Suppose \mathcal{C} is locally finite. Then its automorphism group $\operatorname{Aut}(\mathcal{C})$ is a locally compact topological group. The group $\operatorname{Aut}(\mathcal{C})$ is usually not discrete. (For example, in the the full automorphism group of a regular tree, the stabilizer of a chamber is an inverse limit of finite groups of increasing size.) In addition, suppose that the $\operatorname{Aut}(\mathcal{C})$ -action on the building is chamber transitive. A *lattice* in $\operatorname{Aut}(\mathcal{C})$ is a discrete subgroup *G* of finite covolume. The lattice is *uniform* if it is cocompact (i.e., if $D(K, \mathcal{C})/G$ is compact).

If \mathcal{C} is locally finite, then any discrete subgroup $G \leq \operatorname{Aut}(\mathcal{C})$ acts properly on $D(K, \mathcal{C})$. Since $D(K, \mathcal{C})$ is CAT(0), it is a model for <u>E</u>G. Hence, if G uniform lattice in Aut(\mathcal{C}), we get a cocompact model for <u>E</u>G.

Example 5.7. (*The graph product complex*). We recall the construction of RABs in 3.3. Let Δ be the simplex on a set *S*. The RACG with nerve Δ is $(\mathbb{C}_2)^S$. A building of type $(W_s, \{s\})$ is just a discrete set E_s of cardinality at least two. Its standard realization is Cone E_s . The product building $\prod_{s \in S} E_s$ has type $((\mathbb{C}_2)^S, S)$ with standard realization $\prod_{s \in S} \text{Cone } E_s$. The fundamental chamber for the product building is $K(\Delta)$ (as in ??, $K(\Delta)$ can be identified the cube $[0, 1]^S$). Next suppose that *L* is a flag complex with the same vertex set *S* and with 1-skeleton L^1 . Define a right-angled Coxeter matrix (m_{st}) by

$$m_{st} = \begin{cases} 1, & \text{if } s = t, \\ 2, & \text{if } \{s, t\} \in \text{Edge } L^1, \\ \infty, & \text{if } \{s, t\} \notin \text{Edge } L^1, \end{cases}$$
(5.9)

Let (W_L, S) be the associated RACS with nerve L. (In other words, W_L is the graph product of copies of C_2 over L^1 .) Since L is a subcomplex of Δ , K(L) is a subcomplex of $K(\Delta)$. Let $p : \prod_{s \in S} \text{Cone } E_s \to K(\Delta)$ be the natural projection. As in Definition 3.4, put $Z_L = p^{-1}(K(L))$. (The space Z_L can be identified with the polyhedral product $\text{Cone } \mathbf{E}^L$.) When $L \neq \Delta$, the space Z_L will not be simply connected. It is shown in [49] that the universal cover \tilde{Z}_L is the standard realization of a RAB \mathcal{C} of type (W_L, S) . If $c \in K(L)$ denotes the cone point corresponding to $\emptyset \in S(L)$, then \mathcal{C} can be identified with the inverse image of c in \tilde{Z}_L . Since any residue of type $\{s\}$ in \mathcal{C} is identified with E_s , the building \mathcal{C} is regular. Moreover, any regular RAB comes from this construction. The building \mathcal{C} is locally finite if and only if each E_s is a finite set. For example, if Card(S) = 2 and $L^1 = S^0$, then W_L is the infinite dihedral group and the realization of \mathcal{C} is a biregular tree.

Next suppose that each E_s admits a transitive action of a group G_s . So, E_s can be identified with G_s/B_s , where B_s is the isotropy subgroup at a basepoint. The G_s -action is effective if and only if the subgroup B_s is malnormal. (This means that $gB_sg^{-1} \cap B_s = \{1\}$ if $g \notin B_s$.) We assume each B_s is malnormal. Then $\prod G_s \curvearrowright \prod \text{Cone } E_s$ and the action is chamber transitive. The subcomplex Z_L is stable under the $\prod G_s$ -action. The group G of all lifts of this action to the universal cover \tilde{Z}_L is called the *generalized graph product* of the family $\{(G_s, B_s)\}$ with respect to L^1 . If each B_s is trivial, then G is the ordinary graph product as defined in subsection 3.1.3.

The simple complex of groups GS(L) associated to the *G*-action on this RAB is called the *graph product complex*. Thus, the local group G_{σ} associated to the simplex $\sigma \in S(L)$ is the direct sum, $G_{\sigma} = \prod_{s \in \sigma} G_s$. If $\tau < \sigma$, then $\phi_{\sigma\tau} : G_{\tau} \to G_{\sigma}$ is the natural inclusion.

Should it be Aut C?

Maybe different spot

Example 5.8. (*Constructing buildings from spherical buildings*). We recall the construction of RABs in 3.3. Let Δ be the simplex on a set S. The RACG with nerve Δ is $(\mathbf{C}_2)^S$. For each $s \in S$, let E_s be a discrete set with $\operatorname{Card}(E_s) \geq 2$. Then E_s is a building of type $(\mathbf{C}_2, \{s\})$; its standard realization is $\operatorname{Cone} E_s$. Similarly, the product $\prod_{s \in S} \operatorname{Cone} E_s$ is a building of type $((\mathbf{C}_2)^S, S)$; its realization is $\prod_{s \in S} \operatorname{Cone} E_s$. The fundamental chamber for $\prod_{s \in S} \operatorname{Cone} E_s$ is $K(\Delta)$ (as in ??, $K(\Delta)$ can be identified the cube $[0, 1]^S$). Next suppose that L is a flag complex with the same vertex set S. Define a right-angled Coxeter matrix (m_{st}) by

$$m_{st} = \begin{cases} 1, & \text{if } s = t, \\ 2, & \text{if } \{s, t\} \in \text{Edge } L, \\ \infty, & \text{if } \{s, t\} \notin \text{Edge } L, \end{cases}$$
(5.10)

and let (W, S) be the associated RACS with nerve *L*. Since *L* is a subcomplex of Δ , K(L) is a subcomplex of $K(\Delta)$. Let $p : \prod_{s \in S} \text{Cone } E_s \to K(\Delta)$ be the natural projection. As in Definition 3.4, put $Z_L = p^{-1}(K(L))$. (The space Z_L can be identified with the polyhedral product $\text{Cone } \mathbf{E}^L$.) When $L \neq \Delta$, the space Z_L will not be simply connected. It is shown in [49] that the universal cover \tilde{Z}_L is the standard realization of a RAB. If $c \in K(L)$ denotes the cone point corresponding to $\emptyset \in \mathcal{S}(L)$, then the set of chambers \mathcal{C} in this RAB can be identified with the inverse image of *c* in \tilde{Z}_L . Any regular right-angled building comes from the above construction. More precisely, if \mathcal{C} is a regular RAB with associated RACS (W, S) each residue of type $\{s\}$ has cardinality $q_s + 1$, then taking E_s to be a set of cardinality $q_s + 1$ and applying the above construction we recover \mathcal{C} .

K(L) is an *m*-gon

In [49] it is shown that the process of starting from a product building and constructing a RAB can be generalized to a process for constructing new buildings starting from an arbitrary spherical building \mathcal{C}' of type (W', S). As before, the nerve of (W', S) is the simplex Δ on S. The standard realization of this building is $D(K(\Delta), \mathcal{C}')$ Let (m'_{st}) be the Coxeter matrix of (W', S). Next choose a collection of edges in Δ and for each edge, $\{s,t\}$, in this collection change m'_{st} to the symbol ∞ . This produces a new Coxeter matrix (m_{st}) with associated Coxeter system (W, S) with nerve L. As before, $K(L) \leq K(\Delta)$. By taking the inverse image of K(L) in $D(K(\Delta), \mathcal{C}')$, we get $D(K(L), \mathcal{C}') \leq D(K(\Delta), \mathcal{C}')$. Finally, let \tilde{D} denote the universal cover of $D(K(L), \mathcal{C}')$. This is the standard realization of a building whose set of chambers \mathcal{C} can be identified with the inverse image of the cone point c in the universal cover.

Variations of this constructions are given in [49].

Questions: Tits [135] covering paper: Simply connected local building \implies building?

Example 5.9. (*The complex* AS(L) of spherical Artin subgroups). Given a Coxeter matrix M there is another way to get a complex of groups over S(L). Let A be the associated Artin group with generating set $X = \{a_s\}_{s \in S}$ and relations given by (5.7). For each $\sigma \in S(L)$, let A_{σ} be the spherical Artin group corresponding to M_{σ} . As for Coxeter groups as described in Example 5.5 we have the following.

- (i) For each σ ∈ S(L), there is a natural homomorphism ψ_σ : A_σ → W induced by the inclusion σ → S and similarly, for each τ < σ, there is φ_{στ} : A_τ → A_σ. It is immediate from the definition that A is the direct limit of the family {A_σ}_{σ∈S(L)}.
- (ii) The homomorphisms $\phi_{\sigma\tau}$ and ψ_{σ} are injective (cf. [32], [96]). Since each of the $\phi_{\sigma\tau}$ is injective, the functor $\sigma \mapsto A_{\sigma}$ defines a simple complex of groups over S(L). This simple complex of groups is called the *Artin complex* and denoted by AS(L). Since $\psi_{\sigma} : A_{\sigma} \to A$ is injective, AS(L) is developable.

Let *K* be defined by (5.5) and let $D = D(K,A) = (A \times K) / \sim$ be the result of applying the basic construction (5.4).

5.3 Examples where Q is the dual face poset of a simplicial complex

63

It is probably never a building

- (iii) The complex D(K,A) is not locally finite since each nontrivial local group is infinite. For example, the subgroup $A_{\{s\}}$ generated by the Artin generator a_s is infinite cyclic.
- (iv) If *M* is right-angled, then *A* is a RAAG and D(K,A) is a RAB of the type considered in Example ??. If *A* is not a RAAG, then D(K,A) is not a building.

However, the complex D(K,A) is "building-like" in that it has many properties similar to those of a building with a chamber transitive group action. First, the fundamental chamber for A on D(K,A) isomorphic to that of the associated Coxeter system (W,S). Second, D(K,A) has a large collection of subcomplexes each isomorphic to the Davis complex D(K,W). These subcomplexes are natural candidates for apartments. (To see these subcomplexes use the fact that, as in [64], there is a set theoretic section of the canonical epimorphism $p: A \to W$.) Third, by Theorem 5.2 (vi), D(K,A) is simply connected. It is not known whether or not D(K,A)is contractible. This is the main unanswered question about Artin groups. (As explained below in section ??, it is equivalent to answering the $K(\pi, 1)$ -Question for Artin groups.)

Example 5.10. (Spherical Artin groups and complements of hyperplane arrangements). An Artin group can be understood in terms of the fundamental group of a complement of an arrangement of complex hyperplanes. Suppose *A* is a spherical Artin group associated to a spherical Coxeter system (W, S). As was pointed out in Example 5.5 (f), there is a canonical representation of *W* as a group generated by linear reflections on \mathbb{R}^n . Complexifying we get a *W*-action on \mathbb{C}^n . Let \mathcal{A} denote the collection of linear hyperplanes which are fixed by reflections in *W*. Let

$$M(\mathcal{A}) = \mathbb{C}^n - \bigcup_{H \in \mathcal{A}} H,$$
(5.11)

be the complement of the union of the reflecting hyperplanes. It is straightforward to see that $\pi_1(M(\mathcal{A}) \text{ can})$ be identified with the pure Artin group *PA*. Since *W* acts freely on $M(\mathcal{A}), M(\mathcal{A}) \to M(\mathcal{A})/W$ is a covering space and so,

$$A = \pi_1(M(\mathcal{A})/W). \tag{5.12}$$

The prototypical example of a spherical Artin group is Artin's braid group, [6]. The braid group B_{n+1} on n+1 strands has several different incarnations. For example, it is the fundamental group of the configuration space of n+1 points in the plane (cf. subsection 2.4.4). It can also be described as the fundamental group of a hyperplane arrangement. The associated Coxeter group is S_{n+1} the symmetric group on n+1 letters. It acts as a reflection group on \mathbb{R}^{n+1} by permuting the coordinates, as well as on the subspace ($\cong \mathbb{R}^n$) orthogonal to the diagonal. Complexifying we get a S_{n+1} -action on \mathbb{C}^n with reflecting hyperplanes defined by the equations $x_i = x_j$ giving the arrangement \mathcal{A} . Then $B_{n+1} = \pi_1(\mathcal{M}(\mathcal{A})/S_{n+1})$.

The most important result about spherical Artin groups is the following.

Theorem 5.3. (Deligne's Theorem, [64]). Suppose A is the arrangement of reflecting hyperplanes for a spherical Artin group A. Then M(A) is aspherical. Hence, M(A)/W is a model for the classifying space BA.

Remark 5.5. General Artin groups also have an interpretation as the fundamental groups as complements of complex hyperplane arrangement, cf. [96], [32]. Tits showed that for any Coxeter system (W, S) there is a linear representation on \mathbb{R}^n so that the elements of *S* act as reflections across the facets of a simplicial cone (the fundamental chamber), that the union of all *W*-translates of this chamber is a convex cone, and that the *W*-action is proper on the interior \mathfrak{I} of this cone. (See [15] or [48].) Moreover, *W* acts properly on $\mathbb{R}^n + \sqrt{-1}\mathfrak{I}$. The reflecting hyperplanes give an arrangement \mathcal{A} of codimension two subspaces in $\mathbb{R}^n + \sqrt{-1}\mathfrak{I}$ and the quotient by *W* of the complement of their union has fundamental group \mathcal{A} . It is conjectured that Deligne's Theorem holds in this more general situation, i.e., that $M(\mathcal{A})/W$ is a model for for *BA*.

Given a Coxeter system (W, S) there is an associated *Artin group A*. This group has one generator x_s for each $s \in S$; its relations are the *braid relations*: and where m_{st} denotes the order of st in W. The matrix (m_{st}) is called the *Coxeter matrix*. For any $T \leq S$, let A_T denote the Artin group corresponding to the Coxeter system (W_T, T) . As with Coxeter groups, A_T can be identified with the special subgroup of A generated by $\{x_t\}_{t\in T}$. (When T is a spherical subset of S, this is proved in [64, Thèoréme 4.13 (iii)] and in general in [96, Theorem 4.14].) The Artin group A_T is *spherical* if T is a spherical subgroups of A give a simple complex of groups AS(L) called the *Artin complex*. When σ is a simplex of L and $T = \text{Vert }\sigma$, we shall often write A_{σ} instead of A_T . It is clear that the direct limit $\lim AS(L)$ is A. Since A_{σ} is isomorphic to a subgroup of A, AS(L) is developable. Note that if L' is any subcomplex of the full simplex on S with the same 1-skeleton as L, then $\lim AS(L') = A$.

By Deligne's Theorem in [64], any spherical Artin group A_{σ} has a classifying space BA_{σ} which is a finite CW complex of dimension one greater than dim σ . (There is a specific model for BA_{σ} called the *Salvetti* complex, see [33] or Subsection 2.2.2). The " $K(\pi, 1)$ -Conjecture" for Artin groups is the conjecture that the $K(\pi, 1)$ -Question has a positive answer for any Artin poset AS(L). By Proposition 5.1 this is equivalent to the conjecture that D(|AS(L)|, A) is contractible. It is the most important unsolved problem concerning general Artin groups. A detailed discussion can be found in [32], where the conjecture is proved whenever L is a flag complex (we generalize this in Theorem ?? below).

5.4 Polygons of groups

5.5 Aspherical realizations and the $K(\pi, 1)$ -Question

The classifying space of a discrete group *H* is denoted *BH* and its universal cover by *EH*. A simple complex of groups $G\Omega$ gives the data for a poset of spaces $\{BG_{\sigma}, \overline{\phi}_{\sigma\tau}\}$, where $\tau < \sigma \in \Omega$ and where $\overline{\phi}_{\sigma\tau} : BG_{\tau} \to BG_{\sigma}$ is the map induced by the monomorphism $\phi_{\sigma\tau} : G_{\tau} \to G_{\sigma}$. Using this data we can glue together the disjoint union of spaces $||\Omega|^{\sigma} \times BG_{\sigma}$ by using iterated mapping cylinders. For each $\tau < \sigma$, $|\Omega|^{\sigma}$ is a subcomplex of $|\Omega|^{\tau}$ and one glues the subspace $|\Omega|^{\sigma} \times BG_{\tau}$ of $|\Omega|^{\tau} \times BG_{\tau}$ to $|\Omega|^{\sigma} \times BG_{\sigma}$ via the map:

$$\mathbb{I} \times \overline{\phi}_{\sigma\tau} : |\mathfrak{Q}|^{\sigma} \times BG_{\tau} \to |\mathfrak{Q}|^{\sigma} \times BG_{\sigma},$$

where I denotes the identity map on $|\Omega|^{\sigma}$. The resulting space $BG\Omega$ is called the *aspherical realization* of $G\Omega$. It is well-defined up to homotopy equivalence. If $|\Omega|$ is connected and simply connected, then it follows from van Kampen's Theorem that $\pi_1(BG\Omega) = \lim G\Omega$. We note that

$$\dim BGQ = \sup\{(\dim BG_{\sigma} + \dim |Q|^{\sigma}) \mid \sigma \in Q\}$$
(5.13)

Proposition 5.1. Suppose GQ is a simple complex of groups over Q with |Q| simply connected. Let $G = \lim GQ$ and let $D = D(|Q|, \iota)$ be the basic construction. When GQ is developable, BGQ is homotopy equivalent to the Borel construction $EG \times_G D$.

Proof. Projection on the second factor induces a projection $p : EG \times_G D \to D/G = |\Omega|$ so that the inverse image of the vertex σ is homotopy equivalent to BG_{σ} . This uses the fact that $G\Omega \to G$ is injective on local groups, otherwise, $p^{-1}(\sigma)$ is homotopy equivalent to $B\overline{G}_{\sigma}$, where \overline{G}_{σ} means the image of G_{σ} in G. It follows that $EG \times_G D$ is an aspherical realization of $G\Omega$.

5.6 The semidirect product construction

Corollary 5.1. Suppose GQ is a developable simple complex of groups with |Q| simply connected. Then the $K(\pi, 1)$ -Question for GQ has a positive answer if and only if D is contractible.

Proof. By Proposition 5.1, $BGQ \sim EG \times_G D$. Hence, BGQ is aspherical if and only if the universal cover of $EG \times_G D$ is contractible. This universal cover is $EG \times D$, which yields the corollary.

Remark 5.6. Since $|\Omega|$ is a retract of *D*, a necessary condition for *D* to be contractible is that $|\Omega|$ is contractible.

In the construction of BGQ we can assume that for each σ , dim $BG_{\sigma} = \operatorname{gd} G_{\sigma}$. Next, we seek a condition which implies that the maximum value of the quantity dim $BG_{\sigma} + \dim |Q|^{\sigma}$ in (5.13) occurs when σ is a maximal element of Q, i.e., when $|Q|^{\sigma}$ is a point. The condition is the following:

$$\operatorname{gd} G_{\sigma} > \operatorname{gd} G_{\tau}, \quad \text{whenever } \sigma > \tau.$$
 (5.14)

Since a k-simplex in $|\Omega|^{\tau}$ corresponds to a chain $\tau = \tau_0 < \cdots < \tau_{\kappa}$, we see that (5.14) implies $\operatorname{gd} G_{\tau_k} \geq \operatorname{gd} G_{\tau} + k \geq \operatorname{gd} G_{\tau} + \dim |\Omega|^{\tau}$; so the maximum value occurs when σ is maximal.

Proposition 5.2. Suppose a simple complex of groups GQ is developable and that the $K(\pi, 1)$ -Question for GQ has a positive answer. Then

$$\operatorname{gd} G \leq \operatorname{dim} BG \mathfrak{Q} = \sup_{\sigma \in \mathfrak{Q}} \{ \operatorname{gd} G_{\sigma} + \operatorname{dim} |\mathfrak{Q}|^{\sigma} \}.$$

If (5.14) holds, then $\operatorname{gd} G = \sup_{\sigma \in \Omega} \{ \operatorname{gd} G_{\sigma} \}.$

Proof. Since $\operatorname{gd} G \leq \operatorname{dim} BGQ$, the first formula follows from (5.13). So, if condition (5.14) holds, $\operatorname{gd} G \leq \sup \{ \operatorname{gd} G_{\sigma} \mid \sigma \in Q \}$. Since EG/G_{σ} is a model for BG_{σ} , $\operatorname{gd} G \geq \operatorname{gd} G_{\sigma}$, so the previous inequality must be an equality.

5.6 The semidirect product construction

A simple complex of groups $G\Omega$ is a functor from a poset Ω to the category of groups. To say that Γ acts on $G\Omega$ means that Γ is a group of natural automorphisms of this functor. In other words,

(a) $\Gamma \curvearrowright \Omega$ through order-preserving bijections (for $(\gamma, \sigma) \in \Gamma \times \Omega$, the action is denoted $(\gamma, \sigma) \mapsto \gamma \sigma$),

(b) for each $\sigma \in \Omega$, there is an isomorphism $f_{\sigma} : G_{\sigma} \to G_{\gamma\sigma}$, so that

(c) whenever $\tau, \sigma \in \Omega$ with $\tau < \sigma$, the following diagram commutes

Let *G* be the direct limit, $G = \lim GQ$. Recall that the development of GQ is defined to be the set of pairs (hG_{σ}, σ) in $G/G_{\sigma} \times Q$ (see the proof of (i) in Theorem 5.2). Suppose $\Gamma \curvearrowright GQ$. Then the semidirect product $G \rtimes \Gamma$ acts on D(Q, G) by

$$(g,\gamma) \cdot (hG_{\gamma\sigma},\sigma) = (g\gamma(h)G_{\sigma},\gamma\sigma).$$
 (5.15)

This a corresponding action of $G \rtimes \Gamma$ on the space D(|Q|, G) with quotient space |Q|. When the Γ -action on |Q| does not have a strict fundamental domain, the $(G \rtimes \Gamma)$ -action on D(|Q|, G) will not have a strict fundamental domain; so, the action will not give rise to a simple complex of groups. However, it still gives rise to a nonsimple complex of groups, which we denote by $GQ \rtimes \Gamma$. If |Q| is simply connected, then the *fundamental group* of $GQ \rtimes \Gamma$ is $G \rtimes \Gamma$.

Example 5.11. (*HNN extensions*). Suppose *A* is a subgroup of a group *G* and $\theta : A \to H$ is a monomorphism. (This is the data for a loop of groups: *H* is the vertex group and *A* is the edge group). In [128, Prop. 5, p. 8] Serre defines the HNN construction $*_{\theta}H$ as a semidirect product of \mathbb{Z} with a tree of groups (cf. Definition 5.4). The tree is the real line subdivided into edges of the form [n, n+1] so that the vertex set of the tree is \mathbb{Z} . The poset Ω is the dual of the face poset of the tree. The group H_n associated to a vertex *n* is a copy of *H*. Each edge group is a copy of *A*. For the edge [n, n+1] there are monomorphisms $A \to H_n$ and $A \to H_{n+1}$ corresponding to the left and right endpoints. The first is natural inclusion $i : A \to H$ and the second is θ . This defines a simple complex of groups $G\Omega$. Let $G = \lim G\Omega$, in other words, *G* is an amalgam of infinitely many copies of *H*. The \mathbb{Z} -action on Ω induced by translation on \mathbb{R} gives a \mathbb{Z} -action on $G\Omega$ and represents \mathbb{Z} as a group of automorphisms of *G*. The *HNN extension* is then defined by $*_{\theta}H := G \rtimes \mathbb{Z}$.

Example 5.12. (Semidirect products with Coxeter groups and their relatives). A diagram automorphism of a Coxeter system (W,S) is a permutation γ of S which preserves the associated Coxeter matrix (i.e., $m_{\gamma S,\gamma t} = m_{s,t}$). Such a permutation defines an automorphism of the Coxeter group W as well as an automorphim of the associated Artin group A. Similarly, γ induces an automorphism on any building of type (W,S). (The diagram automorphism can be regarded as a simplicial automorphism of the nerve L(W,S) which preserves the labels m_{st} on the edges.) Suppose Γ is a group of diagram automorphisms (W,S). We can then form the semidirect products $W \rtimes \Gamma$ and $A \rtimes \Gamma$ with corresponding complexes of groups $WS(L) \rtimes \Gamma$ and $AS(L) \rtimes \Gamma$ (cf. Examples 5.5 and 5.9). Similarly, if G is a chamber transitive automorphism group of a building, we can form a corresponding semidirect product $G \rtimes \Gamma$.

If *S* is a finite set, then *L* is a finite simplicial complex and the group of diagram automorphisms Γ is necessarily a finite groups. When *L* is an (n-1)-sphere, the finite group Γ acts on the contractible *n*-manifold D(K,W) (= $\underline{E}(W \rtimes \Gamma)$). By choosing Γ so that the fixed set of Γ on *L* is not equivalent to a linear subsphere one can produce examples where the fixed set of Γ on D(K,W) is either (a) knotted at infinity as in [70] or not a manifold as in [61], see [48, Thm. 11.7.3]

The universal cover of a development. Our principal use for the semidirect product construction is to understand the universal cover of a development of a simple complex of groups over a poset Q' where |Q'|need not be simply connected. Let *P* be the universal cover of |Q'| and let $p: P \to |Q'|$. The elements of Q'are the vertices of the geometric realization |Q'| of its order complex. The simplicial structure on |Q'| lifts to a simplicial structure on *P*. This defines a poset structure Q on the vertex set of *P* so that $p: Q \to Q'$ is order-preserving and |Q'| = P. Explicitly, if $\tau, \sigma \in Q$, then $\tau < \sigma$ if and only if $p(\tau) < p(\sigma)$ and the edge from $p(\tau)$ to $p(\sigma)$ lifts to an edge from τ to σ . Next suppose G'Q' be a developable simple complex of groups over Q'. Define a simple complex of groups GQ over Q by: $G_{\sigma} := G_{p(\sigma)}$. Put $\pi = \pi_1(|Q|)$. Then π acts on |Q| as deck transformations. This extends to $\pi \cap GQ$, where for any $\gamma \in \pi$ each of the corresponding isomorphisms $f_{\gamma}: G_{\sigma} \to G_{\gamma\sigma}$ is the identity. By the universal property of direct limits there is obvious epimorphism $\psi: G \to G'$. Put $N = \ker \psi$. The natural map $D(|Q|, G) \to D(|Q'|, G')$ is a covering projection. Since |Q| is simply connected, it follows from Theorem 5.2 (vi), that D(|Q|, G) is simply connected. Hence, D(|Q|, G)is the universal cover. As in the first paragraph of this section, $G \rtimes \pi$ acts on D(|Q|, G) and $N \rtimes \pi$ is the group of deck transformations of $D(|Q|, G) \to D(|Q'|, G')$. So, we have proved the following generalization of Theorem 5.2 (vii). **Proposition 5.3.** With notation as above, the fundamental group of D(|Q'|, G') is $N \rtimes \pi$ and its universal cover is D(|Q|, G). So, if D(|Q'|, G') is aspherical, then so is |Q'|.

Proof. The last sentence follows from Theorem 5.2 (iv) and the fact that a retraction of an aspherical space is aspherical.

Suppose (W, S) is a Coxeter system (cf. [15] or [48]). This means that W is a group, that S is a distinguished set of generators and that W has a presentation of the form

$$W := \langle s_i \in S \mid s_i^2 = (s_i s_j)^{m_{ij}} = 1 \rangle$$

For any subset *T* of *S*, the subgroup generated by *T* is denoted W_T and called the *special subgroup* corresponding to *T*. It is a standard fact that (W_T, T) also is a Coxeter system (cf. [15, pp.12-13]). The subset *T* is *spherical* if W_T is finite, in this case W_T is a *spherical special subgroup*. Let S(W, S) denote the poset of spherical subsets of *S*. There is a simplicial complex L (= L(W, S)), called the *nerve* of (W, S). Its vertex set is *S* and a subset $T \leq S$ spans a simplex of *L* if and only if *T* is spherical. Thus, S(L) = S(W, S), the poset of spherical subsets. This gives a simple complex of groups over S(L), denoted WS(L), and called the *complex of spherical Coxeter groups* (or simply the *Coxeter complex*). The local group W_{σ} at a simplex $\sigma \in S(L)$ is the spherical subgroup generated by the vertices of σ . The direct limit of WS(L) is the Coxeter group *W*.

The discussion in Example 5.9 yields the following proposition.

Proposition 5.4. Let A be an Artin group such that the nerve L of its associated Coxeter system is ddimensional. If the $K(\pi, 1)$ -Question for the associated Artin poset has a positive answer, then gdimA = d + 1.

Proof. Since the aspherical realization BAS(L) is formed by gluing together the $BA_{\sigma} \times |S(L)|^{\sigma}$ with $\sigma \in S(L)$ and since dim $BA_{\sigma} = \dim \sigma + 1$, we have dim BAS(L) = d + 1. Since BAS(L) is a model for BA, $gdimA \le d + 1$. On the other hand, for any *d*-simplex $\sigma \in S(L)$, the spherical Artin group A_{σ} contains a free abelian subgroup of rank d + 1 (e.g., see [?dh16]). So, $cdA_L \ge d + 1$. Hence, $d + 1 \ge gdimA_L \ge cdA_L \ge d + 1$; so all inequalities are equalities.

Definition 5.5. A simplicial complex *L* is a *flag complex* if it satisfies the following: if *T* is any finite set of vertices of *L* which are pairwise connected by edges, then *T* spans a simplex of *L*. A simplicial graph L^1 determines a flag complex *L*: the simplices of *L* are the cliques in L^1 . (This is also called the "clique complex" of L^1 .)

Definition 5.6. Suppose L^1 is a simplicial graph with vertex set *V* and edge set *E*. Let $\{G_v\}_{v \in V}$ be a collection of groups indexed by *V*. The *graph product* of the G_v , denoted $\prod_{L^1} G_v$, is the quotient of the free product of the G_v , $v \in V$, by the normal subgroup generated by all commutators of the form, $[g_v, g_w]$, where $\{v, w\} \in E$, $g_v \in G_v$ and $g_w \in G_w$.

Example 5.13. (*The graph product complex*). Suppose $\prod_{L^1} G_v$ is a graph product and that *L* is the flag complex determined by L^1 . There is a simple complex of groups GS(L) over S(L) called the *graph product complex*. It is defined by putting G_{σ} equal to the direct product,

$$\prod_{v\in\operatorname{Vert}\sigma}G_v,$$

for each simplex $\sigma \in L$ and letting $\phi_{\sigma\tau} : G_{\tau} \to G_{\sigma}$ be the natural inclusion whenever $\tau < \sigma$. It is immediate that $\prod_{L^1} G_v = \lim GS(L)$.

Remark 5.7. Another approach to the graph product complex is given in [43] or [?davis12] using the notion of a "polyhedral product". Suppose *L* is a simplicial complex with vertex set *V* and that we are given a collection of pairs of spaces $(uy, uy) = \{(X_v, Y_v)\}_{v \in V}$ together with a choice of basepoint $*_v \in Y_v$. Let $\prod_{v \in V} X_v$ denote the subspace of the Cartesian product consisting of all *V*-tuples $(x_v)_{v \in V}$ such that $x_v = *_v$ for all but finitely many *v*. Given a *V*-tuple, $x = (x_v)$, put $\sigma(x) = \{v \in V \mid x_v \in X_v - Y_v\}$. The *polyhedral product* $(ux, uy)^L$ is the subspace of $\prod_{v \in V} X_v$ consisting of all *x* such that $\sigma(x)$ is a simplex in *L*. (Usually, we only shall be concerned with the case where each Y_v equals the basepoint $*_v$, in which case the notation will be simplified to ux^L .) For any $\tau \in S(L)$, the set of all *x* with $\sigma(x) \ge \tau$, can be identified with the product, $X^\tau := \prod_{v \in V \in T} X_v$. Thus, ux^L is the union of the X^τ , with $\tau \in S(L)$.

Next let L^1 be any simplicial graph and L' any simplicial complex with 1-skeleton = L^1 . Let G be the graph product $\prod_{L^1} G_v$ and $H = \prod_{v \in V} G_v$ be the direct product. Let $\iota : GS(L^1) \to G$ and $\psi : GS(L^1) \to G$ be the natural simple morphisms. Consider the polyhedral product:

$$\mathcal{Z}(L') = (\operatorname{Cone} G_v, G_v)^{L'}$$

Then $\mathcal{Z}(L')$ is locally isomorphic to a product of discrete sets. Moreover, $\mathcal{Z}(L')$ can be identified with the development $D(|\mathcal{S}(L')|, \psi)$. The fundamental group of $\mathcal{Z}(L')$ can be identified with the kernel of the natural epimorphism $G \to H$. So, the *H*-action on $\mathcal{Z}(L')$ lifts to a *G*-action on the universal cover $\tilde{\mathcal{Z}}(L')$ with the same strict fundamental domain. It follows from Remark **??** that $\tilde{\mathcal{Z}}(L') = D(|\mathcal{S}(L')|, t)$.

Proposition 5.5. (cf. [63, Theorem 2.22] and [?dk]). Suppose $G = \prod_{L^1} G_v$ is the graph product of nontrivial groups over a simplicial graph L^1 , and let L be the determined flag complex. Then the $K(\pi, 1)$ -Question for GS(L) has a positive answer.

Since $BG_1 \times BG_2$ is a model for $B(G_1 \times G_2)$, it is obvious that

$$gdim(G_1 \times G_2) \leq gdimG_1 + gdimG_2.$$

On the other hand, by using certain torsion-free subgroups of RACGs, Dranishnikov [?dran] showed that there are groups G_1 and G_2 for which the inequality is strict (cf. [48, Example 8.5.9]). Hence, for $G_{\sigma} = \prod_{\nu \in \sigma} G_{\nu}$, we have that $gdimG_{\sigma} \leq \sum_{\nu \in \sigma} gdimG_{\nu}$ and the inequality can be strict.

A corollary to Proposition 5.5 is the following calculation of the geometric dimension of any graph product of groups.

Corollary 5.2. Suppose $G = \prod_{L^1} G_v$ is the graph product of nontrivial groups over a simplicial graph L^1 . Let L be the flag complex determined by L^1 . Then $gdimG = \sup\{gdimG_{\sigma} \mid \sigma \in S(L)\}$.
References

- [1] P. Abramenko and K. S. Brown, *Buildings*, Graduate Texts in Mathematics, vol. 248, Springer, New York, 2008. Theory and applications. MR2439729
- [2] A. Abrams, *Configuration spaces of colored graphs*, Geom. Dedicata 92 (2002), 185–194, DOI 10.1023/A:1019662529807. Dedicated to John Stallings on the occasion of his 65th birthday. MR1934018
- [3] I. Agol, *The virtual Haken conjecture*, Doc. Math. 18 (2013), 1045–1087. With an appendix by Agol, Daniel Groves, and Jason Manning. MR3104553
- [4] S. Alexander, K. Vitali, and A. Petrunin, Invitation to Alexandrov geometry: CAT(0) spaces, Springer-Verlag, Berlin.
- [5] F. D. Ancel, M. W. Davis, and C. R. Guilbault, CAT(0) reflection manifolds, Geometric topology (Athens, GA, 1993), AMS/IP Stud. Adv. Math., vol. 2, Amer. Math. Soc., Providence, RI, 1997, pp. 441–445. MR1470741
- [6] E. Artin, *Theory of braids*, Ann. of Math. (2) **48** (1947), 101–126, DOI 10.2307/1969218. MR0019087
- [7] W. Ballmann and J. Świątkowski, On groups acting on nonpositively curved cubical complexes, Enseign. Math. (2) 45 (1999), no. 1-2, 51–81. MR1703363
- [8] A. Bartels and W. Lück, The Borel conjecture for hyperbolic and CAT(0)-groups, Ann. of Math. (2) 175 (2012), no. 2, 631–689, DOI 10.4007/annals.2012.175.2.5. MR2993750
- [9] N. Bergeron, F. Haglund, and D. T. Wise, *Hyperplane sections in arithmetic hyperbolic manifolds*, J. Lond. Math. Soc. (2) 83 (2011), no. 2, 431–448. MR2776645
- [10] _____, Hyperplane sections in arithmetic hyperbolic manifolds, J. Lond. Math. Soc. (2) 83 (2011), no. 2, 431–448, DOI 10.1112/jlms/jdq082. MR2776645
- [11] V. N. Berestovskii, Borsuk's problem on metrization of a polyhedron, Dokl. Akad. Nauk SSSR 268 (1983), no. 2, 273–277 (Russian). MR688238
- [12] N. Bergeron and D. T. Wise, A boundary criterion for cubulation, Amer. J. Math. 134 (2012), no. 3, 843–859, DOI 10.1353/ajm.2012.0020. MR2931226
- [13] E. A. Bering and D. Futer, Explicit special covers of alternating links. 1909.12266.
- [14] R. Bieri, Homological dimension of discrete groups, Mathematics Department, Queen Mary College, London, 1976. Queen Mary College Mathematics Notes. MR0466344
- [15] N. Bourbaki, *Lie groups and Lie algebras. Chapters 4–6*, Elements of Mathematics (Berlin), Springer-Verlag, Berlin, 2002. Translated from the 1968 French original by Andrew Pressley. MR1890629
- [16] B. Bekka, P. de la Harpe, and A. Valette, *Kazhdan's property (T)*, New Mathematical Monographs, vol. 11, Cambridge University Press, Cambridge, 2008. MR2415834
- [17] M. Bestvina, Geometric group theory and 3-manifolds hand in hand: the fulfillment of Thurston's vision, Bull. Amer. Math. Soc. (N.S.) 51 (2014), no. 1, 53–70. MR3119822
- [18] M. Bestvina and N. Brady, Morse theory and finiteness properties of groups, Invent. Math. 129 (1997), no. 3, 445–470.
- [19] N. Bourbaki, *Lie groups and Lie algebras. Chapters 4–6*, Elements of Mathematics (Berlin), Springer-Verlag, Berlin, 2002. Translated from the 1968 French original by Andrew Pressley. MR1890629
- [20] M. Bourdon, Immeubles hyperboliques, dimension conforme et rigidité de Mostow, Geom. Funct. Anal. 7 (1997), no. 2, 245–268 (French, with English and French summaries). MR1445387
- [21] M. Bourdon, Sur les immeubles fuchsiens et leur type de quasi-isométrie, Ergodic Theory Dynam. Systems 20 (2000), no. 2, 343–364, DOI 10.1017/S014338570000016X (French, with English and French summaries). MR1756974

- [22] N. Brady, I. J. Leary, and B. E. A. Nucinkis, On algebraic and geometric dimensions for groups with torsion, J. London Math. Soc. (2) 64 (2001), no. 2, 489–500. MR1853466
- [23] M. R. Bridson and A. Haefliger, *Metric spaces of non-positive curvature*, Grundlehren der Mathematischen Wissenschaften [Fundamental Principles of Mathematical Sciences], vol. 319, Springer-Verlag, Berlin, 1999. MR1744486
- [24] K. S. Brown, Cohomology of groups, Graduate Texts in Mathematics, vol. 87, Springer-Verlag, New York-Berlin, 1982. MR672956
- [25] M. Burger and S. Mozes, *Finitely presented simple groups and products of trees*, C. R. Acad. Sci. Paris Sér. I Math. 324 (1997), no. 7, 747–752, DOI 10.1016/S0764-4442(97)86938-8 (English, with English and French summaries). MR1446574
- [26] _____, Lattices in product of trees, Inst. Hautes Études Sci. Publ. Math. 92 (2000), 151–194 (2001). MR1839489
- [27] K.-U. Bux and C. Gonzalez, The Bestvina-Brady construction revisited: geometric computation of Σ-invariants for right-angled Artin groups, J. London Math. Soc. (2) 60 (1999), no. 3, 793–801, DOI 10.1112/S0024610799007991. MR1753814
- [28] P.-E. Caprace and F. Haglund, On geometric flats in the CAT(0) realization of Coxeter groups and Tits buildings, Canad. J. Math. 61 (2009), no. 4, 740–761, DOI 10.4153/CJM-2009-040-8. MR2541383
- [29] P.-E. Caprace and M. Sageev, Rank rigidity for CAT(0) cube complexes, Geom. Funct. Anal. 21 (2011), no. 4, 851–891. MR2827012
- [30] R. Charney, An introduction to right-angled Artin groups, Geom. Dedicata 125 (2007), 141–158, DOI 10.1007/s10711-007-9148-6. MR2322545
- [31] R. Charney and M. Davis, Singular metrics of nonpositive curvature on branched covers of Riemannian manifolds, Amer. J. Math. 115 (1993), no. 5, 929–1009, DOI 10.2307/2375063. MR1246182
- [32] R. Charney and M. W. Davis, *The* $K(\pi, 1)$ *-problem for hyperplane complements associated to infinite reflection groups*, J. Amer. Math. Soc. **8** (1995), no. 3, 597–627. MR1303028
- [33] _____, *Finite K*(π , 1)*s for Artin groups*, Prospects in topology (Princeton, NJ, 1994), Ann. of Math. Stud., vol. 138, Princeton Univ. Press, Princeton, NJ, 1995, pp. 110–124. MR1368655
- [34] _____, Strict hyperbolization, Topology 34 (1995), no. 2, 329–350. MR1318879
- [35] R. Charney, M. Davis, and G. Moussong, Nonpositively curved, piecewise Euclidean structures on hyperbolic manifolds, Michigan Math. J. 44 (1997), no. 1, 201–208, DOI 10.1307/mmj/1029005630. MR1439678
- [36] H. S. M. Coxeter, Discrete groups generated by reflections, Ann. of Math. (2) 35 (1934), no. 3, 588–621, DOI 10.2307/1968753. MR1503182
- [37] _____, Regular skew polyhedra in three and four dimensions and their topological analogues, Proc. London Math. Soc. 35 (1938), no. 1, 37–62.
- [38] J. Crisp, Symmetrical subgroups of Artin groups, Adv. Math. 152 (2000), no. 1, 159–177, DOI 10.1006/aima.1999.1895. MR1762124
- [39] J. Crisp and B. Wiest, Embeddings of graph braid and surface groups in right-angled Artin groups and braid groups, Algebr. Geom. Topol. 4 (2004), 439–472, DOI 10.2140/agt.2004.4.439. MR2077673
- [40] M. W. Davis, Groups generated by reflections and aspherical manifolds not covered by Euclidean space, Ann. of Math.
 (2) 117 (1983), no. 2, 293–324, DOI 10.2307/2007079. MR690848
- [41] _____, Coxeter groups and aspherical manifolds, Algebraic topology, Aarhus 1982 (Aarhus, 1982), Lecture Notes in Math., vol. 1051, Springer, Berlin, 1984, pp. 197–221, DOI 10.1007/BFb0075568. MR764580
- [42] _____, Some aspherical manifolds, Duke Math. J. 55 (1987), no. 1, 105–139, DOI 10.1215/S0012-7094-87-05507-4.
 MR883666
- [43] _____, Buildings are CAT(0), Geometry and cohomology in group theory (Durham, 1994), London Math. Soc. Lecture Note Ser., vol. 252, Cambridge Univ. Press, Cambridge, 1998, pp. 108–123. MR1709955
- [44] _____, The cohomology of a Coxeter group with group ring coefficients, Duke Math. J. 91 (1998), no. 2, 297–314, DOI 10.1215/S0012-7094-98-09113-X. MR1600586
- [45] _____, Poincaré duality groups, Surveys on surgery theory, Vol. 1, Ann. of Math. Stud., vol. 145, Princeton Univ. Press, Princeton, NJ, 2000, pp. 167–193. MR1747535
- [46] _____, Exotic aspherical manifolds, Topology of high-dimensional manifolds, No. 1, 2 (Trieste, 2001), ICTP Lect. Notes, vol. 9, Abdus Salam Int. Cent. Theoret. Phys., Trieste, 2002, pp. 371–404. MR1937019
- [47] _____, Nonpositive curvature and reflection groups, Handbook of geometric topology, North-Holland, Amsterdam, 2002, pp. 373–422. MR1886674
- [48] _____, The geometry and topology of Coxeter groups, London Mathematical Society Monographs Series, vol. 32, Princeton University Press, Princeton, NJ, 2008. MR2360474
- [49] _____, Examples of buildings constructed via covering spaces, Groups Geom. Dyn. 3 (2009), no. 2, 279–298. MR2486800

- [50] _____, Lectures on orbifolds and reflection groups, Transformation groups and moduli spaces of curves, Adv. Lect. Math. (ALM), vol. 16, Int. Press, Somerville, MA, 2011, pp. 63–93. MR2883685
- [51] _____, Right-angularity, flag complexes, asphericity, Geom. Dedicata 159 (2012), 239–262, DOI 10.1007/s10711-011-9654-4. MR2944529
- [52] M. W. Davis, J. Fowler, and J.-F. Lafont, Aspherical manifolds that cannot be triangulated, Algebr. Geom. Topol. 14 (2014), no. 2, 795–803, DOI 10.2140/agt.2014.14.795. MR3159970
- [53] M. W. Davis and J. Huang, Determining the action dimension of an Artin group by using its complex of abelian subgroups, Bull. Lond. Math. Soc. 49 (2017), no. 4, 725–741. MR3725492
- [54] M. W. Davis and T. Januszkiewicz, Hyperbolization of polyhedra, J. Differential Geom. 34 (1991), no. 2, 347–388. MR1131435
- [55] _____, Convex polytopes, Coxeter orbifolds and torus actions, Duke Math. J. 62 (1991), no. 2, 417–451, DOI 10.1215/S0012-7094-91-06217-4. MR1104531
- [56] _____, Right-angled Artin groups are commensurable with right-angled Coxeter groups, J. Pure Appl. Algebra 153 (2000), no. 3, 229–235.
- [57] M. W. Davis, T. Januszkiewicz, and R. Scott, Nonpositive curvature of blow-ups, Selecta Math. (N.S.) 4 (1998), no. 4, 491–547. MR1668119
- [58] _____, Fundamental groups of blow-ups, Adv. Math. 177 (2003), no. 1, 115–179. MR1985196
- [59] M. W. Davis, T. Januszkiewicz, and S. Weinberger, *Relative hyperbolization and aspherical bordisms: an addendum to "Hyperbolization of polyhedra"*, J. Differential Geom. 58 (2001), no. 3, 535–541. MR1906785
- [60] M. W. Davis, G. Le, and K. Schreve, Action dimensions of simple complexes of groups, J. of Topology 12 (2019), 1266-1314.
- [61] M. W. Davis and I. J. Leary, Some examples of discrete group actions on aspherical manifolds, High-dimensional manifold topology, World Sci. Publ., River Edge, NJ, 2003, pp. 139–150. MR2048719
- [62] M. W. Davis and G. Moussong, Notes on nonpositively curved polyhedra, Low dimensional topology (Eger, 1996/Budapest, 1998), Bolyai Soc. Math. Stud., vol. 8, János Bolyai Math. Soc., Budapest, 1999, pp. 11–94. MR1747268
- [63] M. W. Davis and B. Okun, Cohomology computations for Artin groups, Bestvina-Brady groups, and graph products, Groups Geom. Dyn. 6 (2012), no. 3, 485–531, DOI 10.4171/GGD/164. MR2961283
- [64] P. Deligne, Les immeubles des groupes de tresses généralisés, Invent. Math. 17 (1972), 273–302 (French). MR0422673
- [65] C. Droms, Graph groups, coherence, and three-manifolds, J. Algebra 106 (1987), no. 2, 484–489.
- [66] S. Eilenberg and T. Ganea, On the Lusternik-Schnirelmann category of abstract groups, Ann. of Math. (2) 65 (1957), 517–518, DOI 10.2307/1970062. MR0085510
- [67] B. Farb, Relatively hyperbolic groups, Geom. Funct. Anal. 8 (1998), no. 5, 810–840, DOI 10.1007/s000390050075. MR1650094
- [68] B. Farb, C. Hruska, and A. Thomas, Problems on automorphism groups of nonpositively curved polyhedral complexes and their lattices, Geometry, rigidity, and group actions, Chicago Lectures in Math., Univ. Chicago Press, Chicago, IL, 2011, pp. 515–560, DOI 10.7208/chicago/9780226237909.001.0001. MR2807842
- [69] B. Farb and L. Mosher, Convex cocompact subgroups of mapping class groups, Geom. Topol. 6 (2002), 91–152, DOI 10.2140/gt.2002.6.91. MR1914566
- [70] F. T. Farrell and J.-F. Lafont, EZ-structures and topological applications, Comment. Math. Helv. 80 (2005), no. 1, 103–121, DOI 10.4171/CMH/7. MR2130569
- [71] J. Fowler, Finiteness properties for some rational Poincaré duality groups, Illinois J. Math. 56 (2012), no. 2, 281–299. MR3161324
- [72] S. R. Gal, Euler characteristic of the configuration space of a complex, Colloq. Math. 89 (2001), no. 1, 61–67, DOI 10.4064/cm89-1-4. MR1853415
- [73] D. Gaboriau and F. Paulin, *Sur les immeubles hyperboliques*, Geom. Dedicata **88** (2001), no. 1-3, 153–197 (French, with English summary). MR1877215
- [74] D. E. Galewski and R. J. Stern, Classification of simplicial triangulations of topological manifolds, Ann. of Math. (2) 111 (1980), no. 1, 1–34, DOI 10.2307/1971215. MR558395
- [75] D. Galewski and R. Stern, A universal 5-manifold with respect to simplicial triangulations, Geometric topology (Proc. Georgia Topology Conf., Athens, Ga., 1977), Academic Press, New York-London, 1979, pp. 345–350. MR537740
- [76] M. Gromov, *Hyperbolic groups*, Essays in group theory, Math. Sci. Res. Inst. Publ., vol. 8, Springer, New York, 1987, pp. 75–263. MR919829
- [77] A. Haefliger, Extension of complexes of groups, Ann. Inst. Fourier (Grenoble) 42 (1992), no. 1-2, 275–311 (English, with French summary). MR1162563
- [78] F. Haglund, Finite index subgroups of graph products, Geom. Dedicata 135 (2008), 167–209. MR2413337
- [79] F. Haglund and D. T. Wise, Special cube complexes, Geom. Funct. Anal. 17 (2008), no. 5, 1551–1620. MR2377497

- [80] A. Hatcher, *Algebraic topology*, Cambridge University Press, Cambridge, 2002. MR1867354
- [81] F. Haglund and D. T. Wise, A combination theorem for special cube complexes, Ann. of Math. (2) 176 (2012), no. 3, 1427–1482, DOI 10.4007/annals.2012.176.3.2. MR2979855
- [82] G. C. Hruska, E. Stark, and H. C. Tran, Surface group amalgams that (don't) act on 3-manifolds. 1705.01361v2.
- [83] G. C. Hruska and D. T. Wise, Finiteness properties of cubulated groups, Compos. Math. 150 (2014), no. 3, 453–506, DOI 10.1112/S0010437X13007112. MR3187627
- [84] B. Z. Hu, A PL geometric study of algebraic K theory, Trans. Amer. Math. Soc. 334 (1992), no. 2, 783–808, DOI 10.2307/2154482. MR1085943
- [85] _____, Whitehead groups of finite polyhedra with nonpositive curvature, J. Differential Geom. 38 (1993), no. 3, 501–517. MR1243784
- [86] _____, Retractions of closed manifolds with nonpositive curvature, Geometric group theory (Columbus, OH, 1992), Ohio State Univ. Math. Res. Inst. Publ., vol. 3, de Gruyter, Berlin, 1995, pp. 135–147. MR1355108
- [87] B. Z. Hu, Relative hyperbolization and Pontrjagin classes, C. R. Math. Acad. Sci. Paris 342 (2006), no. 4, 253–257, DOI 10.1016/j.crma.2005.12.020 (English, with English and French summaries). MR2196008
- [88] T. Hsu and D. T. Wise, Embedding theorems for non-positively curved polygons of finite groups, J. Pure Appl. Algebra 123 (1998), no. 1-3, 201–221, DOI 10.1016/S0022-4049(96)00077-1. MR1492901
- [89] _____, On linear and residual properties of graph products, Michigan Math. J. 46 (1999), no. 2, 251–259.
- [90] T. Hsu and D. Wise, A non-residually finite square of finite groups, Groups St. Andrews 1997 in Bath, I, London Math. Soc. Lecture Note Ser., vol. 260, Cambridge Univ. Press, Cambridge, 1999, pp. 368–378. MR1676633
- [91] P. H. Kropholler and A. Martino, Graph-wreath products and finiteness conditions, J. Pure Appl. Algebra 220 (2016), no. 1, 422–434, DOI 10.1016/j.jpaa.2015.07.001. MR3393469
- [92] I. J. Leary, Uncountably many groups of type FP, Proc. Lond. Math. Soc. (3) 117 (2018), no. 2, 246–276, DOI 10.1112/plms.12135. MR3851323
- [93] I. J. Leary and A. Minasyan.
- [94] I. J. Leary and B. E. A. Nucinkis, Some groups of type VF, Invent. Math. 151 (2003), no. 1, 135–165, DOI 10.1007/s00222-002-0254-7. MR1943744
- [95] I. J. Leary and N. Petrosyan, On dimensions of groups with cocompact classifying spaces for proper actions, Adv. Math. 311 (2017), 730–747, DOI 10.1016/j.aim.2017.03.008. MR3628229
- [96] H. van der Lek.
- [97] C. Manolescu, Pin(2)-equivariant Seiberg-Witten Floer homology and the triangulation conjecture, J. Amer. Math. Soc. 29 (2016), no. 1, 147–176, DOI 10.1090/jams829. MR3402697
- [98] T. Matumoto, *Triangulation of manifolds*, Algebraic and geometric topology (Proc. Sympos. Pure Math., Stanford Univ., Stanford, Calif., 1976), Proc. Sympos. Pure Math., XXXII, Amer. Math. Soc., Providence, R.I., 1978, pp. 3–6. MR520517
- [99] J. Meier, H. Meinert, and L. VanWyk, Higher generation subgroup sets and the Σ-invariants of graph groups, Comment. Math. Helv. 73 (1998), no. 1, 22–44, DOI 10.1007/s000140050044. MR1610579
- [100] G. Mess, Examples of Poincaré duality groups, Proc. Amer. Math. Soc. 110 (1990), no. 4, 1145–1146, DOI 10.2307/2047770. MR1019274
- [101] D. W. Morris, Introduction to arithmetic groups, Deductive Press, [place of publication not identified], 2015.
- [102] G. Moussong, Hyperbolic Coxeter groups, ProQuest LLC, Ann Arbor, MI, 1988. Thesis (Ph.D.)–The Ohio State University. MR2636665
- [103] G. Moussong, Some nonsymmetric manifolds, Differential geometry and its applications (Eger, 1989), Colloq. Math. Soc. János Bolyai, vol. 56, North-Holland, Amsterdam, 1992, pp. 535–546. MR1211680
- [104] G. Niblo and L. Reeves, Groups acting on CAT(0) cube complexes, Geom. Topol. 1 (1997), 1–7, DOI 10.2140/gt.1997.1.1. MR1432323
- [105] G. A. Niblo and L. D. Reeves, *The geometry of cube complexes and the complexity of their fundamental groups*, Topology 37 (1998), no. 3, 621–633, DOI 10.1016/S0040-9383(97)00018-9. MR1604899
- [106] G. A. Niblo and M. A. Roller, *Groups acting on cubes and Kazhdan's property (T)*, Proc. Amer. Math. Soc. **126** (1998), no. 3, 693–699, DOI 10.1090/S0002-9939-98-04463-3. MR1459140
- [107] B. Okun and K. Schreve, *The L²-(co)homology of groups with hierarchies*, Algebr. Geom. Topol. **16** (2016), no. 5, 2549–2569, DOI 10.2140/agt.2016.16.2549. MR3572340
- [108] B. Okun and R. Scott, L^2 -homology and reciprocity for right-angled Coxeter groups, Fund. Math. **214** (2011), no. 1, 27–56, DOI 10.4064/fm214-1-3. MR2845632
- [109] R. Oliver, Fixed-point sets of group actions on finite acyclic complexes, Comment. Math. Helv. 50 (1975), 155–177, DOI 10.1007/BF02565743. MR0375361
- [110] T. E. Panov and Ya. A. Verëvkin, Polyhedral products and commutator subgroups of right-angled Artin and Coxeter groups, Mat. Sb. 207 (2016), no. 11, 105–126, DOI 10.4213/sm8701 (Russian, with Russian summary); English transl., Sb. Math. 207 (2016), no. 11-12, 1582–1600. MR3588981

- [111] T. Panov and Y. Veryovkin, On the commutator subgroup of a right-angled Artin group, J. Algebra 521 (2019), 284–298, DOI 10.1016/j.jalgebra.2018.11.022. MR3892708
- [112] L. Paris, On the fundamental group of the complement of a complex hyperplane arrangement, Arrangements—Tokyo 1998, 2000, pp. 257–272.
- [113] _____, Intersection subgroups of complex hyperplane arrangements, Topology and its Applications **105** (2000), no. 3, 319–343.
- [114] _____, Parabolic subgroups of Artin groups, Journal of Algebra 196 (1997), no. 2, 369–399.
- [115] _____, The Deligne complex of a real arrangement of hyperplanes, Nagoya Mathematical Journal 131 (1993), 39–65.
- [116] N. Petrosyan and Tomasz Prytula, *Bestvina complex for group actions with a strict fundamental domain*, Groups, Geom. Dynamics.
- [117] M. Ronan, Lectures on buildings, Perspectives in Mathematics, vol. 7, Academic Press, Inc., Boston, MA, 1989. MR1005533
- [118] M. Sageev, Ends of group pairs and non-positively curved cube complexes, Proc. London Math. Soc. (3) 71 (1995), no. 3, 585–617. MR1347406
- [119] _____, Codimension-1 subgroups and splittings of groups, J. Algebra 189 (1997), no. 2, 377–389. MR1438181
- [120] _____, CAT(0) cube complexes and groups, Geometric group theory, IAS/Park City Math. Ser., vol. 21, Amer. Math. Soc., Providence, RI, 2014, pp. 7–54. MR3329724
- [121] M. Sageev and D. T. Wise, *The Tits alternative for* CAT(0) *cubical complexes*, Bull. London Math. Soc. **37** (2005), no. 5, 706–710, DOI 10.1112/S002460930500456X. MR2164832
- [122] M. Salvetti, Topology of the complement of real hyperplanes in \mathbb{C}^n , Invent. Math. 88 (1987), 603-618.
- [123] M. Sapir, A Higman embedding preserving asphericity, J. Amer. Math. Soc. 27 (2014), no. 1, 1–42, DOI 10.1090/S0894-0347-2013-00776-6. MR3110794
- [124] R. Scott, Growth series for vertex-regular CAT(0) cube complexes, Algebr. Geom. Topol. 7 (2007), 285–300, DOI 10.2140/agt.2007.7.285. MR2308945
- [125] K. Schreve and E. Stark, Nonplanar graphs in boundaries of CAT(0) groups. 1807.02439v1.
- [126] R. Scott, *Right-angled mock reflection and mock Artin groups*, Trans. Amer. Math. Soc. **360** (2008), no. 8, 4189–4210, DOI 10.1090/S0002-9947-08-04452-8. MR2395169
- [127] _____, Eulerian cube complexes and reciprocity, Algebr. Geom. Topol. 14 (2014), no. 6, 3533–3552, DOI 10.2140/agt.2014.14.3533. MR3302970
- [128] J.-P. Serre, Trees, Springer-Verlag, Berlin-New York, 1980. Translated from the French by John Stillwell. MR607504
- [129] C. C. Squier, The homological algebra of Artin groups, Math. Scand. 75 (1994), no. 1, 5–43, DOI 10.7146/math.scand.a-12500. MR1308935
- [130] J. Stallings, A finitely presented group whose 3-dimensional integral homology is not finitely generated, Amer. J. Math. 85 (1963), 541–543, DOI 10.2307/2373106. MR0158917
- [131] J. Świątkowski, Trivalent polygonal complexes of nonpositive curvature and Platonic symmetry, Geom. Dedicata 70 (1998), no. 1, 87–110, DOI 10.1023/A:1004954009167. MR1612350
- [132] _____, Estimates for homological dimension of configuration spaces of graphs, Colloq. Math. 89 (2001), no. 1, 69–79, DOI 10.4064/cm89-1-5. MR1853416
- [133] J. Tits, Structures et groupes de Weyl, Séminaire Bourbaki, Vol. 9, Soc. Math. France, Paris, 1995, pp. Exp. No. 288, 169–183 (French). MR1608796
- [134] _____, Buildings of spherical type and finite BN-pairs, Lecture Notes in Mathematics, Vol. 386, Springer-Verlag, Berlin-New York, 1974. MR0470099
- [135] J. Tits, A local approach to buildings, The geometric vein, Springer, New York-Berlin, 1981, pp. 519–547. MR661801
- [136] J. Tits, Buildings and group amalgamations, Proceedings of groups—St. Andrews 1985, London Math. Soc. Lecture Note Ser., vol. 121, Cambridge Univ. Press, Cambridge, 1986, pp. 110–127. MR896503
- [137] È. B. Vinberg, Discrete linear groups that are generated by reflections, Izv. Akad. Nauk SSSR Ser. Mat. 35 (1971), 1072–1112 (Russian). MR0302779
- [138] D. T. Wise, A non-Hopfian automatic group, J. Algebra 180 (1996), no. 3, 845–847, DOI 10.1006/jabr.1996.0096. MR1379212
- [139] _____, Non-positively curved squared complexes: Aperiodic tilings and non-residually finite groups, ProQuest LLC, Ann Arbor, MI, 1996. Thesis (Ph.D.)–Princeton University. MR2694733
- [140] _____, The residual finiteness of negatively curved polygons of finite groups, Invent. Math. 149 (2002), no. 3, 579–617, DOI 10.1007/s002220200224. MR1923477
- [141] _____, Complete square complexes, Comment. Math. Helv. 82 (2007), no. 4, 683–724, DOI 10.4171/CMH/107. MR2341837
- [142] _____, From riches to raags: 3-manifolds, right-angled Artin groups, and cubical geometry, CBMS Regional Conference Series in Mathematics, vol. 117, Published for the Conference Board of the Mathematical Sciences, Washington, DC; by the American Mathematical Society, Providence, RI, 2012. MR2986461